Robust Selective Search

Yubin Kim

CMU-LTI-18-009

Language Technologies Institute School of Computer Science Carnegie Mellon University 5000 Forbes Ave., Pittsburgh, PA 15213 www.lti.cs.cmu.edu

Thesis Committee:

Jamie Callan, Chair Jaime Carbonell Ralf Brown Alistair Moffat (The University of Melbourne)

Submitted in partial fulfillment of the requirements for the degree of Doctor of Philosophy In Language and Information Technologies

©2019, Yubin Kim

Yubin Kim: Robust Selective Search, © Jan 2019

ACKNOWLEDGEMENTS

I am fortunate to have had Jamie Callan guide my academic beginnings. I firmly believe that somehow, by sheer luck, I ended up with the best advisor in the world. Thank you for your gentle encouragements when I needed them and your not-so-gentle admonishments when I needed those too. Thank you for being not just an academic advisor, but also a mentor and an advocate.

Warm thanks to J. Shane Culpepper and Alistair Moffat who were practically co-advisors for a large chunk of my PhD. Their attention to detail and high expectations made me a better researcher.

Anagha Kulkarni, Reyyan Yeniterzi, and Zhuyun Dai, my co-authors: thank you for being such a pleasure to work with. You have set a high bar for future collaborators.

To my mentors during my internship at Microsoft Research, Keyvn Collins-Thompson and Jaime Teevan, thank you for a fantastic summer and continuing to be mentors to me long after my internship.

To the West Coast Swing community here in Pittsburgh and world wide: thank you for being the creative outlet and sanctuary I needed to escape from academia.

To my friends. Thanks for the laughs, the cookies, and the board games. I am fatter and happier thanks to all of you.

To my parents and Marilyn, my mother-in-law: thank you for being supportive and understanding. I never had to dread the "so, when are you graduating?" question that so many of my fellow students have come to fear.

To my husband Max, defender of my sanity, conjurer of laughs, slayer of anxieties, my cheerleader and friend: you are a pretty cool and funny dude. I love you.

ABSTRACT

Selective search is a modern distributed search architecture designed to reduce the computational cost of large-scale search. Selective search creates topical shards that are deliberately content-skewed, placing highly similar documents together in the same shard. During query time, rather than searching the entire corpus, a resource selection algorithm selects a subset of the topic shards likely to contain documents relevant to the query and search is only performed on these shards. This substantially reduces total computational costs of search while maintaining accuracy comparable to exhaustive distributed search.

Prior work has shown selective search to be effective in smaller scale, single query-at-a-time environments. However, modern practical, large-scale search and text analysis systems are often multi-stage pipeline systems where an initial, first-stage fast candidate retrieval forwards results onto downstream complex analysis components. These systems often contain other optimization components and are run in a parallel setting over multiple machines. This dissertation aims to bring selective search to wider adoption by addressing the questions related to efficiency and effectiveness in a practical implementation such as: do different instantiations of selective search have stable performance; does selective search combine well with other optimization components; can selective search deliver the high recall necessary to serve as a first-stage retrieval system? In addition, this dissertation provides tools to empower system administrators so that they can easily design and test selective search systems without full implementations.

First, this dissertation research investigates the effects of non-deterministic steps that exist in selective search on its accuracy and found the variance across system instances is acceptable. Then, selective search was combined with WAND, a common dynamic pruning algorithm and it was shown that selective search and WAND has *better*-than-additive gains due to the long posting lists of the topically focused shards. This dissertation also presents new resource selection algorithms to achieve high recall, which has been an elusive goal in prior work. A learning-to-rank based approach to resource selection can be trained without human-judged relevance data to be highly accurate and statistically equivalent to exhaustive search at deeper metrics such as MAP@1000. This result enables selective search to be used as a first-stage retrieval component in realistic multi-stage text analysis systems.

When placed in a parallel query processing environment, it was found that with judicious load-balancing to manage unequal popularity of shards, the efficiency claims of prior work remain relevant in a fully parallel processing setting, and are applicable to larger computational environments as well. A detailed simulator was built to investigate this research question and serve as a powerful tool for administrators to test out different selective search configurations.

Finally, new evaluation metric, AUReC, is presented which can be used to easily evaluate a mapping of documents to shards without implementing a full selective search system. This allows system designers to quickly sift through many potential different document allocations to easily identify the best system configuration.

Ultimately, the dissertation aims to enable cost and energy-efficient use of large-scale data collections in not only information retrieval research, but also in other fields such as text mining and question answering, in academia and industry alike, fueling future innovation.

CONTENTS

1	INT	RODUCTION 1
	1.1	Contributions 3
		1.1.1 Variance of random processes 3
		1.1.2 Performance with dynamic pruning 4
		1.1.3 Resource selection 5
		1.1.4 Performance in parallel processing 6
		1.1.5 Partition evaluation 8
	1.2	Wider Impact 9
	1.3	Future Directions 9
		1.3.1 Selective search in pipelines 9
		1.3.2 Selective search in hierarchical corpora 9
		1.3.3 Selective search in streaming datasets 10
	1.4	Organization 10
2	REL	ATED WORK 11
	2.1	Traditional Distributed Search 11
	2.2	Cluster-based Retrieval 12
	2.3	Federated Search 13
	2.4	Selective Search 14
		2.4.1 Datasets 17
		2.4.2 Shards 17
		2.4.3 Resource selection 18
		2.4.4 Evaluation of accuracy 18
		2.4.5 Evaluation of efficiency 19
3	VAR	IANCE 21
	3.1	Experimental Method 22
	3.2	Experiment Results 22
		3.2.1 Variability for query sets 22
		3.2.2 Variability for queries 24
	3.3	Summary 27
4	DYN	IAMIC OPTIMIZATION 29
	4.1	Related Work 29
	4.2	Experiments 30
		4.2.1 Pruning effectiveness of WAND on topical shards 32
		4.2.2 WAND and resource ranking interactions 33
		4.2.3 Improving efficiency with cascaded pruning thresholds 36
	4.3	Summary 38
5	RES	OURCE SELECTION WITH WAND 39
	5.1	Methods 39
	5.2	Experimental Setup 42
	5.3	Experiments 42

- 5.4 Summary 44
- 6 LEARNING TO RANK RESOURCES 47

6.1 Related Work 47 6.2 Model 48 6.3 Features 48 Query-independent information 6.3.1 48 6.3.2 Term-based statistics 48 6.3.3 Sample-document (CSI-based) features 49 6.4 Experimental Methodology 50 6.5 Experiments 51 6.5.1 Overall comparison 51 Effect of training data 6.5.2 51 6.5.3 Query length 51 6.5.4 Feature analysis 52 6.6 Summary 54 SIMULATOR 57 7 58 7.1 Related Work 7.2 Simulation Model 58 7.3 Experimental Methodology 62 7.3.1 Document collections 62 7.3.2 Resource selection parameter settings 63 7.3.3 Query streams 65 7.4 Experimental Results 66 7.4.1 Selective search efficiency 66 Resource allocation 7.4.2 70 Shard assignment policies 7.4.3 72 Scalability 76 7.4.47.5 Summary 77 8 AUREC 83 8.1 Related Work and Motivation 84 8.2 AUReC 85 **Experimental Setup** 8.3 88 8.3.1 Data and query sets 88 Document retrieval 8.3.2 89 Resource selection and baselines 8.3.3 89 Metrics 8.3.4 91 **Experimental Results** 8.4 92 Comparison with end-to-end systems 92 8.4.1 Different metrics and evaluation depths 8.4.2 96 8.4.3 Weaker retrieval engine 99 Different queries 8.4.4 100 8.5 Summary and Recommendations 102 CONCLUSION 105 9 9.1 Summary of Findings 105 9.2 Contributions and Wider Impact 107 Future Work 108 9.3 Selective search in pipelines 9.3.1 109 9.3.2 Selective search in hierarchical corpora 110

9.3.3 Selective search in streaming datasets 111

BIBLIOGRAPHY 113

LIST OF FIGURES

Figure 1	Illustration of two different search architectures. Each box represents a machine and the cylinders represent parts of the search engine index ("shards"). The dark cylinders are shards that are searched; traditional distributed search always searches all shards while selective search searches only a few for a given query. Selective search also requires an additional resource selection database (grey hashing) in order to select the shards to search. Both architectures require a merge step (triangle) that merges the
Figure 2	partial result lists from the shards. 2 Accuracy distribution of Rank-S and RBR system instances for the Web Track queries. Mid line in boxes represents median. Outer edges of the boxes repre- sent lower and upper quartiles, while whiskers indicate 1.5 interquartile range.
	+ indicates outliers. Blue dot is the performance of the typical 'search all shards' baseline. 23
Figure 3	Average Precision of RBR instances for Web Track queries. Each data point is a box and whisker plot with the edges of the boxes representing the upper and lower quantiles, the mid line the median, and the whiskers the 1.5 interquartile range. + are outliers 26
Figure 4	Distribution of shard sizes, with a total of 100 shards.
Figure 5	Correlation between the number of postings processed for a query and the time taken for query evaluation. Data points are generated from MQT queries using both WAND and full evaluation, applied independently to all 100 topical shards and all 100 random shards. In total, $756 \times 200 \times 2 \approx$ 300,000 points are plotted.
Figure 6	Ratio of savings achieved by WAND as a function of the total postings length of each query in the AOL set, measured on a per shard basis. A total of $100 \times 713 \approx 71,000$ points are plotted. Queries containing only rare terms derive little benefit from WAND. 32
Figure 7	Distribution of query response times for MQT queries on shards: (a) as a box plot distribution, with a data point plotted for each query-shard pair; (b) as a table of corresponding means and medians. In (a), the center line of the box indicates the median, the outer edges of the box the first and third quartiles, and the blue circle the mean. The whiskers extend to include all points within 1.5 times the inter-quartile range of the box. The graph was truncated to omit a small number of extreme points for both Rank-S Full and Taily-Full. The maximum time for both these two runs was 6,611 ms.
Figure 8	Normalized 1,000 th document scores from shards, averaged over queries and then shard ranks, and expressed as a fraction of the collection- wide maximum document score for each corresponding query. The score falls with rank, as fewer high-scoring documents appear in lower-ranked shards. 36

- Figure 9 The micro-average w/b ratio for WAND postings evaluations, as a function of the per query shard rankings assigned by Taily. Early shards generate greater savings. 36
- Figure 10 Normalized 1,000 th document scores from shards relative to the highest score attained by any document for the corresponding query, microaveraged over queries, assuming that shards are processed sequentially rather than in parallel, using the Taily-based ordering of topical shards and a random ordering of the same shards. 37
- Figure 11 Ratio of postings evaluated by WAND for independent shard search versus sequential shard search, AOL queries with micro-averaging. Shard ranking was determined by Taily. 37
- Figure 12 Comparison of impact ordered posting list and source ordered posting list block max score distributions of the term *obama* in a top shard for the query *obama family tree* (shard 57). Block size is n = 128. 40
- Figure 13 An example of the multi-size second scores method. If k = 2, $b_{1,1}$ is the first block selected ($\hat{k} = 1$), then $b_{1,2}$ ($\hat{k} = 2$) resulting in shard 1 being selected for search, as ideal. However, if k = 3, $b_{2,1}$ ($\hat{k} = 3$) is also selected, resulting in both shards being selected for search even though shard 2 is unnecessary. 41
- Figure 14 MAP@1000 for queries on CW09-B, grouped by query length. Parentheses on the X axis present the number of queries in each group. T is the shard rank cutoff. 53
- Figure 15 Architecture of the selective distributed search system. The i'th of the M machines has c_i cores, each of which can be used for resource selection and result merging, or for shard search across any of the p_i shards allocated to this machine. Only a defined subset of the machines are able to perform resource selection. Figure by J. Shane Culpepper, reused from Kim et al. [2017]. 59
- Figure 16 Sizes of shards used in experiments reported by [Kulkarni 2013]. 63
- Figure 17 Exhaustive search and selective search using CWo9-A and Gov2 and random shard assignment for the AOL_W Test and MQT_W Test, AOL_G Test and MQT_G Test query sets (shown as AOL and MQT in the labels for brevity). In these experiments, M = 2, B = 1, and S = 2. 67
- Figure 18 Average utilization of machines using Rank-S, the CWo9-A dataset, the AOL_W Test queries, M = 2, B = 1 and S = 2. Each point is the mean over 10 sequences of 1,000 queries; error bars represent 95% confidence intervals. 70
- Figure 19 Varying the resources assigned to broker processes using Rank-S, with M = 2, and random shard assignment. 71
- Figure 20 Popularity of shards in the CWo9-A training query sets for the top 10% of most frequently accessed shards as selected by Rank-S. The vertical axis indicates the total number of times a shard is selected for all queries. 72
- Figure 21 Utilization of machines for selective search on Gov2, using Taily, the MQT_G Test queries, M = 4, B = 1, S = 4 and a uniform shard distribution. Each point is the mean over 10 sequences of 1,000 queries; error bars represent 95% confidence intervals. The broker resides in m₁ for both configurations. The two frames share a common legend. 73

- Figure 22 The variance of the load spread (max load_i min load_i) of ten Random shard allocations, with M = 4, B = 1. The midline of the box indicates the median, the outer edges the 25th and 75th percentile values, and the whiskers the most outlying values. The load spread for Log-based shard allocation is also plotted as a reference point. Note the different horizontal and vertical scales in the two panes. 75
- Figure 23 The variance of the load spread $(\max \text{load}_i \min \text{load}_i)$ of ten Random shard allocations, now with M = 8, B = 8. All other parameters are as for Figure 22. Note the different horizontal scales in the two panes. 75
- Figure 24 The effect of shard assignment policies on throughput of selective search, using Taily with M = 4, B = 1 (left column) and M = 8, B = 8 (right column). Note the differing horizontal scales in each pair. 79
- Figure 25 Selective search compared to exhaustive search with M = 64, B = 64 using CWo9-A and log-based shard assignment for the AOL_W Test and MQT_W Test query sets (shown as AOL and MQT in the labels for brevity). 80
- Figure 26 The total time required by a selective search system to process 10,000 queries as a function M, the number of machines. Assumes an infinite query arrival rate and B = M in all configurations. The two query streams used are MQT_G Test and AOL_G Test, denoted as MQT and AOL respectively. The red dash-dot line represents the ideal case, where throughput is doubled with doubled machine capacity. 80
- Figure 27 Throughput achieved with Log-based shard assignment and M = 64, B = 64 to process queries against CWo9-A. The mirrored configuration is a doubled M = 32, B = 32 system, including disk space. The two query streams are MQT_G Test and AOL_G Test, denoted as MQT and AOL respectively. 81
- Figure 28 Example of a recall versus fraction of shards searched curve for three different index partitions for the same query. 87
- Figure 29 Contingency table for pair-wise comparison of index partitions using AU-ReC and end-to-end system evaluation (EtE). 91
- Figure 30 Correlation between AUReC scores and P@1000 scores of an end-to-end selective search system using Rank-S resource selection. Different shapes indicate that the shard map was created using the KLD-Rand (★), QKLD-Rand (×), or QKLD-Qinit (+) method as described in Dai et al. [2016]. 95

LIST OF TABLES

Table 1	Standard Deviation (SD) and Variance Coefficient (VC) of accuracy scores
	for Rank-S instances. 24
Table 2	Standard Deviation (SD) and Variance Coefficient (VC) of accuracy scores for RBR
	instances. 25
Table 3	Comparison of MAP@500 scores (CW09-A) for Rank-S and Taily instances. 25
Table 4	Ratio of per shard per query postings evaluated and per shard per query execution time for WAND-based search, as ratios relative to unpruned search, averaged over 100 topical shards and over 100 randomized shards, and over two groups each of 700+ queries. The differences between the Topical and Random macro-averaged ratios are significant for both query
	sets and both measures (naired two-tailed t-test $n < 0.01$)
Table 5	Average number of shards searched, and micro-averaged postings ratios for those selected shards and for the complement set of shards, together with the corresponding query time cost ratios, in each case comparing WAND-based search to unpruned search. Smaller numbers indicate greater
	savings. 33
Table 6	As for Table 5, but showing macro-averaged ratios. All differences between selected and non-selected shards are significant (paired two-tailed t-test,
	p < 0.01). 34
lable 7	The results of second scores and M3S resource selection methods with dif- ferent cut-off alternatives compared to exhaustive search and state of the art resource selection algorithm baselines for 58 single term queries. Avg shards shows the average number of shards selected for each query. For the M3S, $k = 100$ unless otherwise specified. * indicates method is within a 10% margin of exhaustive search using a non-inferiority test with $p = 0.05$. + indicates result is within a 5% margin. 43
Table 8	Search accuracy comparison between 3 L2R models and baselines at two rank cutoffs (T) for two datasets. ▲: statistically significant improvement compared to Jnt, the best resource selection baseline. *: non-inferiority to exhaustive search . 52
Table 9	Effectiveness and efficiency of FAST features. ALL uses all features. FAST does
	not use sample-document features. T: shard rank cutoff. *: non-inferiority to ex-
	haustive. 53
Table 10	Performance of L2R-MQT using feature sets constructed with leave-one-out. 'w/o
	X' means the feature was excluded from FAST. Text in bold indicates the lowest
	value in the column. 54
Table 11	Simulation parameters. 60

Table 12	Resource selection parameter settings, based on TREC Terabyte Track and
	developed by [Kulkarni et al. 2012] and [Alv et al. 2012]. These investiga
	tions used a variaty of CEL sizes for Cave renging from a π^0 to π^0 . We
	uons used a variety of CSI sizes for Gov2, ranging from 0.5% to 4% . We
	used 1%, for consistency with CW09-A. The table also presents the average
- 1 1	number of shards selected by the parameter settings. 63
Table 13	Effectiveness of selective search using various resource selection algorithms
	on the Gov2 dataset. The "Oracle" shard ranking assumes that the most
	useful four shards are identified and searched for each query. 64
Table 14	Effectiveness of selective search using various resource selection algorithms
	on the CWo9-A dataset. The "Oracle" shard ranking assumes that the most
	useful four shards are identified and searched for each query. 64
Table 15	Sizes of the query logs, in thousands. The "Train" column indicates training
-	queries used to set parameters. The "Test" column indicates testing queries
	used for reporting results. Columns "Test _W " and "Test _M " indicate queries
	that were sampled starting one week and one month after the Train gueries,
	respectively. Those two query streams are used in Section 7.4.3.
Table 16	Average number of shards selected by Taily and Rank-S for the Test logs.
	measured using the configurations depicted in Figure 17. 67
Table 17	Dispersion of per-query processing times for selected configurations of
	exhaustive Rank-S and Taily methods at the last point on each of the
	corresponding curves in Figure 17 Columns display the both 75th ooth
	porcentiles the mean and the maximum value of individual query pro-
	cossing times, where time is measured in simulation milliseconds. In these
	M = 2 Recall that the simulation tuned to model the
	experiments, $M = 2$. Recall that the simulation was turled to model the
	costs associated with muri, a research search engine, a commercial system
T-1-10	Result learning of the arrange of the last and the last a single
Table 18	breakdown of the average time spent processing queries, at the last point
	on each of the corresponding curves in Figure 17. Central queue indicates
	the percentage of time spent in the central query queue; network indicates
	the delays caused by network congestion; internal queues indicates time
	spent in system queues such as a machine's shard search queue or merge
	queue; resource selection indicates time spent performing resource selection;
	and <i>merge</i> is time spent performing the final merge operation. The remain-
	ing time not included in these categories was spent in shard search. 69
Table 19	Average load _i and range (max load _i – min load _i), where load _i is the time-
	averaged CPU load of m _i during the simulation, for two shard allocation
	policies and a range of query arrival rates, using Taily resource selection,
	M = 4, $B = 1$, and $S = 4$. Results for Gov2 using MQT _W Test queries are
	also shown in Figure 21. 73
Table 20	Average load _i and range (max load _i – min load _i) of shard search machine
	utilization as the training data ages, using the Log-based shard allocation
	policy. Other simulation settings are as described in Table 19. Test queries
	begin immediately after the training queries. Test _W and Test _M queries be-
	gin from 1 week and 1 month after the training queries, respectively. The
	MOT gueries do not have timestamps and so were not used in this experi-
	ment. 74
	Т Т

- Table 21Effectiveness of the top submitted TREC runs compared to the runs created
from fusing the top 10 submitted runs for each query set with SlideFuse-
MAP [Anava et al. 2016].90
- Table 22 Comparison of AUReC and two baseline metrics against the evaluation of end-to-end (EtE) systems with various resource selection algorithms in the CW09-B dataset. The end-to-end systems used P@1000 to measure the effectiveness of the shard maps. r is Pearson's correlation. Optimal cluster effectiveness is a commonly used heuristic in prior work that is identical to an end-to-end system evaluation using RBR k = 1. The rank-biased overlap baseline uses the the rank-biased overlap (p = 0.999) score of an RBR k = 3 selective search system against exhaustive search.
- Table 23Comparison of AUReC against the evaluation of end-to-end (EtE) systems
with various resource selection algorithms in the Gov2 dataset. The end-to-
end systems used P@1000 to measure the effectiveness of the shard maps. r
is Pearson's correlation. Optimal cluster effectiveness is a commonly used
heuristic in prior work that is identical to an end-to-end system evaluation
using RBR k = 1. The rank-biased overlap baseline uses the the rank-biased
overlap (p = 0.999) score of an RBR k = 3 selective search system against
exhaustive search.
- Table 24Comparison of AUReC and two baseline metrics against a Rank-S end-to-
end system at different rank depths in the CW09-B dataset.97
- Table 25Comparison of AUReC and two baseline metrics against a Rank-S end-to-
end system at different rank depths in the Gov2 dataset.98
- Table 26Comparison of AUReC scores and Rank-S end-to-end system evaluation
using P@1000, when D_q was generated from weaker rankers.99
- Table 27Comparison of AUReC and Rank-S end-to-end system evaluation using
P@1000, when using varying number of MQT queries. The end-to-end sys-
tem was evaluated with TREC queries, as usual.101

INTRODUCTION

Modern proliferation of data has increased the importance of good information retrieval. Cheaper data storage means that the data collections that must be searched are often large. For example, ClueWebo9, a dataset used for information retrieval research, is composed of a billion web pages and is 5 terabytes in compressed size. Searching large collections requires substantial computational resources. Compounding the problem is the increasing expectation from users that searches return results in seconds or less. Large web search engines typically have sub-second response times.

Producing search results in a timely fashion from large data collections is a challenging task. While a single large, very powerful machine may be able to handle searches of a moderate data collection, these machines are expensive. In addition, it is difficult to scale up this architecture to handle larger collections. Such a hardware setup also has a single point of failure; if the large machine goes offline, the entire search system becomes non-functional.

A better solution to the problem of large-scale search is a *distributed search* architecture, which is commonly used by commercial search engines such as Google or Microsoft's Bing. Distributed search is a divide and conquer approach which pools the resources of multiple machines and is set up as follows.

First, the large data collection is divided into many partitions, usually referred to as *shards*. Shards are typically equal in size. The shards are then distributed equally to the commodity computers so that each computer is only responsible for searching its assigned shard. When a user issues a query to the system, all shards of the data collection are searched in parallel by the many machines. The results of each individual shard are then merged and returned to the user. Figure 1a illustrates this process.

Distributed search has several advantages over a monolithic large computer set-up. By splitting the large collection into small shards, cheap computers are able complete their part quickly and greatly reduce the overall time it takes for the system to respond to the user. This enables the search system to achieve the low latency expected by users. In addition, when a single machine fails, most of the system still remains active and the failed component is easier to replace.

Distributed search is an effective architecture for large organizations that have access to many machines. However, because it searches the entire collection for every query, it continues to have a high total computational cost.

Recently, researchers have begun exploring a new search architecture that significantly reduces the computational cost of large-scale search: *selective search* [Kulkarni 2013]. Selective search is similar to a traditional distributed search architecture in that a large collection is divided into small shards and spread across multiple machines. However, it is distinguished from traditional distributed search in a few major ways.



Figure 1: Illustration of two different search architectures. Each box represents a machine and the cylinders represent parts of the search engine index ("shards"). The dark cylinders are shards that are searched; traditional distributed search always searches all shards while selective search searches only a few for a given query. Selective search also requires an additional resource selection database (grey hashing) in order to select the shards to search. Both architectures require a merge step (triangle) that merges the partial result lists from the shards.

The *Cluster Hypothesis* [van Rijsbergen 1979] states that documents that are similar in content tend to be relevant to the same kind of requests. Selective search uses this idea and partitions the data collection such that documents about similar topics are grouped together. For example, newspaper articles covering elections, forum pages dedicated to discussing politics, and web pages of various politicians may be placed together in a single shard that represents the 'politics' topic. Similarly, topical shards may be formed for sports, fashion, etc. In prior research, this topical grouping of the collection was achieved by the K-means clustering algorithm. Once created, these shards are distributed to the available computers.

According to the Cluster Hypothesis, with the topical shards of selective search, it is likely that the documents most relevant to a user's request reside in a few shards. Therefore, in contrast to randomly-partitioned distributed search, not all shards are searched when a user issues a query to a selective search system.

A selective search system uses a *resource selection* algorithm to identify these useful shards. A resource selection algorithm takes a user query and creates a ranked list of the shards it believes are most useful. The selective search system then only searches the shards indicated by the resource selection algorithm. The results of the shards are merged and returned to the user as traditional. The entire pipeline is illustrated in Figure 1b.

A concern of searching only a few shards is the possibility that selective search may miss documents that are relevant to the request and thus become less accurate than searching the entire collection in an exhaustive fashion. However, it was found by Kulkarni [2013] that only a very small percentage of shards was typically needed to achieve similar accuracy to searching the entire collection. This resulted in reducing the computational cost of large-scale search by more than 90%. Compared to a traditional distributed search architecture, selective search is able to return results that are as accurate and as timely, with a fraction of the resources.

Despite the advantages of selective search, the architecture has not yet seen wide acceptance in the academic community or in industry. As a newly introduced architecture, there are important questions about its performance characteristics in a practical setting.

One major concern is whether selective search would be accurate and efficient in various conditions common in the real world. First, trust in the prior results of selective search must be strengthened. Initial work in selective search [Kulkarni and Callan 2010a; Kulkarni et al. 2012] used only one partition per dataset under one specific experimental methodology and it is unknown what the variance of different selective search instances is over different possible collection partitions and experimental conditions. In addition, practical large-scale systems are commonly multi-stage pipelines including other optimizations and re-ranking components with parallel query processing. In such an environment, selective search would likely be deployed as the fast, first-stage retrieval step which can quickly return a list of candidate results. However, the behaviour of selective search in this setting in combination with other components has not been studied in depth.

Another concern is that there is currently a lack of tools to test the efficacy of selective search without implementing an end-to-end system. There is a need for simple, low-cost tools that can help system administrators design and test the best selective search implementation for their needs.

1.1 CONTRIBUTIONS

The purpose of this thesis is to answer outstanding concerns about selective search through thorough analyses and to create novel solutions where the initial approach falls short. It aims to bring selective search into wider acceptance by tackling two goals. First, it provides stronger assurances about the efficiency and effectiveness of selective search in real world settings and solutions to problems that may exist when implementing selective search as a first-stage retrieval system in a practical system. Secondly, this dissertation research provides convenient tools for system administrators to evaluate a selective search system without full implementation, in order to lower the barrier of entry.

1.1.1 Variance of random processes

Selective search contains non-deterministic processes. In the implementation introduced by Kulkarni and Callan [2010a], there are up to three random decisions, two during shard formation and another during resource selection. They used a sampled K-means algorithm to form the topic shards and randomly samples the collection to form the smaller sub-collection and then randomly assigns the initial cluster seeds. An additional source of non-determinism may be present in the resource selection step; sample-based algorithms such as Rank-S [Kulkarni et al. 2012] search a small, random sample of the total collection.

In most prior work [Kulkarni and Callan 2010a,b; Kulkarni et al. 2012; Kulkarni and Callan 2015], accuracy and efficiency figures were reported for one instance of a collection partition and one instance of a resource selection database. Due to the randomness involved, it is possible that the performance of selective search was an outlier and a different instantiation of selective search may perform differently. An in-depth exploration of the effects of the random processes is necessary as it is unclear how each individual random decision affects the final results. Chapter 3 answers the research question:

RQ 1 What is the effect of random decisions on selective search accuracy?

Chapter 3 investigates two major sources of variance in selective search: shard formation and resource selection. One cause of variance was due to poor placement of relevant documents during the partitioning process. These critical documents were sometimes placed in shards that contained dissimilar documents. This suggests that better partitioning may reduce variance. When comparing resource selection methods, two state-of-the-art algorithms, Taily [Aly et al. 2013] and Rank-S [Kulkarni et al. 2012] produced nearly equal variance despite the fact that Rank-S contains an additional random step. When the variance in performances were broken down in a query-by-query fashion, it was discovered that most of the variance of the overall accuracy was the product of a few queries, typically containing rare terms, suggesting that query type may have an impact on variance.

Overall, selective search was stable for most queries on precision-oriented metrics. However, there was higher variance for a small number of rare queries and for recall-oriented metrics, and improvements could be made, especially in the partitioning process.

1.1.2 Performance with dynamic pruning

A criticism of selective search was that it may not retain its efficiency advantage over exhaustive search when used with common dynamic postings pruning algorithms. Kulkarni [2013] briefly explored selective search in combination with the term-bounded max-score algorithm, but a deeper investigation with other algorithms had not been accomplished.

Weighted AND (WAND) [Broder et al. 2003] is a low-level search optimization algorithm that enables the system to avoid unnecessary evaluation of documents that would not be highly-ranked in the final result list. By skipping reading and computing scores for these documents, the cost of search is significantly reduced.

Because selective search skews the topical distribution of documents deliberately, it is unknown whether techniques such as WAND will have increased or decreased effects when combined with selective search. Selective search and WAND create efficiency gains by skipping the evaluation of documents. It is possible that the documents which are not normally evaluated by selective search are the very same documents that WAND would skip. In this scenario, there would be no additive efficiency gain by using selective search if one already uses WAND in their system. Chapter 4 investigates following research question:

RQ 2 Does dynamic pruning improve selective search, and if so, why?

Chapter 4 alleviates this concern by experimenting with a combination of selective search and WAND. It was found that the effectiveness of WAND is enhanced on the top-ranking topic shards, suggesting that grouping similar documents is helpful for WAND. Overall, selective search and WAND were shown to be orthogonal optimization techniques creating independent gains, due to the long posting lists generated by topic clustering; together they create *better*-than-additive efficiency gains.

Chapter 4 also experimented with sequential shard evaluation with WAND, where the selected shards were searched one-at-a-time in sequence rather than searching them simultaneously in parallel. This answers the following question:

RQ 3 *Can the efficiency of selective search be improved further using a cascaded pruning threshold during shard search?*

By keeping track of and passing on the internal evaluation state of WAND from one shard to the next, the total documents evaluated were significantly decreased. This demonstrated the possibility of using tiered retrieval to further decrease the computational cost of search.

1.1.3 Resource selection

In a practical large-scale system that uses a multi-stage retrieval architecture, selective search is most likely to be deployed as the fast first-stage step. In order for this to be feasible, selective search must be able to supply high recall results such that downstream re-rankers are able to see the relevant documents.

One of the most critical components in selective search accuracy is the resource selection algorithm. While existing methods perform well on precision-oriented metrics such as Precision at 10, they fall short of exhaustive search in recall-oriented metrics such as Mean Average Precision [Kulkarni and Callan 2015; Aly et al. 2013]. This dissertation presents two new solutions for resource selection: a light-weight method that uses pre-existing statistics, and a more sophisticated method that has higher accuracy.

1.1.3.1 Light-weight resource selection

Most resource selection algorithms require building an external database. Term-based algorithms such as Taily [Aly et al. 2013] collect corpus statistics and build a database containing model parameters for each term. Sample-based algorithms like Rank-S [Kulkarni et al. 2012] require building an index containing sampled documents from the full collection. This process can be laborious for large collections. Many search systems come with optimization methods which sometimes require similar analyses of the corpus and information for their optimization algorithm. For example, WAND stores the maximum score of each term. It was hypothesized that the information collected could be re-used to produce an effective, light-weight resource selection method that eliminates the need for additional corpus analyses and separate external databases.

Block-Max WAND [Dimopoulos et al. 2013], a variant of the WAND algorithm, collects statistics in finer granularity than WAND to perform dynamic postings pruning more efficiently. In Chapter 5, the block max scores collected by Block-Max WAND were utilized to perform resource selection, answering the following question:

RQ 4 Can BM-WAND information be used to design an effective resource selection algorithm?

When used for single-term queries and searching an equivalent number of shards to two stateof-the-art resource selection algorithms, it was found that the methods based on the block max scores were comparable to exhaustive search with statistical significance, whereas prior state-ofthe-art methods were not (Chapter 7) [Aly et al. 2013]. Furthermore, it was possible to produce a set of shards that would create a result set identical to exhaustive search that was close to the size of an oracle-derived set. This work suggests that it is feasible to produce a good shard ranking from the statistics collected by Block-Max WAND and furthermore creates the possibility of creating a stable variant of selective search which closely reproduces the results of exhaustive search.

1.1.3.2 Accurate resource selection

While Kulkarni and Callan [2015] showed that selective search has comparable accuracy to exhaustive search, it was based on a single partitioning of each of the two data sets and the results were reported as a single aggregated mean over four query sets. When a different partitioning and experimental methodology is used, a more complete picture selective search emerges. By presenting the results of the four query sets separately it was found that selective search performed well in a some query sets and poorly in others. Averaging the results across all query sets masked this issue. In the poorly performing query sets, resource selection was a major source of error in the system.

It was also found by Dai et al. [2016] that selective search performs relatively worse for recalloriented metrics than for precision-oriented metrics. This motivated additional research into resource selection that is more accurate, especially for deeper metrics such as MAP evaluated to a rank depth of 1000 (MAP@1000).

Chapter 7 presents a learning to rank based approach to resource selection. The advantage of using a learning to rank framework as opposed to multiple binary classifiers [Arguello et al. 2009b; Cetintas et al. 2009] is that it is better suited to selective search, where the number of resources to search may number in hundreds. Training hundreds of classifiers can take significant computational resources. Chapter 6 studies the new resource selection approach to answer the following questions:

RQ 5 *Is learning to rank resources an efficient algorithm that is as accurate as exhaustive search in deep recall-oriented metrics?*

RQ 6 Can learning to rank resources be trained without human judgements?

Chapter 6 describes new features suitable for resource selection and show that human-judged relevance data is not necessary for training the model. By assuming that selective search wants to mirror exhaustive search (at a lower computational cost), we can generate training data by searching the entire corpus and using the exhaustive search results as gold standard.

The new resource selection algorithm is highly accurate and is statistically non-inferior to exhaustive search even with MAP@1000. In addition, even when the expensive features that require a retrieval from a sample index are removed, the remaining fast features are sufficient to achieve good MAP scores. Learning to rank resources is an efficient and effective method and advances the state-of-the-art in selective search resource selection.

1.1.4 Performance in parallel processing

Aly et al. [2013] demonstrated the efficiency of selective search in an environment where search requests were processed one at a time, i.e. a new query would be processed by the system only when the old is completed and the results returned to the user. However, in a practical, large-scale selective search system with multiple available machines, it is desirable to process search requests in parallel in order to maximize the usage of available computational resources. Therefore, selective search efficiency in parallel environments must also be considered.

A concern that arises from a system which handles parallel requests is a potential *load imbalance* caused by the unequal popularity of shards. Because selective search shards are topically focused, it is likely that some shards may be more popular than others. For example, a shard with a

sports focus may be selected more frequently than a shard about information retrieval research. Difference in topic popularity can lead to an uneven computational load amongst the machines hosting the shards and reduce system throughput. Note that randomly-partitioned distributed search does not have this potential load balance problem because all shards are searched for every query and due to the random distribution of documents to shards, each shard is expected to do the same amount of work.

One contribution of this work is a software simulator which was used to conduct analyses on different configurations of selective search, including the parallel processing environment described above. The simulator was tuned to model the behavior of selective search under many different hardware configurations. The simulator enabled the exploration of a wide range of system configurations that would have otherwise been infeasible due to time and resource constraints. In particular, it is flexible enough to model selective search architectures as well such as traditional distributed search.

The simulator is a valuable tool for those seeking to evaluate or implement a selective search system as their search architecture. Due to the parameterization of the simulation, a few quick measurements of a computing environment can produce an immediate comparison between an existing system and a selective search implementation. It also allows the implementor to proto-type and evaluate variations of selective search configurations without the expense of creating multiple real implementations. Using the simulator, various configurations of selective search are tested to answer the following research questions:

RQ 7 Is selective search more efficient than exhaustive search in a parallel query processing environment?

RQ 8 How does the choice of resource selection algorithm affect throughput and load distribution in selective search, and how can any imbalances originating from resource selection be overcome?

RQ 9 How do different methods of allocating shards to machines affect throughput and load distribution across machines?

RQ 10 Does selective search scale efficiently when adding more machines and/or shard replicas?

Chapter 7 presents the results of the detailed analyses of the performance characteristic of selective search under various configurations. Using the simulator and an improved experimental methodology, the efficiency improvements reported by prior work were validated under more realistic conditions with some caveats. We found that selective search may need to search more shards than previously reported in order to achieve accuracy equivalent to searching the entire collection. Also, the experiments revealed that load imbalances are present in a naive implementation of selective search due to unequal shard popularity and the cost of resource selection, particularly for sample-based algorithms. However, these can be corrected by using training queries to guide the layout of shards.

Further experiments showed that the result of the training queries are durable; the configuration performed efficiently even a month later. Additional experiments with different hardware configurations indicated that selective search can be a competitive architecture even at a larger scale with many machines. The results of this work asserted the efficiency of selective search over traditional distributed search at both small and larger scales.

1.1.5 Partition evaluation

There are many ways to partition a document collection for selective search to create *shard maps*, a mapping of documents to shards. Prior work has used K-means, source-based clusters [Kulkarni and Callan 2010a], and query-biased partition schemes [Dai et al. 2016]. The effectiveness of the shard map was evaluated by running queries with relevance judgements on an end-to-end selective search system using the shard map. The accuracy of the final results of the end-to-end system using traditional IR metrics was used as proxy for the effectiveness of a shard map.

This method has a few issues. To evaluate an end-to-end selective search system, a resource selection algorithm must be used. Given that resource selection has a large impact on the accuracy of selective search, different choices of resource selection algorithms and their parameters may impact the accuracy results more so than the shard map. In addition, resource selection requires additional corpus analyses and the creation of resource selection databases (term-based methods) or centralized sample indexes (sample-based methods). To do so for many different shard maps is slow and cumbersome and makes evaluating many shard maps difficult.

This dissertation presents Area Under Recall Curve (AUReC), a new metric for evaluating shard maps. For a given query, based on a desired set of documents that should be retrieved for that query, AUReC builds a curve describing the relationship between the amount of shards searched to the maximum possible recall of the desired set of documents from the searched shards. The area under this curve is the effectiveness of a shard map. The set of desired documents is built from exhaustive retrieval, thus AUReC does not require an end-to-end evaluation of selective search or manual gold-standard judgements. Because the shards are ranked based on the overlap with the desired documents and the area under the curve considers the full set of shards, AUReC does not require choosing and implementing a resource selection algorithm and does not have parameters to set or train. Chapter 8 studies AUReC in detail to answer the following research questions:

RQ 11 Can AUReC generate a relative ordering of multiple shard maps that is consistent with orderings generated by end-to-end system evaluations?

RQ 12 *Is AUReC consistent with end-to-end system evaluations on whether differences between two shard maps are statistically significant?*

RQ 13 Is AUReC robust when compared to end-to-end systems using different resource selection algorithms; compared to a variety of IR metrics at different retrieval depths; when the search engine generating the top-k retrieval is weaker; and when it is calculated with a different set of queries than the end-to-end evaluation queries?

Experiments show that AUReC is highly correlated with the end-to-end evaluation of a full selective search system using a variety of resource selection algorithms, while being easier to implement and computationally inexpensive. AUReC is also a robust, stable method, and better able to distinguish finer differences between shard maps than end-to-end evaluation due to having more data points; human-created relevance assessments can be sparse. Experimentation show that AUReC is high correlated with several traditional IR metrics including precision, MAP and NDCG at different depths. Finally, when the sample retrieval was generated from different

queries than the relevance assessed evaluation queries for end-to-end systems, the resulting AU-ReC scores remained highly correlated. More queries produced better correlated results, with diminishing returns.

AUReC is a robust, stable metric that enables quick, easy comparisons of a large number of shard maps. This allows system administrators to quickly identify the best set up for their selective search systems. In addition, it supplies researchers with an optimization function for generating shard maps, empowering research in more efficient and accurate topic-based partitioning of large copora.

1.2 WIDER IMPACT

This dissertation demonstrates that selective search works well in realistic, large-scale production environments as well as in smaller lab settings. In addition, the dissertation provides tools to prototype and evaluate selective search without implementing the full system, paving the way for easier adoption of the system. The adoption of selective search benefits both industry and academia alike. In industry, selective search is a cost and energy saving alternative to traditional distributed search. In academia, the low computational cost of large-scale search delivered by selective search promotes not only information retrieval research, but also other research fields that often use search as a basic step in a more complex task, such as text mining and question answering tasks.

1.3 FUTURE DIRECTIONS

For wider acceptance of selective search, two research directions are necessary. First, selective search must confirm that it is able to easily and successfully fit into existing search and text analysis systems. Second, selective search must be expanded to handle other important domains beyond web documents. These directions are suggested in brief below.

1.3.1 Selective search in pipelines

The work in this dissertation has given confidence that selective search has sufficient recall to be used as a first-stage retrieval system in more complex text analysis pipelines. The logical next step is continued research into how selective search can fit into and benefit existing text pipeline systems. First, a detailed investigation must be conducted on how selective search changes the efficiency and accuracy of a complex pipeline which may contain other components such as caching, tiering, and machine learning re-rankers. Then, by studying the interactions between these components, we may be able to better tailor selective search to improve the entire pipeline. For example, being able to control the recall of selective search with strict guarantees would be useful, as is being able to push certain filtering tasks up to the selective search level for less work to do downstream in expensive, complex components.

1.3.2 Selective search in hierarchical corpora

While selective search research has been conducted with web corpora, structured, hierarchical data is common, especially in business-critical applications such as product search in e-commerce. Extending selective search beyond the limited bag-of-words model and using the additional in-

10 INTRODUCTION

formation to inform its clustering, resource selection, and result merging would better adapt the architecture to these domains and ideally increase efficiency and effectiveness.

1.3.3 Selective search in streaming datasets

Rapidly changing, online data sources such as Twitter and Instagram are becoming an increasingly important source of information on the web. The high volumes and the varied topics seen in such streams make it ideal for selective search. These domains have unique challenges that must be addressed for selective search to be effective. The system must be able to handle constant updates, including assigning them to the correct shard. It also needs detect and respond to fast changes in topic and queries, dynamically load balancing as necessary. Finally, selective search must be extended to be aware of the temporal nature of this domain, which has a large impact on its usage and accuracy in search.

1.4 ORGANIZATION

This dissertation presents the results of research done to answer questions on selective search effectiveness and efficiency and outlines further research to be done on selective search. Chapter 2 outlines related work. The dissertation begins with stronger assurances of selective search effectiveness and efficiency under different conditions. Chapter 3 presents an exploration of the random decisions in selective search and their effect. Chapter 4 investigates the interactions between selective search and WAND optimization techniques. Then, the shortcoming of selective search recall is addressed through developing stronger resource selection algorithms, allowing selective search to be feasibly used as a first-stage retrieval system. Chapter 5 explores the possibility of using statistics collected by Block-Max WAND to perform resource selection. Chapter 6 presents learning to rank resources, a new resource selection method. Finally, two powerful tools to evaluate selective search and provides a simulator tool for modelling selective search. Chapter 8 presents AUReC, a new metric for quickly evaluating shard maps. Chapter 9 concludes, summarizes the contributions and impact of the dissertation, and outlines future directions of research.

RELATED WORK

Selective search draws from prior work in a broad range of research areas including traditional distributed search, cluster-based retrieval, and federated search research.

2.1 TRADITIONAL DISTRIBUTED SEARCH

Modern large-scale retrieval systems often search corpora that are much larger than what one machine can fit. Therefore, by necessity, large-scale search engines have a distributed architecture in which the document collection is partitioned, and each partition is assigned to a distinct machine. A collection may be partitioned by terms or by documents.

In *term-based index partitioning*, each partial index is responsible for a non-overlapping subset of the terms in the vocabulary [Moffat et al. 2006; Lucchese et al. 2007; Zhang and Suel 2007; Cambazoglu et al. 2013]. When the collection is searched, only indexes that contain the query terms are searched. Because queries are typically short, only a few of the term indexes are required for each query, allowing multiple queries to be evaluated in parallel. This style of index has largely fallen out of favor because it is prone to load imbalances [Moffat et al. 2007]. Cambazoglu et al. [2013] provide more details of term-based partitioning approaches.

In *document-based partitioning*, each partial index is responsible for a non-overlapping subset of the documents in the collection. There are two major approaches to creating the document subsets: tiering and sharding.

Tiering creates partitions that have different priorities in order to reduce the total computational cost of search. Initially, only the top priority partition, or the top *tier*, is searched. If the results are deemed insufficient, lower tiers are searched as necessary. Many different strategies exist for deciding when to cascade to lower tiers [Risvik et al. 2003; Baeza-Yates et al. 2009b; Cambazoglu et al. 2010; Brefeld et al. 2011]. Tiers can be defined based on various document characteristics. Geographically defined tiers place documents that are local to the area in physically closer machines [Cambazoglu et al. 2010; Brefeld et al. 2011]. One can also create tiers using estimates of the quality of documents, which are calculated a priori using query independent features such as spam scores or term statistics [Risvik et al. 2003]. Historical query logs can be used to create tiers of the most popularly retrieved documents [Risvik et al. 2003]. In tiering, the order in which the tiers are searched is the same across all queries.

The alternative approach, *sharding*, creates partitions or *shards* that all have the same priority. Sharding is used to pool the resources of many machines to divide-and-conquer the costs of search in order to minimize latency. For each query, all shards are searched in parallel [Puppin et al. 2006; Badue et al. 2007]. Documents are usually assigned to shards by source, randomly, or in a round-robin approach. Typically, it is desirable for all shards to have similar characteristics in order to minimize load imbalances.

Tiering improves throughput and sharding reduces latency; thus, these complementary methods can be combined to improve in both areas. For example, the corpus might be divided into tiers based on document popularity or authority, and then each tier divided into shards, with the shards distributed across a cluster of computers for lower latency search within that tier [Orlando et al. 2001; Barroso et al. 2003; Baeza-Yates et al. 2007a, 2009a; Brefeld et al. 2011; Francès et al. 2014].

A range of work has explored the efficiency of sharded search systems, covering topics including reducing the communications and merging costs when large numbers of shards are searched [Cacheda et al. 2007b]; load balancing in mirrored systems [Macdonald et al. 2012; Freire et al. 2013]; query shedding under high load to improve overall throughput [Broccolo et al. 2013]; and query pruning to improve efficiency [Tonellotto et al. 2013]. Other work focuses on addressing the load imbalances that arise when term-based indexes are used and selective replication of frequently-accessed elements [Moffat et al. 2006, 2007]. A common theme is that when a tier is searched, *all* of its shards are searched, that is, an *exhaustive* search strategy is employed.

2.2 CLUSTER-BASED RETRIEVAL

Selective search is a type of *cluster-based* retrieval architecture [Croft 1980; Voorhees 1985; Griffiths et al. 1986; Willett 1988]. Cluster-based retrieval systems organize the corpus into hierarchical or flat clusters during indexing. When a query is received, clusters are selected by traversing the hierarchy or comparing the query to cluster centroids to identify the most similar cluster. Small-scale systems may return all of the documents in selected clusters; larger-scale systems rank documents in the selected clusters and return just the highest-scoring documents.

Classic cluster-based retrieval systems produce many small clusters. For example, Can et al. [2004] used 1,640 clusters that each contained an average of 128 documents. When clusters are small, many must be selected to maintain acceptable accuracy. Can et al. [2004] searched 10% of the best-matching clusters (164 clusters), a heuristic threshold that was also used in earlier investigations. Similarly, Croft [1980] used over 400 clusters for 1400 documents.

Selective search systems produce and search a smaller number of large clusters. For example, Kulkarni [2013] and Aly et al. [2013] used clusters that contained approximately 500,000 documents, and queries typically searched 3-5 clusters.

Classic cluster-based retrieval systems also used a single index to store the entire corpus. Efficiency improvements are obtained by storing cluster membership information in inverted list data structures and grouping postings by cluster membership so that large portions of an inverted list may be skipped during query processing [Can et al. 2004; Altingovde et al. 2008]. This architecture must bear the I/O costs of reading complete inverted lists, and the computational costs of processing them (albeit, efficiently). Can et al. [2004] note that storing each cluster in its own index would reduce computational costs, but incur prohibitive I/O costs (primarily disk seeks) due to the large number of clusters selected. The selective search architecture stores each cluster in its own index, but avoids the I/O costs by selecting a very small number of clusters for each query.

There are other differences between classic cluster-based retrieval and selective search. For example, selective search uses more sophisticated methods of determining which clusters match each query. Kulkarni and Callan [2015] provide a detailed description of cluster-based retrieval and its relationship to selective search.

2.3 FEDERATED SEARCH

Sharded indexes and their search engines are a special case of federated search systems. Typically, *federated search systems* integrate multiple collections or search systems. Common types of federated search systems include aggregated or vertical search engines, peer-to-peer search networks, and meta-search engines [Shokouhi and Si 2011]. Vertical search combines the results of specialized search engines for different domains or media (e.g. images, maps, news) to provide a single unified search portal [Arguello et al. 2009b; Arguello 2017; Aliannejadi et al. 2018]. In peer-to-peer search networks, each peer node contains some subset of documents and search is conducted by forwarding queries to appropriate nodes [Lu and Callan 2006]. Finally, meta search engines do not index index any documents, but combine the results of multiple search engines controlled by different organizations by forwarding a query to the various systems and merging the results appropriately [Selberg and Etzioni 1997; Salampasis 2017].

A common task shared by federated search systems is determining to which of the underlying search services should a given query be forwarded. Normally the goal of federation is to send queries to as few of the underlying search services as possible, so a *resource selection* algorithm is used.

Three types of resource selection have been proposed: term-based, sample-based, and classification-based algorithms.

Term-based algorithms treat each search service as a bag of words. Common document ranking algorithms can be adapted to the task of ranking resources or services; GlOSS [Gravano et al. 1999], CORI [Callan 2000], and the query likelihood model [Si and Callan 2004a] are examples of this approach. Algorithms developed specifically for resource ranking usually model the distribution of vocabulary across search services [Yuwono and Lee 1997; Hawking and Thistlewaite 1999; Aly et al. 2013; Zhang and Balog 2017]. Term-based algorithms typically only support bag-of-words queries, but a few also support corpus-level or cluster-level preferences, or Boolean constraints [Gravano et al. 1999; Xu and Croft 1999; Callan 2000; Liu and Croft 2004].

Sample-based algorithms represent each search service using a sample of its contents. Samples from all services are combined into a *centralized sample index*, or *CSI*. When a query is received, the CSI is searched, and each top-ranked document found in the CSI is treated as a vote for the resource from which it was sampled. Many different methods for weighting votes from different resources have been described [Si and Callan 2003, 2004b, 2005; Shokouhi 2007; Paltoglou et al. 2008; Thomas and Shokouhi 2009; Kulkarni et al. 2012; Zhang and Balog 2017].

Most of the term-based and sample-based algorithms are unsupervised and do not use external resources. Recently, query logs have been used to estimate resource popularity [Urak et al. 2018] and knowledge bases have been used to improve resource selection by enriching queries with entities [Han et al. 2018; Dragoni et al. 2017].

Classification-based algorithms take the use of additional resources further and represent each search service using a model learned from training data. The features used might include the presence of specific words, the scores of term-based and sample-based algorithms, and the similarity of the query to a resource-specific query log [Arguello et al. 2009a; Kang et al. 2012].

During most of the last decade, sample-based algorithms have been regarded as being a little more effective than term-based algorithms [Shokouhi and Si 2011]; however, recently Aly et al. [2013] argued that *Taily*, a new term-based algorithm, is more effective than the best sample-based algorithms. Term-based and sample-based algorithms are effective when the search engines are mostly homogeneous. Both types of algorithm are unsupervised, meaning that training data is not required. Supervised classification-based algorithms can be more effective than unsupervised

methods; however, their main advantage is their ability to select among heterogeneous resources (for example, "vertical" search engines), and exploit a wide range of evidence. Resource selection algorithms have also been applied to a variety of other tasks, including blog search and desktop search [Seo and Croft 2008; Elsas et al. 2008; Kim and Croft 2010].

The three resource selection algorithms commonly used with selective search are *ReDDE*, *Taily*, and *Rank-S*.

ReDDE, a sample-based algorithm, uses a centralized sample index (CSI) [Si and Callan 2003]. The query is run against the CSI and the documents retrieved from it act as votes towards the shards that they were sampled from. Let n be the set of documents retrieved from the CSI, and n_R represent the subset of n which were sampled from shard R. Then, the score s_R for shard R is:

 $s_R = n_R w_R$

where w_R is a shard weight calculated from the size of the shard and the size of its sample.

Rank-S, is another sample-based algorithm from the SHiRE family of algorithms [Kulkarni et al. 2012]. In Rank-S, the scores of documents retrieved from the CSI are decayed exponentially and then treated as votes for the shards the documents were sampled from. The exponentially-decayed vote of a document for its parent resource is computed as:

$$Vote(d) = Score_{CSI}(d) \times base^{-Rank_{CSI}(d)}$$

where $\text{Score}_{CSI}(d)$ and $\text{Rank}_{CSI}(d)$ are the document score and rank obtained by searching the CSI; and base is a configurable parameter. The scores of the documents naturally converge to zero due to the decay. Resources with a total score above 0.0001 are then selected, as described by Kulkarni et al. [2012].

Taily assumes that the distribution of document scores for a single query term is approximated by a Gamma distribution. The allocation algorithm uses two parameters, n_c and v, where n_c is roughly the depth of the final ranked list desired, and v is the number of documents in the top n_c that a resource must be estimated as contributing in order to be selected. Term scores are calculated from simple corpus statistics and fitted to a Gamma distribution for each shard-term pair. Taily's resource selection database records these fitted Gamma distributions for each term t in resource i, describing the term's score distribution in that shard. Gamma distributions are represented by two parameters, the shape parameter k_i^t and the scale parameter θ_i^t . At query time, the cumulative distribution function of the Gamma distribution is used to estimate the number of documents from each resource that will have a score above a threshold derived from n_c . Each resource that provides v or more documents is selected [Aly et al. 2013].

When a query is received, Taily looks up two floating point numbers for each index shard per query term, whereas Rank-S must retrieve an inverted list for each query term. Taily's computational costs are linear in the number of index shards |S|, and nearly identical for each query of length |q|. The computational costs for Rank-S vary depending on the size of the CSI and the document frequency (df) of each query term in the CSI. The Rank-S approach is more efficient only when df_t < |S| for query term t. For most applications Taily is substantially more economical than Rank-S [Aly et al. 2013].

2.4 SELECTIVE SEARCH

Selective search is a distributed search architecture that combines ideas from cluster-based retrieval and federated search to make large-scale search more computationally efficient [Kulkarni and

Callan 2010a,b; Kulkarni 2013; Kulkarni and Callan 2015]. A large collection comprised of D documents is partitioned into P *topic-based shards* and at query time, a *resource selection* algorithm is used to select $k \subset P$ shards that are most likely to contain relevant documents. The selected shards are then searched in parallel and their results are merged and returned. Typically, k < P << D and only a small portion of the total collection is searched for each query resulting in large savings in total computational effort.

Prior research established selective search as a viable architecture and demonstrated its advantages in small laboratory settings. Kulkarni and Callan [2010a] compared partitioning a collection into shards by topic against partitioning by source or partitioning randomly. They showed when documents are topically allocated into shards, queries were able to be accurately answered by searching only a few shards rather than searching every shard, introducing selective search as an architecture for the first time.

As a critical component of selective search, the shard selection process in selective search has received much research attention in past work. Kulkarni et al. [2012] noted that in the selective search system presented by Kulkarni and Callan [2010a], the ReDDE resource selection algorithm was used to rank the shards then the same shard cut-off (i.e. the number of shards to search) was applied for all queries. Kulkarni et al. [2012] presented the SHiRE family of sample-based resource selection algorithms which create query dependent, dynamic shard cut-offs. Rank-S, a commonly used algorithm in selective search, is a member of the SHiRE family. They demonstrated that accuracy and efficiency improvements can be realized by jointly formulating the shard ranking and cut-off estimation problem. Kulkarni et al. [2012] compared SHiRE algorithms against existing resource selection baselines and demonstrated that they are more effective and efficiency improvements over static cut-off system cut-off system cut-off for ground by SHiRE, which creates significant efficiency improvements over static cut-offs by searching fewer shards when possible.

Aly et al. [2013] presented a new term-based resource selection for selective search, Taily. They showed that sample-based resource algorithms such as ReDDE and the SHiRE family of algorithms can have significant efficiency costs because they require a retrieval from a centralized sample index (CSI), the cost of which is linear to the number of documents in the CSI that contain the query terms. Taily is term-based and thus the cost is bounded by the number of shards in the selective search system, which is typically much smaller than the cost incurred by a retrieval from the CSI. Aly et al. [2013] compared Taily against sample-based and term-based baselines and demonstrated that it is as effective as prior algorithms while being more efficient, especially in latency.

Chuang and Kulkarni [2017a] and Chuang and Kulkarni [2017b] presented variations of existing resource selection algorithms ReDDE and CORI that use bigrams and a learning-to-rank based method. The learning to rank method was combined with query expansion for single term queries using Wikipedia. They combined the shard ranking generated with three different dynamic shard cut-off estimators and saw consistent improvements in both early and deeper ranks of the result list, and were able to achieve accuracy equivalent to exhaustive search with MAP@1000. In addition, they concluded that if the relevant documents for a query are spread across less than 10% of the shards, selective search would be able balance precision and recall well. This work was published contemporaneously with Chapter 6.

Kulkarni [2015] focused on the shard cut-off estimation aspect of resource selection and developed ShRkC, a query and metric dependent cut-off estimator that reduces the search costs of a selective search system without degrading efficiency. The paper formulates the cut-off estimation as a supervised regression problem, using training queries to train a different regression model for each metric of interest. The trained models were used to estimate cut-offs for a ReDDE shard ranking and were able to search less shards on average while maintaining similar effectiveness as baseline methods.

Mohammad et al. [2018] studied shard cut-off estimation from a recall-oriented perspective and presented a supervised method to generate query-specific shard cut-offs using the similarity of the selective search results to exhaustive search results as a target for training. They found that selective search can produce 70% agreement with exhaustive search at rank 5,000 with only 16 - 18% of the computational cost, lending credence to the choice selective search as a good first-stage retrieval system.

Dai et al. [2016] studied shard creation, another component of selective search important for accuracy, and presented an improved partitioning method for creating topical shards. The QInit step clusters terms from training queries to initialize cluster seeds. During the clustering step, the QKLD similarity function places more weight on terms that are important in the training query logs when calculating document to centroid similarity. In combination, the QInit-QKLD method creates more evenly sized shards that are better aligned with the topics in the query traffic and deliver higher accuracy.

The efficiency of the selective search system as a whole has recently began receiving more attention. Kulkarni and Callan [2015] extended the work of Kulkarni and Callan [2010a] and conducted a more detailed examination of selective search as an architecture with stronger exhaustive search baselines that use a distributed search model. The paper presented a new variation of K-means, size-bounded sample-based K-means, that creates more evenly sized shards and studied its effects on selective search efficiency. The paper also studied how selective search behaves in conjunction with term-bounded max score, a dynamic optimization method; how it behaves with queries of different lengths; and its effectiveness when different sizes of CSI are used.

Chapter 6 of Kulkarni [2013] implemented a proof-of-concept selective search system over two machines and conducted experiments which confirmed the efficiency advantages of selective search over exhaustive search.

Dai and Callan [2018] investigated the impact of selective search on traditional list caching algorithms. They found that QtfDf, a popular static caching algorithm [Baeza-Yates et al. 2007b], is as effective on the topic-based shards of selective search as it is on randomly partitioned shards of traditional distributed search architectures. A mixed strategy of caching terms determined from both global and shard local term value estimations reduced total I/O without negatively impacting query hit rates.

Hafizoglu et al. [2017] presents a cluster-skipping index for selective search, in which topical clusters are sliced and randomly distributed across many shards. While the logic of selective search remains the same, this alternative physical layout evenly distributes the load of selective search and also reduces the query latency by up to 55%.

Overall, past work in selective search was largely motivated by increasing selective search accuracy and studying its efficiency in resource-scarce environments. Prior work successfully established selective search as a viable distributed search architecture and demonstrated that it can deliver exhaustive-equivalent accuracy in shallow metrics and produce significant efficiency gains over exhaustive search under small-scale laboratory settings for traditional web search. Extensions of selective search to different tasks is an emerging research area; Wang and Lin [2017] adapted selective search to a real-time streaming domain.

In Section 2.4.1 through Section 2.4.5, the experimental methodologies used by other authors in selective search are organized and elaborated in more detail to provide context for the research presented in this dissertation.

2.4.1 Datasets

The most common datasets used with selective search are the Gov2, ClueWebo9 Category A English (CW09-A) and ClueWebo9 Category B (CW09-B) datasets. Gov2 is a web collection containing 25 million web pages crawled in early 2004 from .gov domains¹. ClueWebo9 Category A English dataset is the English document subset of ClueWebo9 and is a web collection of 500 million web pages crawled in January and February of 2009. CW09-B is the first 50 million English documents of Category A².

The Gov2 dataset was used consistently by all prior selective search research and past research used queries provided by the TREC Terabyte Track 2004–2006, 150 topics with graded relevance judgements.

The CWo9-B dataset was consistently used by all prior work, excluding Kulkarni [2015]. The larger CWo9-A dataset was used by about half of past papers [Kulkarni and Callan 2010a, 2015; Kulkarni 2015; Aly et al. 2013]. While most papers used the datasets as provided, Aly et al. [2013] used the Waterloo Fusion spam scores [Cormack et al. 2011] to filter out documents whose spam scores were 50% or higher in CWo9-A, stating that the exhaustive search accuracy of the unfiltered dataset was too low to provide good signals about the efficacy of selective search. This assertion is supported by other papers which used the dataset, which report MAP values low as 0.06 [Kulkarni and Callan 2015; Kulkarni 2015]. CWo9-B does not share these problems as the first 50 million English documents contain many high quality documents from Wikipedia and spam filtering for higher effectiveness is not necessary.

The ClueWebo9 datasets were used with queries provided by the TREC Web Track. The number of queries used vary. Earlier papers used less of the query set, with Kulkarni and Callan [2010a] using 50 topics from the 2009 iteration of the Web Track. Kulkarni et al. [2012], Kulkarni et al. [2012] and Aly et al. [2013] used 100 topics from Web Track 2009–2010. Others [Kulkarni and Callan 2015; Kulkarni 2015; Dai et al. 2016] used 200 topics from 2009–2012. The queries used with ClueWeb09 are particularly important when interpreting the accuracy results because selective search does particularly well with the 2010 queries compared to exhaustive search, but does particularly poorly with the 2011 queries (Chapter 7) [Dai et al. 2016]. Thus, if a paper's query set contains the 2010 queries but not the queries from later years, the accuracy of selective search compared to exhaustive search may be misleading. In summary, the most modern practices with the ClueWeb09 dataset are to filter spam documents if using CW09-A and to use the full 200 topics from the 2009–2012 years of the TREC Web Track.

2.4.2 *Shards*

Selective search was used with a wide variety of shard maps, i.e. a mapping of documents to shards. Kulkarni and Callan [2010a] created the first set of shard maps created for selective search using a sampling-based variation of K-means. A small percentage of the dataset is randomly sampled then K-means is run on the sample using the symmetric version of negative Kullback-Liebler divergence as the similarity function. Afterwards, the rest of the dataset is projected to the nearest cluster centroid. The paper used a 0.1% sample for CW09-A and a 1% sample for Gov2 and CW09-B. K was set such that the shards contain about 500,000 documents on average,

¹ http://ir.dcs.gla.ac.uk/test_collections/gov2-summary.htm

² ttps://www.lemurproject.org/clueweb09.php

equating to 50 shards for Gov2, 100 shards for CW09-B and 1,000 shards for CW09-A. This shard map was used by several papers [Kulkarni et al. 2012; Aly et al. 2013; Kulkarni 2015].

Kulkarni [2013] noted a flaw of the above shard maps, which is that the shards were of very skewed sizes, with a majority of shards being much larger or smaller than the target average shard size of 500,000 documents. To create more evenly-sized shards, Kulkarni presented size bounded sampling-based K-means (SB² K-means). In SB² K-means, a *split* phase splits up large shards and a *merge* step combines small shards together. The process created 208 shards for Gov2, 92 shards for CW09-B, and 807 shards for CW09-A. Kulkarni and Callan [2015] presents the same process, but define 50 shards for Gov2 instead.

The most recent advance in selective search shard creation was presented in Dai et al. [2016] where historical query logs are used to inform the topics created in the clustering step. A note of interest is that the shard maps presented in this paper use smaller shards for Gov2 than prior work as they resulted in more accurate results.

This dissertation was developed in parallel to the development of improved selective search shard maps and uses different shard maps depending on the state-of-the-art at the time of the work. This process of deciding between many available shard maps partially inspired Chapter 8.

2.4.3 *Resource selection*

Resource selection has been a major focus in prior research, with many papers being devoted to more accurate or efficient resource selection [Kulkarni et al. 2012; Aly et al. 2013; Kulkarni 2015]. The two most commonly used resource selection algorithms in prior selective search research are ReDDE [Kulkarni and Callan 2010a, 2015; Kulkarni 2015; Kulkarni et al. 2012] and Rank-S [Aly et al. 2013; Dai et al. 2016]. Both methods share an important parameter, the size of the centralized sample index (CSI). Earlier papers used a larger CSI size, of up to 4% of the entire collection [Kulkarni et al. 2012], which delivers more accurate results. However, Aly et al. [2013] showed that the resource selection using such large CSIs can be costly and have argued for smaller CSIs. Later work [Kulkarni and Callan 2015; Kulkarni 2015; Dai et al. 2016] used a 0.5% sample for their CSI and this setting is followed in this dissertation. Of the two common methods, Rank-S is more modern and was shown to be more accurate and is used in this dissertation.

Aly et al. [2013] presented Taily and argued that it is more accurate and efficient than Rank-S. As Taily is term-based and represents a different family of resource selection algorithms than Rank-S, this dissertation uses both Taily and Rank-S in conjunction so that the results are broadly applicable.

2.4.4 Evaluation of accuracy

Selective search accuracy has been measured by scoring its result list using queries with relevance judgements and comparing these scores against exhaustive search baseline scores. Standard information retrieval metrics are used, such as Precision, Normalized Discounted Cumulative Gain (NDCG), and Mean Average Precision (MAP). Measured at lower ranks such as 10 or 30 (typically abbreviated as P@10, NDCG@30 etc.) the metrics measure precision. At deeper ranks (i.e. NDCG@100, MAP@1000 etc.), the metrics are recall-oriented. Because selective search only searches and returns a small portion of the index, it is expected to perform relatively better in precision-based metrics than recall-oriented metrics.

Selective search papers have demonstrated that it is as accurate as exhaustive search baselines, but there are a few caveats to this claim. The first selective search paper [Kulkarni and Callan 2010a] used precision-oriented metrics such as P@10, P@20, and P@30 to evaluate selective search, where it successfully met exhaustive search baselines. However, its performance in recall-oriented metrics was not measured. Later papers reported recall-oriented metrics such as MAP@1000 but often had difficulty meeting exhaustive search results, especially in Gov2 [Kulkarni and Callan 2015; Dai et al. 2016; Aly et al. 2013]. Finally, when comparing an exhaustive search system (that has a mean of μ_A) to a selective search system (that has a mean of μ_B), prior works use H₀ : $\mu_A = \mu_B$ (i.e. "these two systems have the same mean") as the null hypothesis. This is mathematically problematic; failing to reject the null hypothesis does not mean that the null hypothesis is true. That is, one cannot conclude that by failing a typical paired t-test or permutation test that selective search is equivalent to exhaustive search. A more theoretically sound alternative which uses the null hypothesis of H'₀ : $\mu_A - \mu_B \ge \delta$ is presented in Chapter 5.

2.4.5 Evaluation of efficiency

Studies of selective search efficiency determined computation cost in several different ways. Initially, Kulkarni and Callan [2010a] used the proportion of documents in the collection that were present in the selected shards as the cost, using this as an estimate of the total amount of work done by selective search. This idea was improved in Kulkarni et al. [2012], where rather than counting *all* the documents in selected shards, only documents that contained one of the query terms contributed to the cost.

Aly et al. [2013] observed that resource selection significantly added to the computation cost of selective search and added a term representing resource selection term to the cost equation. In addition, to the total cost metric, Aly et al. proposed an alternate metric that combined the cost of resource selection and cost of the most expensive shard. This loosely represents the lowest possible latency of selective search that can be achieved in a parallel setting if there was one machine for every selected shard and resource selection database. These metrics are used by Kulkarni and Callan [2015]; Dai et al. [2016]; Kulkarni [2015, 2013].

Kulkarni [2013] implemented a proof-of-concept selective search system that was distributed over two machines and measured the total execution time as the cost. This system was used to conduct the first preliminary explorations into the throughput of selective search under a parallel query processing environment. In addition, Kulkarni presented a further refinement to the cost metric by counting the total number of postings retrieved from disk rather than counting the number of documents. Kulkarni showed that this Cost-in-Postings metric is better correlated with actual execution times than the Cost-in-Documents used by most prior work, especially for larger datasets. This dissertation uses the Cost-in-Postings when measuring total computation cost. The details of measuring selective search efficiency under parallel environments are given in Chapter 7.
VARIANCE¹

For wider acceptance of selective search, some outstanding questions must be answered, related to possible issues of the architecture that could arise in a real world setting. This is the first of several chapters which will tackle those issues and offer assurances about the performance of selective search accuracy and effectiveness in common real world situations and present solutions to potential problems.

Prior research [Aly et al. 2013; Kulkarni 2013; Kulkarni et al. 2012] showed that selective search can be as accurate as a typical 'search all shards' distributed architecture but at a substantially reduced computational cost. However, it was based on a single partitioning of the corpus. This is a potential concern to a system designer implementing selective search; a different partitioning might yield different results because selective search contains several non-deterministic steps. Kulkarni [2013] used sample-based K-means clustering to create topical shards, which has random sampling and cluster seeds. Some resource selection algorithms, such as Rank-S [Kulkarni et al. 2012], use a random sample of the corpus to decide which shards to search for each query. The lack of knowledge of the possible variance in the performance of selective search systems reduces confidence in the architecture.

We are not the first to notice the potential problems that may stem from the non-determinism present in selective search. Jayasinghe, et al. [Jayasinghe et al. 2014] proposed a linear model to statistically compare non-deterministic retrieval systems, using selective search as an example. Their model took into consideration the multi-dimensional variance in selective search. They found that there could be significant differences between selective search and exhaustive search based on the collection sample used to build the CSI, especially at low sample rates. However, their work focused on testing whether selective search has equivalent mean effectiveness as a baseline search architecture. In our work, we focus on the variance itself and seek to answer the following research question.

RQ 1 What is the effect of random decisions on selective search accuracy?

We compare results obtained with different partitionings of two ClueWeb datasets and examine the variance of a system across a fixed set of four query sets (Section 3.2.1), and for individual queries (Section 3.2.2).

¹ This chapter is a lightly-revised version of Zhuyun Dai, Yubin Kim, and Jamie Callan. How random decisions affect selective distributed search. In *Proceedings of the 38th Annual International ACM SIGIR Conference on Research and Development in Information Retrieval*, pages 771–774, 2015. The initial investigation was started by Callan and the experiments were conducted by Dai. Kim joined the project soon after and guided the direction of the experiments and provided analytical insight. The paper writing was equally shared by the authors.

3.1 EXPERIMENTAL METHOD

We define a *system instance* to be a partition of a corpus index and its corresponding resource selection database for algorithms such as Rank-S [Kulkarni et al. 2012] or Taily [Aly et al. 2013]. Both Rank-S and Taily require that values be set for two system parameters. For Rank-S the two key parameters are the size of the sample for the centralized index (the CSI), and the quantity base used for the exponential discounting. Taily also requires two parameters to be specified: n, conceptually the depth of the result list used for estimating relevant document counts and ν , the score cut-off. The parameters used for Taily and Rank-S were suggested by Aly et al. [2013] and Kulkarni et al. [2012]. They were n = 400, ν = 50 for Taily and B = 5, C S I = 1% for Rank-S.

Our system instances were defined by a slight adaptation of a process defined by Kulkarni and Callan [2010a] also used by Aly et al. [2013]. First, 500K documents were randomly sampled from the corpus, K of those documents were selected to be seeds, and K-means clustering was used to form clusters. Second, the remaining documents were projected onto the most similar clusters to form index shards. Third, documents scoring lower than 50% in the Waterloo Fusion spam scores [Cormack et al. 2011] were removed and a resource selection index was constructed. The Rank-S resource selection index was formed from a 1% random sample of each cluster. The Taily resource selection index is created deterministically. Thus, each system instance involved 2-3 random processes.

The experiments also tested Relevance-Based Ranking (RBR), an oracle resource selection algorithm that ranks shards by the number of relevant documents that they contain. RBR makes it possible to distinguish between different types of variance. For query q, RBR searched the average number of shards that Rank-S and Taily searched for q.

Experiments were conducted on the ClueWebo9 Category A (*CW09-A*) corpus, which contains 500 million English web pages, and ClueWebo9 Category B (*CW09-B*), which contains 50 million English web pages. Queries were from the TREC 2009–2012 Web Tracks: 4 sets of 50 queries.

Parameters were set to produce shards of 500K documents on average. Each CW09-A system instance had 1000 shards; CW09-B system instances had 100 shards.

3.2 EXPERIMENT RESULTS

The effects of random decisions during partitioning and indexing can be measured across query sets or individual queries. Measuring across query sets provides information about the reliability of metrics such as MAP and NDCG that are typically reported in research papers. Measuring across individual queries provides information about the behavior that people will observe when they use the system. The experiments below provide both types of measurement.

3.2.1 Variability for query sets

The first experiment examined how random decisions affect the 'average case' metrics typically reported in research papers. It used three resource selection algorithms: Rank-S, which has a random component; Taily, which is deterministic; and RBR, the oracle algorithm. Ten random partitions were created for each dataset. Thus there were 2 datasets \times 10 random partitions \times 3 resource selection algorithms = 60 system instances. On average, Rank-S searched 3.91 CW09-A shards and 3.75 CW09-B shards; Taily searched 2.58 CW09-A shards and 2.53 CW09-B shards; and RBR searched 3.13 CW09-A shards and 2.80 CW09-B shards.



Figure 2: Accuracy distribution of Rank-S and RBR system instances for the Web Track queries. Mid line in boxes represents median. Outer edges of the boxes represent lower and upper quartiles, while whiskers indicate 1.5 interquartile range. + indicates outliers. Blue dot is the performance of the typical 'search all shards' baseline.

Retrieval accuracy was evaluated using three metrics: Precision at rank 10 (P@10), Normalized Discounted Cumulative Gain at rank 100 (NDCG@100), and Mean Average Precision at rank 500 (MAP@500) for compatibility with prior research [Aly et al. 2013]. These metrics help capture the performance of selective search at precision-oriented (P@10) tasks and more recall-oriented tasks (NDCG@100 and MAP@500).

The distributions of measurements over each type of system instance and the 'search all shards' baseline are shown in Figure 2. Table 1 and Table 2 report the standard deviation and variance coefficient of Rank-S and RBR system accuracy scores. Results for Taily and NDCG@100 were similar to results for Rank-S and other metrics. The standard deviation is a common measure, however it has two flaws: i) values produced by different metrics (e.g., P@10, MAP@500) cannot be compared because it is not dimensionless; and ii) the *importance* of a given level of deviation depends upon the value of the mean. The *variance coefficient* (VC) normalizes the standard deviation by the mean, thus eliminating these flaws.

The Relevance-Based Ranking (RBR) was generally more accurate and had lower variance across system instances than Rank-S and Taily, as expected. The one exception was the 2011 query set and the MAP@500 metric; Rank-S had lower variance than RBR under this condition. This exception was caused by a small set of queries. Rank-S had consistently poor accuracy on these queries, whereas RBR had greater accuracy but also greater variance.

Ora orașe C at	CWog	CW09-B		A
Query Set	SD	VC	SD	VC
Web Track 2009	2.22×10^{-2}	5.06%	2.00×10^{-2}	7.30%
Web Track 2010	2.11×10^{-2}	5.57%	1.50×10^{-2}	4.90%
Web Track 2011	1.52×10^{-2}	4.84%	1.51×10^{-2}	5.06%
Web Track 2012	1.79×10^{-2}	5.44%	1.54×10^{-2}	6.14%
	(a)	P@10		
Oregoring Cost	CW09-B		CW09-	A
Query Set	SD	VC	SD	VC
Web Track 2009	7.77×10^{-3}	10.04%	6.65×10^{-3}	10.65%
Web Track 2010	$7.56 imes 10^{-3}$	7.16%	$3.31 imes 10^{-3}$	3.66%
Web Track 2011	$8.52 imes 10^{-3}$	8.89%	7.67×10^{-3}	7.62%
Web Track 2012	7.17×10^{-3}	6.37%	7.96×10^{-3}	8.23%

Table 1: Standard Deviation (SD) and Variance Coefficient (VC) of accuracy scores for Rank-S instances.

(b) MAP@500

The experimental results suggest that there are two sources of variance in selective search: Partitioning, and resource selection. RBR always selects the best shards, thus variance in RBR results is due to differences in partitioning effectiveness. Rank-S and Taily use models (resource selection indices) to make decisions about which shards to select for each query, however those models are necessarily incomplete. The increases in variance for Rank-S and Taily as compared to RBR are due to resource selection errors caused by weaknesses in the models of shard contents.

Selective search instances displayed differing levels of variability across different metrics. P@10 overall had a lower variance coefficient than MAP@500 (Table 1 and Table 2). Similar behavior is observed for Taily instances. This behavior indicates that selective search is more stable at the top of the ranking.

Rank-S is affected by one more random component than Taily, thus it might be expected to have greater variability across system instances. Table 3 shows the variance coefficient of MAP@500 for Rank-S and Taily across ten CW09-A system instances; similar behavior was observed across all metrics for both collections. Taily only had lower variance than Rank-S on 1 of the 4 query sets. These results might be considered surprising. The additional random component does not appear to cause greater instability in Rank-S results. As far as we know, prior research has not investigated the variability of results produced by sample-based and term-based algorithms such as Rank-S and Taily.

3.2.2 Variability for queries

Ideally selective search would provide similar results for a given query regardless of which system instance is used. The second experiment examined variability on a query-by-query basis.

Ore over 5 at	CW09-	В	CW09-A	CW09-A		
Query Set	SD	VC	SD	VC		
Web Track 2009	$1.38 imes 10^{-2}$	2.71%	1.05×10^{-2}	2.95%		
Web Track 2010	$1.16 imes 10^{-2}$	2.15%	$9.73 imes 10^{-3}$	2.04%		
Web Track 2011	1.36×10^{-2}	3.53%	1.31×10^{-2}	3.04%		
Web Track 2012	5.39×10^{-3}	1.39%	$9.73 imes 10^{-3}$	2.63%		
	(a)]	P@10				
Ora orrea C a t	CW09-B		CW09-A	A		
Query Set	SD	VC	SD	VC		
Web Track 2009	$3.56 imes 10^{-3}$	3.25%	3.54×10^{-3}	2.97%		
Web Track 2010	$3.87 imes 10^{-3}$	2.24%	$2.46 imes 10^{-3}$	1.35%		
Web Track 2011	1.10×10^{-2}	7.21%	$9.47 imes 10^{-3}$	4.40%		
Web Track 2012	3.53×10^{-3}	2.41%	4.97×10^{-3}	2.91%		

Table 2: Standard Deviation (SD) and Variance Coefficient (VC) of accuracy scores for RBR instances.

(b) MAP@500

Table 3: Comparison of MAP@500 scores (CW09-A) for Rank-S and Taily instances.

Outomy Cot	Mean		_	Variance Coefficient	
Query Set	Rank-S	Taily		Rank-S	Taily
Web Track 2009	0.062	0.065		10.65%	10.98%
Web Track 2010	0.090	0.087		3.66%	6.89%
Web Track 2011	0.101	0.084		7.62%	15.67%
Web Track 2012	0.097	0.085		8.23%	5.36%

We examined the variance of Average Precision of each query across Rank-S and Taily instances on CW09-A and observed some highly variant queries in both Rank-S and Taily. Rank-S had 14 queries with AP standard deviation higher than 0.1; Taily had 17 such high variance queries.

High variance could be due to errors from the partitioning process or errors from resource selection. To discover the source of the variance, a query-by-query experiment was conducted with RBR on the same partitions. RBR does not make resource selection errors but *is* affected by partitioning errors. Thus, if Rank-S, Taily, and RBR all have trouble with a query, the problem is likely to be partitioning.

Figure 3 shows the average precision (AP) of each query for ten RBR system instances. A notable observation is that most of the queries have stable AP; 177 of the 200 queries had a standard deviation of less than 0.05. However, there were a few queries with very high variance that contributed most of the average variance.



Figure 3: Average Precision of RBR instances for Web Track queries. Each data point is a box and whisker plot with the edges of the boxes representing the upper and lower quantiles, the mid line the median, and the whiskers the 1.5 interquartile range. + are outliers.

Whether the query is variable or stable remained mostly consistent across RBR, Rank-S, and Taily. While there were some highly variable queries in RBR that were not so in Taily and Rank-S, these were difficult queries (low 'search all shards' search accuracy), and had nearly zero AP@500 in Rank-S and Taily. All easy queries with high variance in RBR also had high variance in Rank-S and Taily. The consistency across RBR, Rank-S and Taily suggests that errors from the partitioning stage are a major source of variance of selective search accuracy; however, most of that variation comes from a small number of queries.

We investigated why some partitions performed much better than others on the high variance queries. One might expect that 'good' partitions group relevant documents into fewer shards. However, in all of our partitions, relevant documents were distributed across a small number of shards. Typically the 3 most relevant shards contained more than 60% of the relevant documents, which is consistent with the standards of most prior federated search research. Furthermore, in this experiment, all ten instances retrieved 100% of the relevant documents for 4 of the 5 most variant queries. However, these queries had AP values ranging from values similar to exhaustive search to as much as ten times *better* than exhaustive search.

An examination of the queries with unusually high AP in the RBR experiment revealed that the relevant documents and the most likely false positives were in different shards. Thus, the combination of partitioning and resource selection 'filtered out' non-relevant documents that otherwise would have been ranked highly. Although this behavior might seem desirable, it occurred because the partitioning process incorrectly assigned relevant documents to shards that contained documents that were largely dissimilar to the relevant documents. This poor clustering made it nearly impossible for Rank-S and Taily to identify which shards contained relevant documents due to the overwhelming number of dissimilar documents in the shard. RBR, which *knows* the number of relevant documents in each shard, had no such problem.

Figure 3 also indicates that different query sets produced different levels of variability. The 2011 query set was notably more unstable than the others. Among the 24 queries with AP standard deviation higher than 0.05, 14 were from TREC 2011. We believe that this difference is due to the other three query sets using topics of medium-to-high frequency, while TREC 2011 used more obscure topics [Clarke et al. 2011]. Our results indicate that the *type* of query has an impact on not only the average accuracy of a system, but also the variance of a system with random components. Topic-based partitioning may produce stable results for common topics, but may need improvement for queries about rare topics.

3.3 SUMMARY

Understanding the variance of selective search effectiveness is critical for evaluating and improving selective search. To answer RQ 1, this chapter explored selective search variance and the effects of document collections, resource selection algorithms, and query characteristics. Partitioning and resource selection processes both introduced variance into search behavior, however in our experiments the effect on most queries was not large. Results were more stable at the top of the ranking, but variance at rank 500 was not unreasonable, especially for an architecture that avoids searching most of the index.

Rank-S and Taily produced nearly equal variance, which might be considered surprising given that Rank-S has one more random component than Taily.

Most of the variance observed in our per-query experiments was caused by a small number of queries – typically, rare queries. In some instances the partitioning process correctly grouped relevant documents together, but placed them in unrepresentative shards, which caused poor resource selection. This behavior may indicate the need for partitioning processes more sophisticated than simple K-means clustering.

Overall, our results show that selective search has reasonable variance and that the accuracy claims about the architecture can be trusted. The next chapter will address whether the efficiency claims of selective search can be trusted in combination with optimization techniques that may be encountered in the real world.

DYNAMIC OPTIMIZATION¹

Efficient production search engines are likely to include optimization techniques. Dynamic pruning algorithms such as *Weighted AND* (WAND) [Broder et al. 2003] and *term-bounded max score* (TBMS) [Strohman et al. 2005] improve the computational efficiency of retrieval systems by eliminating or early-terminating score calculations for documents which cannot appear in the top-k of the final ranked list.

But topic-based partitioning and resource selection change the environment in which dynamic pruning is performed, and query term posting lists are likely to be longer in shards selected by the resource selection algorithm than in shards that are not selected. As well, each topic-based shard should contain similar documents, meaning that it might be difficult for dynamic pruning to distinguish amongst them using only partial score calculations. Conversely, the documents in the shards that were *not* selected for search might be the ones that a dynamic pruning algorithm would have bypassed if it had encountered them. That is, while the behavior of dynamic pruning algorithms on randomly-organized shards is well-understood, the interaction between dynamic pruning and the topic-based shards used by selective search is not. As an extreme position, it might be argued that selective search is simply achieving the same computational savings that dynamic pruning would have produced, but incurs the additional overhead of clustering the collection and creating the shards.

To address these concerns, we investigate the behavior of the well-known *Weighted AND* (WAND) dynamic pruning algorithm in the context of selective search, considering two research questions:

RQ 2 Does dynamic pruning improve selective search, and if so, why?

RQ 3 *Can the efficiency of selective search be improved further using a cascaded pruning threshold during shard search?*

4.1 RELATED WORK

Weighted AND (WAND) is a dynamic pruning algorithm that only scores documents that may become one of the current top k based on a preliminary estimate [Broder et al. 2003]. Dimopoulos et al. [2013] developed Block-Max WAND (BM-WAND) in which continuous segments of postings data are bypassed under some circumstances by using an index where each block of postings

¹ This chapter is a lightly-revised version Yubin Kim, Jamie Callan, J. Shane Culpepper, and Alistair Moffat. Does selective search benefit from WAND optimization? In *Advances in Information Retrieval: 38th European Conference on IR Research*, pages 145–158, 2016 and reuses text jointly written by all authors. Kim designed and ran the experiments and generated the figures for the paper.



Figure 4: Distribution of shard sizes, with a total of 100 shards.

has a local maximum score. Petri et al. [2013] explored the relationship between WAND-style pruning and document similarity formulations. They found that WAND is more sensitive than BM-WAND to the document ranking algorithm. If the distribution of scores is skewed, as is common with BM25, then WAND alone is sufficient. However, if the scoring regime is derived from a language model, then the distribution of scores is top-heavy, and BM-WAND should be used. Rojas et al. [2013] presented a method to improve performance of systems combining WAND and a distributed architecture with random shards.

Term-Bounded Max Score (TBMS) [Strohman et al. 2005] is an alternative document-at-a-time dynamic pruning algorithm that is currently used in the Indri Search Engine. The key idea of TBMS is to precompute a "topdoc" list for each term, ordered by the frequency of the term in the document, and divided by the document length. The algorithm uses the union of the topdoc lists for the terms to determine a candidate list of documents to be scored. The number of documents in the topdoc list for each term is experimentally determined, a choice that can have an impact on overall performance. Kulkarni and Callan [2015] explored the effects of TBMS on selective search and traditional distributed search architectures. Based on a small set of queries they measured efficiency improvements of 23-40% for a traditional distributed search architecture, and 19-32% for selective search, indicating that pruning can improve the efficiency of both approaches.

4.2 EXPERIMENTS

The observations of Kulkarni and Callan [2015] provide evidence that dynamic pruning and selective search can be complementary. Our work extends that exploration in several important directions. First, we investigate whether there is a correlation between the rank of a shard and dynamic pruning effectiveness for that shard. A correlation could imply that dynamic pruning effectiveness depends on the number of shards searched. We focus on the widely-used WAND pruning algorithm, chosen because it is both efficient and versatile, particularly when combined with a scoring function such as BM25 that gives rise to skewed score distributions [Dimopoulos et al. 2013; Petri et al. 2013].

Experiments were conducted using the ClueWebo9 Category B dataset, containing 50 million web documents. The 100 topical shards from Section 3.1 were used, but spam documents were kept in the index for a larger index size, so that it provides clearer signals about changes in



Figure 5: Correlation between the number of postings processed for a query and the time taken for query evaluation. Data points are generated from MQT queries using both WAND and full evaluation, applied independently to all 100 topical shards and all 100 random shards. In total, $756 \times 200 \times 2 \approx 300,000$ points are plotted.

efficiency. The index was stopped using the default Indri stoplist and stemmed using the Krovetz stemmer.

A second partition of 100 random shards was also created, a system in which exhaustive "all shards" search is the only way of obtaining effective retrieval. Each shard in the two systems was searched using BM25, with $k_1 = 0.9$, b = 0.4, and global corpus statistics for *idf* and average document length.² BM25 was chosen over the language model scores typically used by Indri because WAND has reduced effectiveness on systems using language model scoring [Petri et al. 2013].

Each selected shard returned its top 1,000 documents, which were merged by score to produce a final list of k = 1,000 documents. In selective search, deeper ranks are necessary because most of the good documents may be in one or two shards due to the term skew. Also, deeper k supports learning-to-rank algorithms. Postings lists were compressed and stored in blocks of 128 entries using the FastPFOR library [Lemire and Boytsov 2015], supporting fast block-based skipping during the WAND traversal.

Two resource selection algorithms were used: Taily [Aly et al. 2013] and Rank-S [Kulkarni et al. 2012]. The parameters for Taily (n = 400, v = 50) and Rank-S (B = 5, CSI = 1%) remain unchanged from Section 3.1. We were unable to find parameters that consistently yielded better results than the original published values.

We conducted evaluations using the first 1,000 unique queries from each of the AOL query log and the TREC 2009 Million Query Track. We removed single-term queries, which do not benefit from WAND, and queries where the resource selection process did not select any shards. Removing single-term queries is a common procedure for research with WAND [Broder et al. 2003] and allows our results to be compared with prior work. That left 713 queries from the AOL log, and 756 queries from MQT, a total of 1,469 queries.

² The values for b and k_1 are based on the parameter choices reported for Atire and Lucene in the 2015 IR-Reproducibility Challenge, see github.com/lintool/IR-Reproducibility.



Figure 6: Ratio of savings achieved by WAND as a function of the total postings length of each query in the AOL set, measured on a per shard basis. A total of $100 \times 713 \approx 71,000$ points are plotted. Queries containing only rare terms derive little benefit from WAND.

Our focus in this chapter is on the efficiency of shard search, rather than resource selection. To compare the efficiency of different shard search methods, we count the number of postings scored, a metric that is strongly correlated with total processing time [Broder et al. 2003], and is less sensitive to system-specific tuning and precise hardware configuration than is measured execution time. As a verification of this relationship, Figure 5 shows the correlation between processing time per query, per shard, and the number of postings evaluated. There is a strong linear relationship; note also that more than 99.9% of queries completed in under 1 second with only a few extreme outliers requiring longer.

4.2.1 Pruning effectiveness of WAND on topical shards

The first experiment investigated how WAND performs on the topical shards constructed by selective search. Each shard was searched independently, as is typical in distributed settings – parallelism is crucial to low response latency. w, the number of posting evaluations required in each shard by WAND-based query evaluation was recorded. The total length of the postings for the query terms in the selected shards was also recorded, and is denoted as b, representing the number of postings processed by an unpruned search in the same shard. The ratio w/b then measures the fraction of the work WAND carried out compared to an unpruned search. The lower the ratio, the greater the savings. Values of w/b can then be combined across queries in two different ways: micro- and macro-averaging. In micro-averaging, w and b are summed over the queries and a single value of w/b is calculated from the two sums. In macro-averaging, w/b is calculated for each query, and averaged across queries. The variance inherent in queries means that the two averaging methods can produce different values, although broad trends are typically consistent.

Figure 6 and Table 4 provide insights into the behavior of macro- and micro-averaging. Figure 6 uses the AOL queries and all 100 topical shards, plotting *w*/b values on a per query per shard basis as a function of the total length of the postings lists for that query in that shard. Queries involving only rare terms benefit much less from WAND than queries with common terms. Thus, the macro-average of *w*/b is higher than the micro-average. Micro-averaging more accurately

		WAND postings cost ratio		WAND runt	time cost ratio
		Topical shards	Random shards	Topical shards	Random shards
AOL	micro-averaged	0.35	0.34	0.36	0.38
MQT	micro-averaged	0.36	0.36	0.39	0.43
AOL	macro-averaged	0.51	0.52	0.51	0.53
MQT	macro-averaged	0.60	0.63	0.58	0.63

Table 4: Ratio of per shard per query postings evaluated and per shard per query execution time for WAND-based search, as ratios relative to unpruned search, averaged over 100 topical shards and over 100 randomized shards, and over two groups each of 700+ queries. The differences between the Topical and Random macro-averaged ratios are significant for both query sets and both measures (paired two-tailed t-test, p < 0.01).

	Shards searched	WAND postings cost ratio		WAND runtime cost ratio	
		Selected Non-selected		Selected	Non-selected
Taily AOL	3.1	0.32	0.35	0.36	0.36
Taily MQT	2.7	0.23	0.37	0.30	0.40
Rank-S AOL	3.8	0.27	0.36	0.30	0.37
Rank-S MQT	3.9	0.24	0.37	0.30	0.40

Table 5: Average number of shards searched, and micro-averaged postings ratios for those selected shards and for the complement set of shards, together with the corresponding query time cost ratios, in each case comparing WAND-based search to unpruned search. Smaller numbers indicate greater savings.

represents the total system savings, whereas macro-averaging allows paired significance testing. We report both metrics in Table 4. The second pair of columns gives millisecond equivalents of w/b, to further validate the postings-cost metric. These values are micro- and macro-averaged w_t/b_t ratios, where w_t is the time in milliseconds taken to process one of the queries on one of the 100 shards using WAND, and b_t is the time taken to process the same query with a full, unpruned search. A key result of Table 4 is that WAND is just as effective across the full set of topical shards as it is on the full set of randomly formed shards. Moreover, the broad trend of the postings cost ratios – that WAND avoids nearly half of the postings – is supported by the execution time measurements.

4.2.2 WAND and resource ranking interactions

The second experiment compares the effectiveness of the WAND algorithm on the shards that the resource ranking algorithm would, and would not, select in connection with each query. The Taily and Rank-S resource selection algorithms were used to determine which shards to search.

	WAND postings cost ratio Selected Non-selected		WAND runtime cost ratio		
			Selected	Non-selected	
Taily AOL	0.42	0.52	0.45	0.52	
Taily MQT	0.52	0.61	0.53	0.59	
Rank-S AOL	0.42	0.53	0.44	0.52	
Rank-S MQT	0.52	0.61	0.53	0.60	

Table 6: As for Table 5, but showing macro-averaged ratios. All differences between selected and non-selected shards are significant (paired two-tailed t-test, p < 0.01).

For each query the WAND savings were calculated for the small set of selected shards, and the much larger set of non-selected shards.

Table 5 lists micro-averaged w/b ratios, and Table 6 the corresponding macro-averaged ratios. While all shards see improvements with WAND, the selected shards see a greater efficiency gain than the non-selected shards, reinforcing our contention that resource selection is an important component in search efficiency. When compared to the ratios shown in Table 4, the selected shards see substantially higher benefit than average shards; the two orthogonal optimizations generate better-than-additive savings.

Figure 7a shows the distribution of the individual per query per shard times for the MQT query set, covering only the shards chosen by the two resource selection processes, in the first four configurations. The fifth exhaustive search configuration includes data for all of the 100 randomlygenerated shards making up the second system, and is provided as a reference point. Figure 7b gives numeric values for the mean and median of each of the five distributions. When WAND is combined with selective search, it both reduces the average time required to search a shard and also reduces the variance of the query costs. Note the large differences between the mean and median query processing times for the unpruned search and the reduction in that gap when WAND is used; this gain arises because query and shard combinations that have high processing times due to long postings lists are the ones that benefit most from WAND. Therefore, in typical distributed environments where shards are searched in parallel, the slowest, bottleneck shard will benefit the most from WAND and may result in additional gains in latency reduction. Furthermore, while Figure 7 shows similar per shard query costs for selective and exhaustive search, the total work associated with selective search is substantially less than exhaustive search because only 3-5 shards are searched per query, whereas exhaustive search involves all 100 shards. Taken in conjunction with the previous tables, Figure 7 provides clear evidence that WAND amplifies the savings generated by selective search, answering the first part of RQ 2 with a "yes". In addition, these experiments have confirmed that execution time is closely correlated with measured posting evaluations. The remaining experiments utilize postings counts as the cost metric.

We now consider the second part of RQ 2 and seek to explain *why* dynamic pruning improves selective search. Part of the reason is that the postings lists of the query terms associated with the highly ranked shards are longer than they are in a typical randomized shard. With these long postings lists, there is more opportunity for WAND to achieve early termination. Figure 8 shows normalized final heap-entry thresholds, or equivalently, the similarity score of the 1,000 th ranked document in each shard. The scores are expressed as a fraction of the maximum document



Figure 7: Distribution of query response times for MQT queries on shards: (a) as a box plot distribution, with a data point plotted for each query-shard pair; (b) as a table of corresponding means and medians. In (a), the center line of the box indicates the median, the outer edges of the box the first and third quartiles, and the blue circle the mean. The whiskers extend to include all points within 1.5 times the inter-quartile range of the box. The graph was truncated to omit a small number of extreme points for both Rank-S Full and Taily-Full. The maximum time for both these two runs was 6,611 ms.

score for that query across all shards, then plotted as a function of the resource selector's shard ranking using Taily, averaged over queries. Shards that Taily did not score because they did not contain any query terms were ordered randomly. For the AOL log the 1,000 th document in the shard ranked highest by Taily attains, on average across queries, a score that is a little over 80% of the maximum score attained by any single document for that same query. The downward trend in Figure 8 indicates that the resource ranking process is effective, with the high heap-entry thresholds in the early shards suggesting – as we would hope – that they contain more of the high-scoring documents.

To further illustrate the positive relationship between shard ranking and WAND, *w/b* was calculated for each shard in the per query shard orderings, and then micro-averaged at each shard rank. Figure 9 plots the average as a function of shard rank, and confirms the bias towards greater savings on the early shards – exactly the ones selected for evaluation. As a reference point, the same statistic was calculated for a random ordering of the randomized shards (random since no shard ranking is applied in traditional distributed search), with the savings ratio being a near-horizontal line. If an unpruned full search were to be plotted, it would be a horizontal line at



Figure 8: Normalized 1,000 th document scores from shards, averaged over queries and then shard ranks, and expressed as a fraction of the collection-wide maximum document score for each corresponding query. The score falls with rank, as fewer high-scoring documents appear in lower-ranked shards.



Figure 9: The micro-average w/b ratio for WAND postings evaluations, as a function of the per query shard rankings assigned by Taily. Early shards generate greater savings.

1.0. The importance of resource selection to retrieval effectiveness has long been known; Figure 9 indicates that effective resource selection can improve overall efficiency as well.

4.2.3 Improving efficiency with cascaded pruning thresholds

In the experiments reported so far, the rankings were computed on each shard independently, presuming that they would be executing in parallel and employing private top-k heaps and private heap-entry thresholds, with no ability to share information. This approach minimizes search latency when multiple machines are available, and is the typical configuration in a distributed search architecture. An alternative approach is suggested by our second research question: what happens if the shards are instead searched sequentially, passing the score threshold and top-k heap from each shard to the next? The heap-entry score threshold is then non-decreasing across



Figure 10: Normalized 1,000 th document scores from shards relative to the highest score attained by any document for the corresponding query, micro-averaged over queries, assuming that shards are processed sequentially rather than in parallel, using the Taily-based ordering of topical shards and a random ordering of the same shards.



Figure 11: Ratio of postings evaluated by WAND for independent shard search versus sequential shard search, AOL queries with micro-averaging. Shard ranking was determined by Taily.

the shards, and additional savings should result. While this approach would be unlikely to be used in an interactive system that demands low latency, it provides an upper bound on the efficiency gains that are possible if a single heap was shared by all shards, and would increase throughput when limited resources are available and latency is not a concern: for example, in off-line search and text analytics applications.

Figure 10 demonstrates the threshold in the sequential WAND configuration, with shards ordered in two ways: by Taily score, and randomly. The normalized threshold rises quickly towards the maximum document score through the first few shards in the Taily ordering, which is where most of the documents related to the query are expected to reside. Figure 11 similarly plots the w/b WAND savings ratio at each shard rank, also micro-averaged over queries, and with shard ordering again determined by the Taily score. The independent and sequential configurations diverge markedly in their behavior, with a deep search in the latter processing far fewer postings than a deep search in the former. The MQT query set displayed similar trends. Sharing the dynamic pruning thresholds has a large effect on the efficiency of selective search.

Our measurements suggest that a hybrid approach between independent and sequential search could be beneficial. A resource-ranker might be configured to underestimate the number of shards that are required, with the understanding that a second round of shard ranking can be instigated in situations where deeper search is needed, identified through examining the scores or the quantity of documents retrieved. When a second wave of shards is activated, passing the maximum heap-entry threshold attained by the first-wave process would reduce the computational cost. If the majority of queries are handled within the first wave, a new combination of latency and workload will result.

4.3 SUMMARY

To date there has been only limited consideration of the interaction between dynamic pruning and selective search [Kulkarni and Callan 2015], and it has been unclear whether dynamic pruning methods improve selective search, or whether selective search is capturing some or all of the same underlying savings as pruning does, just via a different approach.

This chapter explores WAND dynamic pruning using a large dataset and two different query sets to answer RQ 2. In contrast to Kulkarni's findings with TBMS [Kulkarni and Callan 2015], we show that WAND-based evaluation and selective search generate what are effectively independent savings, and that the combination is more potent than either technique is alone – that is, that their interaction is a positive one. In particular, when resource selection is used to choose query-appropriate shards, the improvements from WAND on the selected shards is greater than the savings accruing on random shards, confirming that dynamic pruning further improves selective search – a rare situation where orthogonal optimizations are better-than-additive. We also demonstrated that there is a direct correlation between the efficiency gains generated by WAND and the shard's ranking. While it is well-known that resource selection improves effectiveness, our results suggest that it can also improve overall efficiency too.

Finally, two different methods of applying WAND to selective search were compared and we found that passing the top-k heap through a sequential shard evaluation greatly reduced the volume of postings evaluated by WAND, answering RQ 3 in the affirmative. The significant difference in efficiency between this approach and the usual fully-parallel mechanism suggests avenues for future development in which hybrid models are used to balance latency and throughput in novel ways.

The results of this chapter show that selective search remains efficient in conjunction with a well-known dynamic optimization technique. With assurances on accuracy and efficiency in hand, we now move on to solving problems in the natural use-case of selective search, that of serving as a fast, first-stage retrieval system in a multi-pipeline retrieval architecture.

RESOURCE SELECTION WITH WAND

Prior work [Kulkarni 2013; Aly et al. 2013; Dai et al. 2016] showed that while selective search is equivalent to exhaustive search for shallow metrics (e.g. P@10), it performs worse for recalloriented metrics (e.g. MAP) under commonly studied parameter settings. This is a problem because multi-stage retrieval systems apply re-ranking operations to a base retrieval, which can require deep result lists [Macdonald et al. 2013]. Poor recall prevents selective search from being used as an efficient first-stage retrieval in a multi-stage search system. This motivates us to design a more accurate resource selection.

Block-Max WAND (BM-WAND) [Dimopoulos et al. 2013] is an extension of WAND, a dynamic pruning algorithm which skips evaluating sections of the postings list during query time when those segments are guaranteed not to contain high scoring documents. In order to accomplish this, at index time BM-WAND segments a term posting list into blocks that span n documents each and stores the local maximum term score found in the block as the block max score. This provides information on the term score distribution of the postings which may be useful for resource selection. This chapter shows that using BM-WAND statistics for resource selection is a light-weight solution to some resource selection tasks that does not require gathering additional corpus statistics or building a resource selection database. To summarize, our research question is:

RQ 4 Can BM-WAND information be used to design an effective resource selection algorithm?

5.1 METHODS

At first glance, it seems obvious that the maximum term scores stored by BM-WAND could be used to model term distributions in a shard and thus be used for resource selection. However, a naive implementation of BM-WAND is less helpful than expected.

Typically, BM-WAND gathers the statistics it needs from a postings list from the index without any special sorting operations. Consequently, the documents in the postings list are in the order that was supplied to the search index at index time, which is usually random or approximately grouped by the source domains of the documents. The shards used in this work roughly ordered the documents by source and preliminary experiments found that source ordering was not ideal. We found that the block max scores calculated from the source ordered postings list contained very little signal for the actual term score distribution of the documents in the list.

Figure 12 shows the block max scores of the term *obama* for a topical shard that was highly ranked for the query *obama family tree* where the postings list was in source order. The distribution of scores in this document order was near random, which is typical. When the max scores



Figure 12: Comparison of impact ordered posting list and source ordered posting list block max score distributions of the term *obama* in a top shard for the query *obama family tree* (shard 57). Block size is n = 128.

were calculated from each block, it formed a flat distribution with little variation, which was not representative of the full range of document scores.

Alternatively, the documents in a postings list can be sorted in *impact ordering* [Anh and Moffat 2002]. In this ordering, a term's postings list is sorted such that the highest scoring documents for that term appear first in the list. As seen in Figure 12, the block max scores calculated from this ordering give much better insight into the distribution of document scores for the term across the shard. Furthermore, block max scores of impact ordered lists have a clear intuitive meaning; the block max score of the first block is the score of the highest scoring document for the term in that shard, the block max score of the second block using size n is the score of the n + 1'th highest scoring document for the term, etc. Therefore, we use block max scores for impact ordered posting lists over source order posting lists in our experiments.

Initially, we experimented with a range of query lengths and attempted to handle multi-term queries in two different ways. In the first method, for a given shard, a score was generated for each query term and then the whole query score was produced by adding the individual term scores. In the second method, we sought to better capture term co-occurrence using source-ordered posting lists, where document ordering is consistent across all terms. We identified the blocks where the query terms overlapped and calculated the density of the maximum possible score for the overlapped blocks, as defined by the maximum score possible in the overlapped block divided by the number of documents in that block.

Neither of these methods produced good results. Our hypothesis was that we were mishandling merging information across terms in multi-term queries rather than an innate lack of signal in the block max scores. In order to test this hypothesis, we narrowed our focus to single term queries. It is true that for single term queries, the best shards to search can be pre-computed based on the top scoring documents for the term. However, the goal of this work is exploratory, to gauge the usefulness of block max scores and lay the foundation for future work to merge the term scores to serve multi-term queries as well.

We compared two methods of using block max scores for shard selection of single term queries. The first method, *second score*, orders the shards by the score of second highest block, where the block size n = 128. In a single term query, this is equivalent to ranking the shards by the score of the 129th highest scoring document from that shard. After the shards were ranked, a static cut-off

Algorithm 1 Multi-size second scores (M₃S) algorithm for shard selection.

k = input threshold> document ranking length $N = \{1, 2, 4, 8, 16, 32, 64, 128\}$ ▷ block size steps used ▷ set of selected shards $S = \{\}$ $B = \{b_{n_i,s} \mid \forall n_i \in N, \forall s \in all \text{ shards}\}$ > all second score blocks from all shards $\hat{\mathbf{k}} = \mathbf{0}$ ▷ estimated number of top scoring documents in S while $\hat{k} < k$ do $b_{n_i,s}$ = block with highest max score in B add shard s to S remove $b_{n_i,s}$ from B $\hat{\mathbf{k}} = \hat{\mathbf{k}} + \mathbf{n}_i - \mathbf{n}_{i-1}$ \triangleright where n_0 is defined as 0 for convenience return S



Figure 13: An example of the multi-size second scores method. If k = 2, $b_{1,1}$ is the first block selected $(\hat{k} = 1)$, then $b_{1,2}$ $(\hat{k} = 2)$ resulting in shard 1 being selected for search, as ideal. However, if k = 3, $b_{2,1}$ $(\hat{k} = 3)$ is also selected, resulting in both shards being selected for search even though shard 2 is unnecessary.

was applied. In our experiments, we used a cut-off of 3 and 5 in order to produce comparable results to Taily and Rank-S, two state-of-the-art resource selection algorithms to which we compare our method.

The second method, *multi-size second scores* (M₃S), extends the above idea by using block max scores from blocks of different sizes. Rather than using a single data point to represent a shard's worth, the block max scores of the second highest block of block sizes $N = \{1, 2, 4, 8, 16, 32, 64, 128\}$ are collected. (Alternatively, this is equivalent to sampling the score of the impact ordered posting list at document rank 2, 3, 5 etc.) These block max scores are then pooled and blocks with the highest scores are selected and removed from the pool. Each selected block of size n_i contributes $n_i - n_{i-1}$ additional documents to the estimated ranked list. Because of the impact ordered postings list, it is impossible that a block of a larger size would be selected before a block of a smaller size from the same shard. Therefore, a new block contributes additional documents equal to its size less the total number of documents already contributed by the shard, i.e. the size of a block of one size step lesser. This process continues until the estimated ranked list size exceeds a threshold, k. Then, all shards represented by the selected blocks become the set of shards to search. This algorithm is summarized in Algorithm 1.

The M₃S method uses the block max scores to attempt to find all shards represented in the top-k of a document ranking created by exhaustive search. In the example shown in Figure 13,

for a desired k = 2, the algorithm first selects $b_{1,1}$ ($\hat{k} = 1$), then $b_{1,2}$ ($\hat{k} = 2$) and stops, resulting in shard 1 being selected for search.

In practice, the algorithm may select more shards than necessary. If the desired k = 3 in Figure 13, the algorithm would select an additional block $b_{2,1}$ ($\hat{k} = 3$) before halting. This results in both shard 1 and 2 being selected even though the true top 3 documents all reside in shard 1.

Unlike the second score method, M₃S can select a different number of shards for each query. In practice, M₃S selects a large number of shards because there are many shards that contribute only one or two documents to the final result list. Thus, we also explored a static cut-off variation, in which shards are sorted by the number of documents they are estimated to contribute to the top-k and a static cut-off of 3 or 5 is applied, similar to the setup for the second scores method. In our experiments, we set k = 100 unless otherwise specified.

5.2 EXPERIMENTAL SETUP

The experiments were carried out with the ClueWebo9 Category B dataset, which includes 50 million English documents, and single term queries from the TREC Web track query set from 2009–2012, a total of 58 queries with relevance judgments.

The dataset was clustered into 296 topical shards using the SB² K-means clustering algorithm [Kulkarni 2013], which is a refinement in shard creation methodology from the method used in Section 3.1 and Section 4.2. The method initially formed 200 shards then in a second step reclustered the largest shards to smaller pieces in order to balance the shard sizes. The larger number of shards used for the dataset compared to previous experiments was due to the fact that that smaller shards produced more accurate results [Dai et al. 2016].

The shards were searched using BM25 with similar settings as Section 4.2: $k_1 = 0.9$, b = 0.4, and using the global corpus statistics for idf and average document length. BM25 was chosen for its simple operator for combining term scores (i.e. addition) and because it is known to work well with WAND. Each shard returned 1,000 results and the top 1,000 documents from the final merged list were used to calculate effectiveness metrics.

The baselines are exhaustive search, i.e. the effectiveness gained by searching the entire corpus, and two state-of-the-art resource selection algorithms, Taily [Aly et al. 2013] and Rank-S [Kulkarni et al. 2012]. Exhaustive search was performed using BM25 with the same parameters as above. The parameters used for Taily and Rank-S were the best parameters as reported by Aly et al. [2013] ($n_c = 400$, $\nu = 50$) and Kulkarni [2013] (base = 5, CSI = 1%).

Statistical significance testing was performed with a non-inferiority test commonly used in the medical field to test that two treatments have similar effectiveness. In this test, rather than testing to reject the null hypothesis $H_0 : \mu_A = \mu_B$, we test to reject $H'_0 : \mu_A - \mu_B \ge \delta$ for some small margin, δ . Intuitively, by rejecting H'_0 we accept the alternative hypothesis, which is that any reduction of performance in system B compared to system A is inconsequential. Note that if $\delta = 0$, H'_0 tests for the statistical superiority of system B. Jayasinghe et al. [2015] presents a more thorough treatment of the subject of equivalence testing in IR systems. In our experiments, we set the margin δ to 5% and 10% of the mean of the system A.

5.3 EXPERIMENTS

Table 7 presents the results of the experiments. We first present the exhaustive search result, which is the target that the resource selection algorithms attempt to meet and is the system used

Table 7: The results of second scores and M₃S resource selection methods with different cut-off alternatives compared to exhaustive search and state of the art resource selection algorithm baselines for 58 single term queries. Avg shards shows the average number of shards selected for each query. For the M₃S, k = 100 unless otherwise specified. * indicates method is within a 10% margin of exhaustive search using a non-inferiority test with p = 0.05. + indicates result is within a 5% margin.

	Avg shards	P@10	NDCG@30	MAP@1000
Exhaustive	-	0.2741	0.1724	0.0903
Taily	2.78	0.2810	0.1797	0.0756
M3S (3 shards)	3	0.2948*	0.1856*	0.0769
2nd Score (3 shards)	3	0.2983*	0.1711	0.0727
Rank-S	4.76	0.2776	0.1690	0.0666
M ₃ S (5 shards)	5	0.2948+*	0.1947+*	0.0815
2nd Score (5 shards)	5	0.3207+*	0.1927+*	0.0855
M ₃ S	29	0.2741+*	0.1728+*	0.0873*
M ₃ S (k = 1000)	135	0.2741+*	0.1724+*	0.0903+*

as the basis of comparison in all non-inferiority testing. The following two groupings of results present our methods compared to two state-of-the-art resource selection baselines. A static cut-off was applied such that our methods search an equivalent number of shards to the baseline. This ensures that the accuracy of the results are presented for an equivalent efficiency level in the shard search step. The last group are two versions of the M₃S method with different k settings that were allowed select a different number of shards for each query.

Taily performs slightly better than Rank-S in single term queries across all metrics, whereas we have previously seen Rank-S outperform Taily when using the full query set (see Table 14). This result is expected, given that Rank-S has more information about term co-occurrence from its sample index, whereas Taily uses an explicit term independence assumption in its estimation. The results of Table 7 show that while Taily and Rank-S have lower accuracy in MAP@1000, they have similar or higher mean accuracies to exhaustive search when using shallow metrics. However, the results fail the non-inferiority test at both 5% and 10% margins, indicating that we cannot conclude that the Taily and Rank-S do not perform significantly worse than exhaustive search, even for shallow metrics.

When searching a similar number of shards as Taily and Rank-S, second score produced comparable results to Taily and more accurate results than Rank-S. M₃S was more accurate than both Taily and Rank-S. The strong performance in precision metrics is unsurprising, as both second score and M₃S explicitly focus on shards that deliver top-ranked documents. Unlike Taily and Rank-S, second score and M₃S were statistically non-inferior to exhaustive search in precisionoriented metrics (P@10 and NDCG@30) in most cases, which allows us to conclude that resource selection using these methods produces results that are as accurate as exhaustive search.

In MAP@1000, a recall-oriented metric, while M₃S using cut-offs performed better than Taily and Rank-S, it failed the non-inferiority test. When the full set of selected shards is searched for M₃S (i.e. 29 on average for k = 100 and 1₃₅ on average for k = 1000), the results pass the

non-inferiority test for MAP as well. Full M₃S also passes the non-inferiority test for both P@10 and NDCG@30, even though the mean for these metrics is sometimes lower than that of Taily or Rank-S. This is due to the variance of the systems. While Taily and Rank-S sometimes perform better than exhaustive search, they perform worse in more queries. For P@10, the win-loss ratio of Taily compared to exhaustive search is 15 wins and 17 losses. For Rank-S, it is 20 wins and 22 losses. However, M₃S exactly replicates the exhaustive search ranking to these depths and produces a 0 win 0 loss record, passing the non-inferiority test unlike Taily and Rank-S. While M₃S does not have a higher mean performance, the results are stable and would be particularly useful in systems that have a high penalty for missing documents, such as web search systems that use re-ranking algorithms.

The amount of shards searched for M₃S may seem large, but it is only necessary for tasks requiring high recall. Otherwise, searching 5 shards is sufficient for shallower metrics such as P@10 and NDCG@30. We can also compare the number of shards searched for M₃S to that of an oracle method. If one had perfect knowledge of which documents would appear in the top-k results of the final result list and which shards they came from, one can construct the ideal, minimum set of shards required to exactly reproduce the final ranked list. Compared to this ideal minimum, the number of shards selected by M₃S is quite reasonable, especially for the k = 100 configuration. The oracle method selects an average of 22 shards for k = 100 and an average of 87 shards for k = 1000.

The efficiency of the resource selection methods used in this chapter are independent of the shard search step and do not vary with the shard cut-offs. In the order of increasing latency, the efficiency of the algorithms are as follows: second score < Taily < M₃S << Rank-S. That is, the second score method is the most efficient and Rank-S, the least. For every shard-term pair, the second score method stores one value (the score of the second block), Taily stores two values (the n_c and ν parameters), and M₃S stores eight values (one second score per block size in N). Rank-S stores 1% of all postings in its CSI, which is typically much larger.

This result verifies our hypothesis that BM-WAND scores can be used to perform resource selection in single term queries. When searching an equivalent number of shards, the new methods out-performed existing baselines and were statistically significantly non-inferior to exhaustive search in precision-oriented metrics. When M₃S was allowed to search all shards that it selected, it produced significantly non-inferior results in recall-oriented MAP@1000 and stable precision results. Overall, M₃S was more accurate and had greater stability than Taily or Rank-S while not requiring additional corpus analysis or resource selection databases.

5.4 SUMMARY

Some dynamic pruning algorithms such as BM-WAND [Dimopoulos et al. 2013] collect statistics during indexing time to perform online optimizations. We showed that this information can be used to perform accurate, stable resource selection for single term queries. When impact and source ordered posting lists were compared, the block max scores for impact ordered postings displayed more score variation and a stronger signal for resource selection.

We experimented with two variations of using block max scores for resource selection with single term queries. When compared to two state-of-the-art resource selection algorithms, results from searching the shards selected by M₃S were found to be highly accurate in precision-oriented conditions and statistically non-inferior to exhaustive search in high-recall conditions, a difficult goal that was not achieved by the baselines. This indicates that a way of combining the term

scores from the block max score information may produce a good resource selection method for all queries. While this result does not conclusively answer RQ 4, the initial results are promising. In the next chapter, an alternative, supervised approach is presented that addresses the recall issue of selective search more completely.

In addition, we saw that it is feasible to determine a set of shards that would exactly replicate exhaustive search results that was close to the size of an oracle-based method. This presents the possibility of performing a low-variance, stable search within a selective search paradigm, which closely or exactly replicates exhaustive search results.

6

LEARNING TO RANK RESOURCES¹

In this chapter, we present learning to rank resources, a resource selection method based on learning-to-rank [Liu 2009]. While learning-to-rank has been widely studied for ranking documents, its application to ranking resources has not been studied in depth. We take advantage of characteristics of the resource ranking problem that are distinct from document ranking; we present new features; and we propose a training approach that uses exhaustive search results as the gold standard and show that human judgments are not necessary. We seek to answer the following major research questions:

RQ 5 *Is learning to rank resources an efficient algorithm that is as accurate as exhaustive search in deep recall-oriented metrics?*

RQ 6 Can learning to rank resources be trained without human judgements?

6.1 RELATED WORK

Learning to rank is a supervised approach to the ranking problem of information retrieval and can be grouped into three broad approaches: point-wise, pair-wise, and list-wise [Liu 2009]. Point-wise approaches attempt to predict the exact relevance degree of each document. Because point-wise approaches take as input a single document, they cannot see the relative or positional information of the ranked list. Pair-wise approaches such as SVM^{r a n k} [Joachims 2006] reduce the ranking problem into a binary classification problem, where a pair of documents are given as input and the classifier must decide the ordering of the documents. List-wise approaches such as ListMLE [Xia et al. 2008] take the entire set of documents for a given query as input and attempt to produce the correct permutation of documents. List-wise approaches are thought to generally perform better than pair-wise or point-wise approaches.

Supervised resource selection algorithms use training data to learn models to evaluate shard relevance. Kulkarni [2015] used a regression classifier to learn the appropriate number of shards to search for a given query and metric, but did not use supervised information for shard ranking. Most methods that rank shards using supervised methods train a classifier per shard [Arguello et al. 2009b; Cetintas et al. 2009]. However, training a classifier for every shard is expensive in selective search, where shards number in hundreds. Thus, supervised methods have not been used for selective search. Techniques that train a single classifier would be more suitable for

¹ This chapter is a lightly-revised version of Zhuyun Dai, Yubin Kim, and Jamie Callan. Learning to rank resources. In Proceedings of the 40th Annual International ACM SIGIR Conference on Research and Development in Information Retrieval, pages 837–840, 2017. The learning-to-rank model framework was implemented and evaluated by Dai. Dai and Kim designed and implemented features for the model. Kim created the Taily, champion list, and average distance to centroid features. The design of experiments, analysis, and paper writing were jointly shared by Kim and Dai.

selective search. Balog [Balog 2014] trained a learning-to-rank algorithm for Federated Web Search TREC track and presented preliminary results. Hong et al. [Hong et al. 2010] learned a joint probabilistic classifier called Jnt, which is used as a baseline in this work.

6.2 MODEL

Let q denote a query, P(q) the distribution of queries, and $\Phi(q, s_i)$ denote features extracted from the i'th shard for the query. The goal of learning-to-rank is to find a shard scoring function $f(\Phi(q, s))$ that can minimize the loss function defined as:

$$L(f) = \int_{q \in Q} l(q, f) dP(q)$$

We use:

$$l(q, f) = \sum_{s_i > q} s_j \, l\{f(\Phi(q, s_i)) < f(\Phi(q, s_j))\}$$

where $s_i >_q s_j$ denotes shard pairs for which s_i is ranked higher than s_j in the gold standard shard ranking with respect to query q. The 1{} function returns 1 if the boolean statement contained in the braces is true, and o if false. In other words, the loss is simply the number of times that the learned function was wrong about the ordering of shard pairs.

We used SVM^{rank} [Joachims 2006], which optimizes pair-wise loss. List-wise algorithms such as ListMLE [Xia et al. 2008] produced similar results, thus we only report results with SVM^{rank}.

The training process requires a gold standard shard ranking for each training query. We propose two definitions of the ground truth, *relevance-based* and *overlap-based*. In the *relevance-based* approach, the optimal shard ranking is determined by the number of relevant documents a shard contains. Thus, the training data require queries with relevance judgments, which can be expensive to obtain. The *overlap-based* approach assumes that the goal of selective search is to reproduce the document ranking of exhaustive search. The optimal shard ranking is determined by the number of documents in a shard that were ranked highly by exhaustive search. This does not require manual relevance judgments.

6.3 FEATURES

For a given query and a shard, we use various sources of information or *features* to generate a vector representation of how well the shard matches the query. We use three sources of information to define our features: query-independent information, term-based statistics, and information about sample-document rankings.

6.3.1 Query-independent information

Shard Popularity: Indicates how often the shard had relevant (relevance-based) or top-ranked (overlap-based) documents for training queries. It is query-independent and acts as a shard prior.

6.3.2 Term-based statistics

Term-based features can be easily precomputed, thus are efficient.

Taily Features: One feature is the Taily [Aly et al. 2013] score calculated for query q and shard s. However, Taily scores can vary greatly across shards and queries. For robustness, we add two additional features. If shard s is ranked r_s for query q, the *inverse rank* is $1/r_s$ [Guan et al. 2014], which directly describes the importance of s relative to other shards. The *binned rank* is ceiling(r_s/b), where b is a bin-size. We use b = 10, meaning that every 10 consecutive shards are considered equally relevant. This feature helps the model to ignore small differences between shards with similar rankings.

Champion List Features: For each query term, the top-k best documents were found in an offline process; for each term, the term scores for all documents containing the term are computed and the documents with the top-k highest scores for a given term are determined. The number of documents each shard contributes to the top-k was stored for each shard-term pair. For multi-term queries, the feature values of each query term were summed. We use two values of $k = \{10, 100\}$, generating 2 features.

Query Likelihood Features: The log-likelihood of a query with respect to the unigram language model of each shard is: $L(q|s) = \sum_{t \in q} \log p(t|s)$, where p(t|s) is the shard language model, the average of all document language models p(t|d) in the shard [Si et al. 2002]. Document language model p(t|d) is estimated using MLE with Jelinek-Mercer smoothing. Query likelihood, inverse query likelihood, and binned query likelihood features are created for *body*, *title*, and *inlink* representations, yielding a total of 9 features.

Query Term Statistics: The maximum and minimum shard term frequency across query terms, e.g. $stf_{max}(q, s) = max_{t \in q} stf(t, s)$, where stf(t, s) is the frequency of term t in shard s. We include the maximum and minimum of $stf \cdot idf$ where idf is the inverse document frequency over the collection. These 4 features are created for *body*, *title*, and *inlink* representations, yielding 12 features.

Bigram Log Frequency: The frequency of each bigram of the query in a shard is $bf_q(s) = \sum_{b \in q} \log bf_b(s)$, where $bf_b(s)$ is the frequency of bigram b in shard s. This feature can estimate term correlation. To save storage, we only store bigrams that appear more than 50 times in the collection.

6.3.3 Sample-document (CSI-based) features

These features are based on retrieval from the centralized sample index (CSI), which may provide term co-occurrence information. CSI retrieval is expensive, and thus is slower to calculate.

Rank-S and ReDDE Features: Similar to Taily features, the shard scores given by Rank-S [Kulkarni et al. 2012] and ReDDE [Si and Callan 2003], as well as the inverse rank and binned rank features for a total of 6 features.

Average Distance to Shard Centroid: The distance between the top-k documents retrieved from the CSI to their respective shards' centroids. Intuitively, if the retrieved documents are close to the centroid, the shard is more likely to contain other similar, highly-scoring documents. For multiple documents from the same shard, the distances are averaged. We use two distance metrics: KL divergence and cosine similarity Note that because KL divergence measures distance rather than similarity, we use the inverse of the averaged KL divergence as the metric. We generated features for $k = \{10, 100\}$ and also a feature measuring the distance between the shard's centroid to its single highest scoring document in the top 100 of the CSI results, for a total of 6 features.

6.4 EXPERIMENTAL METHODOLOGY

Datasets: Experiments were conducted with ClueWebo9-B and Gov2. Gov2, a new dataset introduced in this chapter, is a TREC collection containing 25 million .gov domain websites².

For *relevance-based* models, 200 queries from the TREC 2009–2012 Web Track topics were used for CW09-B, and 150 queries from the TREC 2004–2006 Terabyte Track topics were used for Gov2. Models were trained by 10-fold cross-validation. For *overlap-based* models, training queries were sampled from the AOL and Million Query Track query logs. For CW09-B, we sample 1000 queries from each source, giving us 2 training sets (AOL and MQT). For Gov2, the 1000 MQT queries were used as one training set. The second training set consisted of 1000 queries sampled from the AOL query log that have clicks on '.gov' URLs. Models were tested with the TREC queries. Optimal shard ranking for the overlap method was defined by the number of documents each shard contains that were within the top N = 2K retrieved from exhaustive search. We found $N \in [1K, 3K]$ produced stable results.

Proposed methods and baselines: We used three sources of training data: relevance-based training data (L2R-TREC), and overlap-based training data (L2R-AOL and L2R-MQT). We used linear SVM^{rank}, where C was chosen by cross validation. Our method was compared against state-of-the-art unsupervised models (Taily [Aly et al. 2013], ReDDE [Si and Callan 2003], and Rank-S [Kulkarni et al. 2012]); and a supervised model Jnt [Hong et al. 2010].

In order to ensure that the accuracy results are apples-to-apples across resource selection algorithms (searching more shards usually results in higher accuracy), we compare the methods at various imposed static shard cutoffs rather than the built-in cutoffs from the resource selection algorithms.

Jnt was trained and tested using TREC queries with 10-fold cross-validation. ReDDE parameter n was set to 1000. The Rank-S exponential decay was set to B = 1.1, which makes the score decay slower compared to prior work, so that more shards had non-zero scores to use with the static shard cutoffs. Taily scores were generated with $n_c = 400$.

Evaluation Metrics: Search accuracy was measured by P@10, NDCG@30 and MAP@1000. Evaluation was performed at each shard rank cutoff and the automatic cutoff of Rank-S and Taily. To test the proposed methods' superiority to baselines, a query-level permutation test with p < 0.05 was used. To test the equivalence to exhaustive search, a non-inferiority test [Walker and Nowacki 2011] was used to assert that results of the more efficient selective search were at least as accurate as exhaustive search. The equivalence is established by rejecting the null hypothesis that selective search is at least 5% worse than exhaustive search with a 95% confidence interval.

Selective Search Setup: We used 123 shards for CW09-B and 199 shards for Gov2 created by the QKLD-Qinit method from Dai et al. [2016]. A 1% central sample index (CSI) was created for ReDDE and Rank-S baselines and CSI based features. Jnt followed the original implementation and used a 3% CSI. The larger CSI increases the cost of resource selection for Jnt but is expected give Jnt an advantage in accuracy.

Search Engine Setup: Retrieval was performed with Indri, using default parameters. Queries were issued using the sequential dependency model (SDM) with parameters (0.8, 0.1, 0.1) [Metzler and Croft 2005]. For CW09-B, documents with a Waterloo spam score [Cormack et al. 2011] below 50 were removed ³.

² http://ir.dcs.gla.ac.uk/test_collections/access_to_data.html

³ https://plg.uwaterloo.ca/~gvcormac/

6.5 **EXPERIMENTS**

Four experiments investigated the effectiveness of learned resource selection models, and their sensitivity to training data, query lengths, and features of varying computational costs and characteristics.

6.5.1 Overall comparison

Our method was compared to four baselines and exhaustive search. We tested a variety of shard rank cutoffs ranging from 1 up to 10 for CW09-B and 16 for Gov2. Shard rankings by L2R enabled more accurate search than all baselines in both datasets (Table 8). The search accuracy of L2R models was higher than the baselines at nearly every shard rank cutoff.

Table 8 presents a comparison of L2R models and the baselines at two particular shard rank cutoffs. The first cutoff is the point where the shallow metrics (P@10 and NDCG@30) stabilize: 4 for CW09-B and 6 for Gov2. The second cutoff is where MAP@1000 become stable: 8 for CW09-B and 12 for Gov2. The automatic cutoffs of Rank-S and Taily performed similarly to fixed cutoffs and are not shown. L2R models improve over the baselines at both shard cutoffs. For shallow metrics, L2R reaches exhaustive search at the first cutoff. Furthermore, searching the first 8 out of 12 shards ranked by L2R is statistically non-inferior to searching all shards exhaustively, even for the recall-oriented MAP@1000. All the baselines are substantially worse than exhaustive search in MAP@1000.

6.5.2 Effect of training data

One might expect the relevance-based model (L2R-TREC) to be better than overlap-based models (L2R-AOL and AOL-MQT), because it uses manual relevance judgments. A model trained with overlap data might favor shards that contain false-positive documents. However, there is little difference between the two training methods. L2R-TREC was statistically significantly better than L2R-AOL or L2R-MQT for MAP@1000 in Gov2, but the relative gain is only 2%; in all other cases, there is no statistically significant differences among the three models. Furthermore, when the feature weights were examined, models trained with relevance and overlap data agreed on which features are important. This analysis indicates that unlike learning to rank document models, we can train a learning to rank resource selector on a new dataset before we have relevance judgments, positively answering RQ 6.

6.5.3 Query length

We compare L2R-MQT to the baselines using MAP@1000 for queries with different lengths on CW09-B, shown in Figure 14. Gov2 and other training data produced similar results. For single-term queries, existing methods are already equivalent to or better than exhaustive search, and L2R-MQT retains this good performance. The advantage of L2R-MQT comes from multi-term queries, where the best baseline Jnt still has a 10% gap from exhaustive search. For these queries, the improvement of L2R-MQT over the Taily is expected, because Taily does not model term co-occurrence. However, L2R-MQT also out-performs ReDDE and Rank-S, which account for term co-occurrence by retrieving documents from the CSI, but are limited by only having a sample view of the collection. L2R draws evidence from both the sample and the whole collection. Jnt

		T=4			T=8	
Method	P@10	NDCG@30	MAP@1000	P@10	NDCG@30	MAP@1000
Redde	0.355	0.262	0.176	0.363*	0.275*	0.187
Rank-S	0.350	0.259	0.175	0.360*	0.268	0.183
Taily	0.346	0.260	0.172	0.346	0.260	0.175
Jnt	0.370*	0.269	0.178	0.367*	0.277*	0.192
L2R-TREC	0.374*	0.281*	0.192	0.377*	0.286▲*	0.202▲*
L2R-AOL	0.374*	0.281▲*	0.191	0.375*	0.287▲*	0.202▲*
L2R-MQT	0.382*	0.285▲*	0.193	0.375*	0.286▲*	0.202▲*
Exh	0.372	0.288	0.208	0.372	0.288	0.208
(a) CW09-B						
		T=6			T=12	
Method	P@10	T=6 NDCG@30	MAP@1000	P@10	T=12 NDCG@30	MAP@1000
Method Redde	P@10 0.580*	T=6 NDCG@30 0.445	MAP@1000 0.267	P@10 0.587*	T=12 NDCG@30 0.460*	MAP@1000 0.289
Method Redde Rank-S	P@10 0.580* 0.570	T=6 NDCG@30 0.445 0.440	MAP@1000 0.267 0.263	P@10 0.587* 0.585*	T=12 NDCG@30 0.460* 0.461*	MAP@1000 0.289 0.286
Method Redde Rank-S Taily	P@10 0.580* 0.570 0.518	T=6 NDCG@30 0.445 0.440 0.403	MAP@1000 0.267 0.263 0.235	P@10 0.587* 0.585* 0.530	T=12 NDCG@30 0.460* 0.461* 0.418	MAP@1000 0.289 0.286 0.256
Method Redde Rank-S Taily Jnt	P@10 0.580* 0.570 0.518 0.582*	T=6 NDCG@30 0.445 0.440 0.403 0.459	MAP@1000 0.267 0.263 0.235 0.278	P@10 0.587* 0.585* 0.530 0.588*	T=12 NDCG@30 0.460* 0.461* 0.418 0.465*	MAP@1000 0.289 0.286 0.256 0.292
Method Redde Rank-S Taily Jnt L2R-TREC	P@10 0.580* 0.570 0.518 0.582* 0.593*	T=6 NDCG@30 0.445 0.440 0.403 0.459 0.469*	MAP@1000 0.267 0.263 0.235 0.278 0.299▲	P@10 0.587* 0.585* 0.530 0.588* 0.591*	T=12 NDCG@30 0.460* 0.461* 0.418 0.465* 0.475▲*	MAP@1000 0.289 0.286 0.256 0.292 0.313▲*
Method Redde Rank-S Taily Jnt L2R-TREC L2R-AOL	P@10 0.580* 0.570 0.518 0.582* 0.593* 0.593*	T=6 NDCG@30 0.445 0.440 0.403 0.403 0.459 0.469* 0.470▲*	MAP@1000 0.267 0.263 0.235 0.278 0.299▲ 0.291▲	P@10 0.587* 0.585* 0.530 0.588* 0.591* 0.587*	T=12 NDCG@30 0.460* 0.461* 0.418 0.465* 0.475▲* 0.470*	MAP@1000 0.289 0.286 0.256 0.292 0.313▲* 0.307▲*
Method Redde Rank-S Taily Jnt L2R-TREC L2R-AOL L2R-MQT	P@10 0.580* 0.570 0.518 0.582* 0.593* 0.593* 0.586*	T=6 NDCG@30 0.445 0.440 0.403 0.403 0.459 0.469* 0.470▲* 0.465*	MAP@1000 0.267 0.263 0.235 0.278 0.299▲ 0.291▲ 0.292▲	P@10 0.587* 0.585* 0.530 0.588* 0.591* 0.587* 0.593*	T=12 NDCG@30 0.460* 0.461* 0.418 0.465* 0.475▲* 0.470* 0.474▲*	MAP@1000 0.289 0.286 0.256 0.292 0.313▲* 0.307▲* 0.309▲*
Method Redde Rank-S Taily Jnt L2R-TREC L2R-AOL L2R-AOL L2R-MQT Exh	P@10 0.580* 0.570 0.518 0.582* 0.593* 0.593* 0.586* 0.585	T=6 NDCG@30 0.445 0.440 0.403 0.403 0.459 0.469* 0.470▲* 0.465* 0.479	MAP@1000 0.267 0.263 0.235 0.278 0.299▲ 0.291▲ 0.292▲ 0.315	P@10 0.587* 0.585* 0.530 0.588* 0.591* 0.587* 0.593* 0.585	T=12 NDCG@30 0.460* 0.461* 0.418 0.465* 0.475▲* 0.470* 0.474▲* 0.479	MAP@1000 0.289 0.286 0.256 0.292 0.313▲* 0.307▲* 0.309▲* 0.315

Table 8: Search accuracy comparison between 3 L2R models and baselines at two rank cutoffs (T) for two datasets. ▲: statistically significant improvement compared to Jnt, the best resource selection baseline. *: non-inferiority to exhaustive search .

also fuses sample- and term-based features, but most of its features are derived from ReDDE or Taily-like methods and do not carry new information. L2R improved over Jnt by using novel features that encode new evidence.

6.5.4 *Feature analysis*

The L2R approach uses three classes of features: query-independent, term-based, and sampledocument (CSI). These three feature classes have substantially different computational costs and contributions.



Figure 14: MAP@1000 for queries on CW09-B, grouped by query length. Parentheses on the X axis present the number of queries in each group. T is the shard rank cutoff.

Table 9: Effectiveness and efficiency of FAST features. ALL uses all features. FAST does not use sampledocument features. T: shard rank cutoff. *: non-inferiority to exhaustive.

	Method	P@10	NDCG@30	MAP@1000	Average Cost
CW09-B	Redde	0.363*	0.275*	0.187	156,180
(T=8)	Taily	0.346	0.260	0.175	470
	Jnt	0.367*	0.277*	0.192	468,710
	ALL	0.375*	0.286*	0.202*	158,529
	FAST	0.373*	0.285*	0.201*	2,349
Gov2	Redde	0.579*	0.445*	0.289	105,080
(T=12)	Taily	0.518	0.403	0.256	758
	Jnt	0.588*	0.465*	0.292	315,875
	ALL	0.593*	0.474*	0.309*	108,306
	FAST	0.587*	0.471*	0.310*	3,226

Fast vs. Slow features: Sample-document (CSI-based) features have a high computational cost, because they search a sample (typically 1-2%) of the entire corpus. Term-based features have a low computational cost, because they lookup just a few statistics per query term per shard. Costs for query-independent features are lower still. The third experiment compares a *slow* model that uses all features (ALL) to a *fast* version that does not use sample-document features (FAST).

We estimate the resource selection cost by the amount of data retrieved from storage. For CSIbased features, the cost is the size of postings of every query term in the CSI [Kulkarni 2013]. For term-based features, the cost is the amount of *sufficient* statistics required to derive all term-based features [Aly et al. 2013]. The query-independent feature only looks up the shard popularity, so the cost is one statistic per shard.

Table 9 compares FAST with ALL and baselines by their accuracy and average resource selection cost per query. ReDDE results were similar to Rank-S and are not shown. Taily has been the state-of-the-art term-based ('faster') resource selection algorithm. However, FAST is substantially more accurate. FAST also outperformed Jnt with over $100 \times$ speed up. Compared to ALL, FAST is 67

_	Feature Set	P@10	NDCG@30	MAP@1000
CW09-B	FAST	0.373	0.285	0.201
(T=8)	w/o Unigram	0.303	0.226	0.138
	w/o Bigram	0.364	0.275	0.187
	w/o Independent	0.368	0.282	0.199
Gov2	FAST	0.592	0.471	0.310
(T=12)	w/o Unigram	0.592	0.468	0.301
	w/o Bigram	0.582	0.462	0.296
	w/o Independent	0.591	0.471	0.303

Table 10: Performance of L2R-MQT using feature sets constructed with leave-one-out. 'w/o X' means the feature was excluded from FAST. Text in bold indicates the lowest value in the column.

times faster on CWo9-B and 34 times faster on Gov2. Although FAST has slightly lower search accuracy than ALL, the gap is not large and is not statistically significant, indicating that the information from the CSI features can be covered by the more efficient features.

We conclude that a resource ranker composed of only query-independent and term-based features is as accurate as exhaustive search and a ranker that includes CSI features (RQ 5). CSI features improve accuracy slightly, but at a significant additional computational cost.

Importance of Feature Types: We investigate the contribution of other types of features: queryindependent features and term-based features, where the term-based features were sub-divided into unigram and bigram features. Table 10 presents the results for the *leave-one-out* analysis conducted on FAST. On CW09-B, removing any feature set from FAST led to lower performance. This indicates that each set of features covers different types of information, and all are necessary for accurate shard ranking. Among these features, unigram features were most important because CW09-B has many single-term queries. On Gov2, the only substantial difference is observed when bigram features are excluded. This is expected as Gov2 on average has longer queries than CW09-B.

6.6 SUMMARY

This chapter investigates a learning-to-rank approach to resource selection for selective search. Much attention has been devoted to learning-to-rank documents, but there has been little study of learning-to-rank resources such as index shards.

We first consider RQ 5 and show that the learned resource selection algorithm produces search accuracy comparable to exhaustive search down to rank 1,000. This work is the first that we know of to demonstrate results that are statistically significantly equivalent to exhaustive search for MAP@1000 on an index that does not have badly skewed shard sizes. Accuracy this deep in the rankings opens up the possibility of using a learned reranker on results returned by a selective search system, which was not practical in the past.

Our investigation of RQ 6 shows that training data for learning-to-rank resources can be generated automatically using a slower system that searches all index shards for each query. This approach assumes that the goal of selective search is to mimic the accuracy of an exhaustive search system, but with lower computational cost. This assumption is not entirely true—we would like selective search to also be more accurate—but it is convenient and effective.

Prior selective search research [Kulkarni 2013; Dai et al. 2016] found that sample-document algorithms such as ReDDE and Rank-S are a little more accurate than term-based algorithms such as Taily for selective search resource selection; however, sample-document resource selection algorithms have far higher computational costs that increases query latency in some configurations (Chapter 7). This chapter suggests that sample-document features provide only a small gain when combined with other types of features. It may no longer be necessary to choose between accuracy and query latency when using a learned resource ranker.

This chapter addresses the concerns of selective search recall and strengthens the argument for the system's inclusion into a modern multi-stage retrieval architecture. In the following chapters, we present convenient tools that system administrators can use to evaluate selective search prior to implementing the system and use those tools to present additional insights into selective search performance.
7

SIMULATOR¹

Previous investigations argued that selective search is substantially more efficient than a typical distributed search architecture based on the number of postings processed when evaluating a single query stream [Kulkarni 2013; Kulkarni and Callan 2010a; Kulkarni et al. 2012]. While this metric is suitable for comparing the work done between different architectures, it does not consider how work is divided across processors, or how to compare multiple query streams being evaluated in parallel. The traditional distributed search architecture using a random assignment of documents to shards tends to spread the workload evenly across processors, and is relatively immune to bottlenecks. In contrast, a selective search architecture, which deliberately concentrates similar documents into a few index shards, might be more prone to uneven workloads, and hence risk leaving processing resources idle. Selective search might also be more sensitive to tuning parameters.

This chapter uses an event-based simulator to investigate the efficiency of the selective search architecture. A simulator makes it possible to investigate a wider range of machine configurations than would be practical in a live system. Our simulator was tuned to provide realistic measurements of query waiting times, query processing costs, query latency, system throughput, and hardware utilization under a parallel query processing environment representative of a practical real-world implementation (Section 7.2). Our investigation also extends prior work by defining a more realistic experimental methodology for studying efficiency that uses more similarly sized index shards, employs long query streams extracted from web search logs, and varies query arrival rates in order to better measure the trade-off between query latency and query throughput (Section 7.3). In particular, we present a detailed study of the computational costs, load distribution, and throughput of selective search in order to address four specific research questions:

RQ 7 Is selective search more efficient than exhaustive search in a parallel query processing environment?

RQ 8 How does the choice of resource selection algorithm affect throughput and load distribution in selective search, and how can any imbalances originating from resource selection be overcome?

RQ 9 How do different methods of allocating shards to machines affect throughput and load distribution across machines?

RQ 10 Does selective search scale efficiently when adding more machines and/or shard replicas?

Our experiments address many of the questions that have arisen in connection with the efficiency and load balancing characteristics of selective search, to both broaden and deepen our understanding of this retrieval architecture.

¹ This chapter is a lightly-revised version of Yubin Kim, Jamie Callan, J. Shane Culpepper, and Alistair Moffat. Efficient distributed selective search. *Information Retrieval*, 20(3):221–252, 2017 and reuses text written jointly by all authors. Kim designed and conducted the experiments and produced the figures unless otherwise noted.

7.1 RELATED WORK

In search and distributed systems, simulations are commonly used to evaluate the efficacy of new methods and diagnose problems because acquiring, configuring, and producing repeatable experiments with a variety of real hardware configurations and system variables is expensive and time-consuming [Webber and Moffat 2005].

Tomasic and García-Molina [1993] used a simulation to compare the performance of different physical organizations for inverted lists and evaluated the impact of various system variables on the optimal strategy. In particular they found the access time of the storage device and interprocessor communication bandwidth had a strong impact on the optimal layout.

Simulations are particularly popular in evaluating distributed systems. Cahoon et al. [2000] and Cacheda et al. [2004, 2007a,b] used simulations to evaluate distributed search system architectures and identify system bottlenecks. A range of system parameters were used and based on these experiments, they recommend an ideal architecture. Cacheda et al. [2004] placed a stronger focus on searching large text collections and Cacheda et al. [2007b] focused on a more detailed and accurate simulation of network models and outlined improvements to distributed search that can be made to reduce network bottlenecks. Our network implementation follows this detailed model. [Cacheda et al. 2007a] compared clustered and replicated distributed search architectures.

Simulated systems and workloads have been used to judge query scheduling and evaluation schemes as well [Freire et al. 2013; Moffat et al. 2006].

Badue et al. [2007] explores some of the limitations of these previous works, which assume that homogeneous servers have balanced execution times. They found that even when documents are randomly distributed, load imbalances occur due to differences in caching behavior in the shard servers that process queries, which is further exacerbated by any differences in main memory size. This work is particularly relevant to selective search, which may have an even stronger load imbalance due to the topically focused shards.

7.2 SIMULATION MODEL

A simulator was developed based on DESMO-J, a discrete event simulation modeling framework [DESMO-J]. The simulator has the benefit of allowing us to quickly investigate a range of hard-ware configurations, and provides precise estimates of a broad suite of performance indicators. The implementation models a selective search system that incorporates a cluster of multi-core machines, and mimics parallel query execution across those machines. Figure 15 describes the computational model embedded in the simulator and Table 11 lists the quantities that are manipulated.

The hardware is assumed to consist of M machines, with the i'th of those, machine m_i , providing c_i CPU cores ($c_i = 8$ throughout the chapter). Machines may be configured to act as a *broker*, or as a *searcher*, or may be configured to handle both roles.

A *broker* machine holds a copy of the resource selection database and performs two tasks: resource selection, and result merging. For resource selection, the machine has access to a shared *central query queue*, from which it extracts incoming queries, determines which shards need to be searched, and then assigns shard search tasks to other machines. Each broker machine also has a job queue for pending *result merge* processes. This queue contains results returned by the searcher machines, now waiting to be merged to produce a final result list for some query.



Figure 15: Architecture of the selective distributed search system. The i'th of the M machines has c_i cores, each of which can be used for resource selection and result merging, or for shard search across any of the p_i shards allocated to this machine. Only a defined subset of the machines are able to perform resource selection. Figure by J. Shane Culpepper, reused from Kim et al. [2017].

A machine m_i is a *searcher* if it is allocated $p_i > 0$ selective search shards. A searcher also has a job queue that holds *shard search* requests pertinent to the shards hosted on that machine. Each of the available cores on the machine can access any of the shards assigned to the machine, and hence can respond to any request in that machine's search queue. When a search job is finished, the result is returned to the result merge queue of the originating broker. The assignment of shards and copies of the resource selection database to machines is assumed to be fixed at indexing time, and machines cannot access shards that are not hosted locally. A key factor for success is thus the manner in which the P shards are partitioned across the M machines.

Algorithm 2 describes the sequence of actions that take place in each of the machines. First, if the machine is a broker, the local result merge queue is checked for queries for which all shard searches have been completed, and merged output lists are generated if any queries can now be finalized. Otherwise, if the machine is a searcher, the local shard search queue is checked to see if there are any shard searches pending; if so, the next one is extracted and actioned, and the results directed to the merge queue for the machine that acted as broker for that query, which might be the same machine. A shard search process on machine m_i can search any shard assigned to that machine.

- M Number of machines; m_i is the i'th of these.
- c_i Number of cores on m_i ; the default is $c_i = 8$.
- C Total number of cores, $\sum_{i=1}^{M} c_i$.
- p_i Number of shards assigned to m_i.
- P Total number of shards. When each shard is assigned to just one machine, $P = \sum_{i=1}^{M} p_i$.
- B Number of broker machines.
- S Number of searcher machines.
- T Query arrival rate described by an exponential distribution with mean $1/\lambda$, T = λ .
- t_s Seek plus latency access time, milliseconds per postings list, $t_s = 4$ throughout.
- t_p Processing cost, milliseconds per posting, $t_p = 9 \times 10^{-4}$ throughout.
- t_m Merging cost, milliseconds per item, $t_m = 5 \times 10^{-5}$ throughout.

Algorithm 2 – Processing loop for each core on machine m_i. All authors contributed to this figure.

while forever do
if $isBroker(m_i)$ and $ mergequeue_i > 0$ and all shard responses received for q then
remove those responses from mergequeue _i
finalize the output for query q and construct a document ranking
else if $isSearcher(m_i)$ and $ searchqueue_i > 0$ then
remove a query request (q, p, b) from searchqueue _i
perform a shard search for query q against shard p
append the results of the search to mergequeue _b
else if $isBroker(m_i)$ and $ centralqueue > 0$ then
remove a query q from centralqueue
perform resource selection for q
for each partition p to be searched for q do
determine the machine m_h that is host for p
append (q,p,i) as a search request to searchqueue _h
endwhile

If neither of these two activities are required, and if the machine is a broker, the next query (if one exists) is taken from the central query queue and resource selection carried out. The result of resource selection is a list of shards to be searched in order to resolve the query; that list is mapped to a set of machine identifiers, and the query q is added to the shard search queues for those machines. The processing order for these operations prioritizes query completion over query initiation. This minimizes the processing time for each query, and ensures that no query has an infinite wait time.

In the simulation, queries are assumed to arrive at the central queue at random intervals determined by an exponential distribution governed by a mean query arrival rate T. The number of machines permitted to host broker processes may be less than M, the total number of machines, but is always at least one. Query processing costs at the shards are computed based on the number of postings processed from the shard index, plus an overhead cost to account for initial latency for a disk seek. A postings list of ℓ postings is thus presumed to require $t_s + \ell \cdot t_p$ milliseconds, where $t_s = 4$ milliseconds [Shimpi], and $t_p = 9 \times 10^{-4}$ milliseconds per posting. The processing rate – around a million postings per second – is based on measurement of the cost of handling posting lists in the open source Indri search engine [The Lemur Project] on a machine with a 2.44 GHz CPU, and encompasses I/O costs as well as similarity calculation costs. In the case of in-memory execution, t_s can be set to zero. The parameter t_s can also be used to selectively emulate caching of postings lists. For simplicity, we assume all postings are cold cached in the experiments presented in this chapter.

Resource selection costs differ according to the approach used. For sample-based algorithms such as ReDDE or Rank-S, the cost is dominated by the need to process postings from the central sample index (CSI). For these approaches, the same computational model is used as for shard search, and a cost of $t_s + l \cdot t_p$ milliseconds is assumed for a postings list of length l. On the other hand, term-based algorithms such as Taily process statistics from each shard that contains a query term. The cost for term-based approaches is thus equivalent to processing a posting list of length equal to the number of shards that contain the query term, which is always less than or equal to P, the number of shards.

Result merging may require network transfer if the results are returned to a broker that is not located within the same machine as the shard.

This requires transferring of up to k $\langle doc-id, score \rangle$ results returned from each shard searched, where k is either fixed on a system-wide basis, or is determined as part of the resource selection step. Network messages are also generated when brokers request searches for shards that are not stored within the same machine. To ensure that the simulation was accurate, the cost of network communication over a Gigabit switched network was modeled as described by Cacheda et al. [2007b]. The cost of merging was measured on the same 2.44 GHz machine, and an allocation of $t_m = 5 \times 10^{-5}$ milliseconds per document was found to be appropriate.

The system is configured to return the top 1,000 documents to support applications such as learning to rank algorithms, text-mining applications, and TREC evaluations that need deeper rankings. Cacheda et al. [2007b] showed that when shards are formed randomly, only a small number of documents need to be returned from each shard for the true top-k documents to be returned with a high probability. Therefore, in the exhaustive search system used as a baseline, each shard only returns the number of documents that results in a 10^{-5} probability of missing a result in the top 1,000. This equates to returning 102 documents per shard for a 16 shard configuration and 19 documents per shard for 512 shards. The assumption that documents are distributed randomly across shards does not apply to the topical indexes used by selective search; clustering concentrates similar documents in a small number of shards. Thus, Cacheda's technique cannot be used with selective search and each selective search shard returns 1,000 documents. Since the exhaustive search baseline accesses all shards, whereas selective search typically accesses 3-5 shards, the total number of documents that must be merged by the two architectures is roughly comparable. In our experiments, the total number of documents merged by the two architectures usually varied between 1,600 and 9,700, depending upon the number of machines (exhaustive search) and query (selective search). Generally, exhaustive search merges fewer documents than selective

search in configurations with fewer machines (and random shards), and more documents than selective search in configurations with more machines (and random shards).

Overall, the simulator takes as input a list of queries, the resource selection cost for each query, the shards to be searched for the query, and the search cost for each shard. The cost is described by the total length of the posting lists retrieved for all query terms for the specified shard or resource selection method. The simulator then converts these posting list costs into "simulator milliseconds". The simulator calculates the overall elapsed time required to process each query as the difference between the arrival time of that query in the central queue, and the moment at which all processing of that query is completed. This cost includes time spent waiting in queues, network delays, and computation time. The median end-to-end elapsed query processing time is used as the primary measure of query latency, but we also show the latency distributions in some key cases. The principal control variable in the simulator is the query arrival rate, which determines the load in the system, and hence the extent to which query response times are affected by queuing delays.

The simulator also tracks the load on each simulated machine. The load is measured by the utilization of the simulated cores, as a time-weighted fraction of available capacity that is utilized for query computation. The output of this measurement was used to evaluate the evenness of load across the simulated machines.

7.3 EXPERIMENTAL METHODOLOGY

The simulator was used for a detailed evaluation of selective search schemes, applying them to two large experimental collections. This section describes the data resources employed, including a revised partitioning scheme for the experimental collections that constructs more similarly-sized partitions, and gives details of the effectiveness baselines we compare against.

7.3.1 Document collections

Two web datasets, ClueWebo9 Category A English (abbreviated to CWo9-A) and Gov2 were used in the experimentation. These are large collections, covering more than 500 million and 25 million web pages respectively. The selective search configurations applied to them are derived from the shard definitions for CWo9-A and Gov2 generated as part of the investigation carried out by Kulkarni [2013]. Those investigations established the validity of the selective search approach, and explored clustering techniques and resource selection processes in order to determine how to achieve search effectiveness comparable to exhaustive search.

One issue with the CWo9-A shard map produced by Kulkarni is that the shard sizes can vary. Figure 16 plots the distribution of shard sizes, and shows that the largest one is 12 times the average size. This imbalance is problematic for two reasons. Firstly, there is a significant correlation between the size of the shard and the frequency at which it is selected for querying (Pearson's correlation of 0.5). That is, the larger the shard, the more likely it is to be selected for querying. Secondly, when they do get selected, large shards typically take longer to process queries, because of their size. In combination, these two effects produce an imbalance in computation load which is not present in the Gov2 shards.

To address the imbalance of shard sizes in CW09-A, the largest shards in the original shard partitioning of Kulkarni were divided; large shards were split via a random assignment when the resulting sub-shards would be closer to the average shard size (by number of documents) than



Figure 16: Sizes of shards used in experiments reported by Kulkarni [2013].

Table 12: Resource selection parameter settings, based on TREC Terabyte Track and TREC Web Track queries for which relevance judgments are available, as developed by Kulkarni et al. [2012] and Aly et al. [2013]. Those investigations used a variety of CSI sizes for Gov2, ranging from 0.5% to 4%. We used 1%, for consistency with CW09-A. The table also presents the average number of shards selected by the parameter settings.

Dataset	Selection algorithm	Parameters	Average shards searched
Gov2	Taily	n = 400, v = 50	3.0
	Rank-S	CSI = 1%, base = 3	4.7
CW09-A	Taily	n = 400, v = 50	3.3
	Rank-S	CSI = 1%, base = 5	4.2

the complete shard. A total of 51 shards were split into two or more smaller shards, resulting in an increase of 77 shards, for a total of 884. Gov2 used the 50 shard definition that was used by Kulkarni [2013].

7.3.2 *Resource selection parameter settings*

Two resource selection mechanisms are used in our experiments: Rank-S [Kulkarni et al. 2012] and Taily [Aly et al. 2013]. The parameters used in our experiments are as suggested by Aly et al. [2013] and Kulkarni [2013], and are summarized in Table 12. Note that the parameters for Gov2 are slightly different from the ClueWeb09 parameters.

Table 13 and Table 14 list the overall effectiveness of selective search using the default parameters, using queries that span multiple years of the TREC Terabyte and Web Tracks. We use Precision at rank 10 (P@10) and Normalized Discounted Cumulative Gain at rank 30 (NDCG@30) to measure the effectiveness of the systems, which makes these results comparable to prior work [Aly et al. 2013; Kulkarni et al. 2012]. These metrics are precision-oriented, that is, they place importance in the top of the ranked result list.

Algorithm	Teraby	te Track 2004	Teraby	e Track 2005	Teraby	te Track 2006
Algorithm	P@10	NDCG@30	P@10	NDCG@30	P@10	NDCG@30
Baseline	0.56	0.43	0.62	0.49	0.57	0.48
Rank-S	0.57	0.42	0.59	0.45	0.54	0.44
Taily	0.51	0.37	0.48	0.35	0.50	0.42
Oracle	0.64	0.47	0.64	0.51	0.61	0.51

Table 13: Effectiveness of selective search using various resource selection algorithms on the Gov2 dataset. The "Oracle" shard ranking assumes that the most useful four shards are identified and searched for each query.

Table 14: Effectiveness of selective search using various resource selection algorithms on the CW09-A dataset. The "Oracle" shard ranking assumes that the most useful four shards are identified and searched for each query.

Algorithm	Web Track 2009		Web Track 2010		Web 7	Track 2011	Web Track 2012		
Aigoinniin	P@10	NDCG@30	P@10	NDCG@30	P@10	NDCG@30	P@10	NDCG@30	
Baseline	0.30	0.21	0.27	0.19	0.36	0.28	0.27	0.15	
Rank-S	0.28	0.19	0.32	0.20	0.30	0.21	0.25	0.14	
Taily	0.28	0.18	0.31	0.20	0.24	0.16	0.23	0.13	
Oracle	0.37	0.25	0.45	0.32	0.42	0.35	0.33	0.20	

Note that the effectiveness results are independent of the number of machines and how shards are assigned to machines. Once the assignment of documents to shards has been completed, and the resource selection parameters determined, selective search generates the same final ranked list regardless of the number of machines or shard allocations.

That fact means that these effectiveness results apply uniformly to all of the efficiency experiments we report. The baseline scores reported for exhaustive search were generated using structured queries and the sequential dependency model (SDM) [Metzler and Croft 2005], removing spam documents from the final result using the Waterloo Fusion spam scores [Cormack et al. 2011] at a threshold of 50%. The selective search runs use a similar set-up whenever possible; for Rank-S, the CSI was searched using SDM queries. For both Taily and Rank-S, the selected shards were searched using SDM queries and the result list was filtered for spam. These arrangements provide a more effective baseline than that used by Kulkarni [2013], and one that is comparable to the baseline used by Aly et al. [2013], producing the strongest possible baseline run.

In addition to the 2009 and 2010 Web Track queries used in previous investigations, our experiments include query sets from the 2011 and 2012 Web Tracks. Results for query sets are reported separately rather than averaging across query sets, in order to distinguish the effects of different tasks on selective search. Notable in these results is that selective search is not always as competitive as previously reported, particularly for the CWo9-A dataset.

In the results reported by Kulkarni [2013] and by Aly et al. [2013], effectiveness was averaged across the 2009 and 2010 Web Track query sets. However, the two query sets have different characteristics. The 2010 queries tend to be more difficult, resulting in lower effectiveness from the

Dataset	Log	Train	Test	Test _W	$\operatorname{Test}_{\mathcal{M}}$
Gov2	AOL _G	1K	10K	10K	10K
Gov2	MQT _G	1K	10K		
CW09-A	AOL_W	1K	10K	10K	10K
CW09-A	MQT_W	1K	10K		

Table 15: Sizes of the query logs, in thousands. The "Train" column indicates training queries used to set parameters. The "Test" column indicates testing queries used for reporting results. Columns "Test_W" and "Test_M" indicate queries that were sampled starting one week and one month after the Train queries, respectively. Those two query streams are used in Section 7.4.3.

exhaustive search baseline. The exhaustive search system had difficulty separating relevant documents from similar-looking non-relevant documents. However, selective search was less affected because clustering assigned these documents to different shards, and then resource selection ignored the shards that contained the similar-looking non-relevant documents. Thus, as observed by Powell et al. [2000], a partitioned index and good resource selection became more effective than exhaustive search. This effect is further studied in detail in Chapter 3.

When the 2010 results are averaged with the 2009 results, the weaker 2009 results are masked, and it appears as if selective and exhaustive search have comparable performance. While the accuracy results are reported with our modified shard maps, these results hold when using the original shard map with very little difference in accuracy.

The effectiveness results reported in Table 13 and Table 14 suggest that the parameters used for Taily and Rank-S might not be optimal. To address that concern, we performed a parameter sweep to check for better settings, but were unable to identify parameters that both yielded accuracy comparable to exhaustive search and also searched a moderate number of shards. We also experimented with a clairvoyant "oracle" shard ranking, in which the four shards containing the greatest number of relevant documents were always presumed to always be selected. Those hypothetical results are shown in the final row of Table 13 and Table 14. The marked gap between the attained effectiveness and the goal set by the oracle system indicates that the accuracy of selective search is a function of the resource selection process, and makes it clear that further algorithmic improvements may be possible. This subject is later explored in Chapter 3 and Chapter 6. We use the configurations reported in Table 12, and in the remainder of this chapter focus solely on efficiency.

7.3.3 Query streams

A key contribution of this work is the measurement of the performance of selective search under realistic query loads, and determination of the point at which each configuration saturates. For these experiments a much larger query log is needed than the Web Track and Terabyte Track query sets used in the effectiveness study. The experiments that are the main focus of this chapter make use of the AOL query log² and the TREC Million Query Track query set [Carterette et al. 2009]. These query logs are from a different timespan than the collections, but the queries are

² Queries span March through May 2006.

66

only used to measure efficiency, not effectiveness, and the small discrepancy in their temporal coverage is not a concern.

In order to simulate a live query stream, the AOL log was sorted by timestamp, and deduplicated to only contain unique queries. Deduplication is performed because a production system would serve repeated queries from a cache, and most queries would be seen only once by the search engine. For CW09-A, the first 1,000 queries from the log were then used as training data to set configuration parameters such as B, the number of brokers (Section 7.4.2), and assignments of shards to machines (Section 7.4.3). The next 10,000 queries were used for testing. Together, the queries cover a time period of 2006-03-01 00:01:03 to 2006-03-01 01:31:45. For Gov2, the timestampsorted AOL query stream is also filtered to form a query stream where there is at least one .govdomain result click recorded for each query. The first 1,000 queries were again used as training data, and the next 10,000 queries used for testing, covering the time period 2006-03-01 00:01:08 to 2006-03-01 20:09:09. Two further test query sets were also extracted from the AOL log: 10,000 queries starting from 2006-03-08, one week after the main query stream (Test_W); and another 10,000 queries commencing 2006-04-01, one month after the main query stream (Test_M). These two additional test sets were used in the experiments described in Section 7.4.3.

Another pair of query streams was created using the TREC Million Query Track (MQT) queries. The MQT queries have no timestamps and were used in the order they appear in the files provided by the National Institute of Standards and Technology (NIST). For Gov2, the first 1,000 queries of the 2007 query set were used for training; 10,000 queries from the 2008 query set were used for testing. For CW09-A, both the 1,000 training and 10,000 testing queries were extracted from the TREC 2009 MQT sequence. In total, 84,000 queries were used at various stages of the evaluation, split across the categories summarized in Table 15.

7.4 EXPERIMENTAL RESULTS

Our experiments investigate the efficiency and load characteristics of a selective search architecture under a variety of conditions. The first set of experiments evaluates an environment similar to the one proposed by Kulkarni [2013], but allowing for parallel execution of queries and tasks (Section 7.4.1). The second suite of experiments explores resource selection costs, resource selection algorithms, and how to overcome any computational load imbalances originating from resource selection (Section 7.4.2). The third round of experiments investigate the load and throughput effects of two different policies for assigning shards to machines (Section 7.4.3). The last experiment compares index-spreading and mirroring strategies when using additional computational resources (Section 7.4.4).

7.4.1 Selective search efficiency

In this section, we address RQ 7 – Is selective search more efficient than exhaustive search in a parallel *query processing environment*? The selective search architecture is compared to a typical exhaustive search architecture in small-scale environments similar to those examined by Kulkarni and Callan [2010a] and Kulkarni [2013]. Each test collection was divided into topical shards and the shards randomly distributed across all machines, with each machine receiving the same number of shards. The same collections were also used to build an exhaustive search baseline by constructing C (the total number of cores) evenly sized shards via a random assignment of documents, and then allocating one shard per core to each machine; $c_i = 8$ in our experiments, thus 8 shards were

Query log	Resource selection	G	lov2	CW	CW09-A		
Query log	Resource selection	avg	sddev	avg	sddev		
MQT	Taily	2.5	1.4	3.6	2.9		
	Rank-S	4.2	1.8	4.4	1.7		
AOL	Taily	2.9	1.6	11.9	31.3		
	Rank-S	4.6	1.9	4.2	1.7		

Table 16: Average number of shards selected by Taily and Rank-S for the Test logs, measured using the configurations depicted in Figure 17.



Figure 17: Exhaustive search and selective search using CW09-A and Gov2 and random shard assignment for the AOL_W Test and MQT_W Test, AOL_G Test and MQT_G Test query sets (shown as AOL and MQT in the labels for brevity). In these experiments, M = 2, B = 1, and S = 2.

assigned to each machine. This allows exhaustive search to minimize latency by making use of all cores for every query. In both configurations, only one machine (B = 1) accepted broker tasks, so as to emulate previous arrangements as closely as possible.

Table 16 summarizes the average number of shards selected by the resource selection algorithms with parameters as discussed in Section 7.3.2. Note the high variance in the number of shards selected by Taily for the AOL query set used with CWo9-A.

When the AOL query log is used with the CWo9-A collection, the distribution of Taily shard scores levels off quickly and becomes flat, which makes it difficult for Taily to distinguish between shards. This causes the mean and standard deviation of the number of shards selected to be much higher than in other configurations. The AOL query set used with Gov2 does not experience the same issues; these queries are cleaner due to the filtering process documented in Section 7.3.3. Even with these variances affecting behavior, selective search examines only a small fraction of available shards and produces significant efficiency gains regardless of the resource selection method deployed.

68

Table 17: Dispersion of per-query processing times for selected configurations of exhaustive, Rank-S and Taily methods, at the last point on each of the corresponding curves in Figure 17. Columns display the 50th, 75th, 99th percentiles, the mean, and the maximum value of individual query processing times, where time is measured in simulation milliseconds. In these experiments, M = 2. Recall that the simulation was tuned to model the costs associated with Indri, a research search engine; a commercial system would likely have lower processing times.

Collection	Collection Queries Method $1/\lambda$ Percentile					e	Moon	Мах
Collection	Queries	Method	(qry/s)	50%	75%	99%	wiean	IVIAX
Gov2	MQT	Exh	0.28	2,268	4,409	11,676	2,936	20,868
		Rank-S	3.20	1,632	2,651	6,206	1,883	10,229
		Taily	5.50	1,189	1,901	4,579	1,366	8,145
	AOL	Exh	0.42	250	1,126	5,016	768	9,135
		Rank-S	7.50	397	865	2,606	590	5,352
		Taily	12.00	265	612	1,868	407	4,329
CW09-A	MQT	Exh	0.04	5,819	26,510	110,232	17,784	206,595
		Rank-S	3.20	1,468	3,537	13,082	2,516	32,888
		Taily	9.00	510	1,012	3,340	719	6,709
	AOL	Exh	0.03	21,736	50,525	172,978	34,019	427,116
		Rank-S	1.80	3,263	6,220	18,351	4,208	76,103
		Taily	0.75	902	3,869	21,350	3,032	46,770

Figure 17 shows the throughput of two selective search variants, compared to exhaustive search. In each of the frames the vertical axis shows the median time to process each query, plotted as a function of the query arrival rate on the horizontal axis. Alternative summary metrics such as the mean or the 95% percentile processing times were also explored, and produced similar outcomes. Each plotted curve represents a combination of query set and processing regime, and the two frames correspond to the two collections. The elbow in each plotted curve indicates the point at which that system configuration approaches saturation. To avoid extreme results, each curve was truncated at the point at which the median processing time for queries exceeded twice the median processing time for queries in a unloaded system, determined by the latency at 0.01 queries per second, at the left-hand end of the curve. Query arrival rates were chosen dynamically for each curve, so that the region of greatest gradient can be clearly distinguished. Configurations where the curve is lower have lower latency. Configurations with elbows that are further to the right require fewer resources per query and attain higher throughput rates when the system is approaching saturation.

To describe the variation of query processing times, Table 17 presents the 50th, 75th and 99th percentile, mean, and max values of query processing time for several configurations of selective search. The dispersion of query processing time is computed at the last point in the plotted curves of Figure 17 – the point where the system is approaching saturation.

Table 18 presents the breakdown of average costs for queries at the system saturation point for the configurations studied. The network costs and merging costs are minimal. As expected of a system under full load, queries spend a significant amount of time in the central query Table 18: Breakdown of the average time spent processing queries, at the last point on each of the corresponding curves in Figure 17. *Central queue* indicates the percentage of time spent in the central query queue; *network* indicates the delays caused by network congestion; *internal queues* indicates time spent in system queues such as a machine's shard search queue or merge queue; *resource selection* indicates time spent performing resource selection; and *merge* is time spent performing the final merge operation. The remaining time not included in these categories was spent in shard search.

Collection	Oueries	Method	$1/\lambda$	Central	Network	Internal	Resource	Merge
	Querres	1110011001	(qry/s)	queue	1.00000110	queues	selection	
Gov2	MQT	Exh	0.28	21.88%	0.06%	14.91%	-	0.09%
		Rank-S	3.20	32.16%	< 0.01%	3.35%	13.38%	0.01%
		Taily	5.50	30.38%	<0.01%	3.44%	1.58%	0.01%
	AOL	Exh	0.42	21.41%	0.07%	16.92%	-	0.01%
		Rank-S	7.50	25.67%	0.01%	2.58%	14.67%	0.04%
		Taily	12.00	20.55%	0.01%	2.68%	2.50%	0.03%
CW09-A	MQT	Exh	0.04	31.57%	<0.01%	25.24%	-	<0.01%
		Rank-S	3.20	25.55%	<0.01%	1.72%	53.44%	0.01%
		Taily	9.00	26.36%	0.01%	5.10%	1.66%	0.02%
	AOL	Exh	0.03	30.86%	<0.01%	31.08%	-	<0.01%
		Rank-S	1.80	22.40%	<0.01%	1.93%	59.67%	<0.01%
		Taily	0.75	36.77%	<0.01%	7·54%	0.46%	0.02%

queue, and in queues internal to each system. Note the differences in the time spent in resource selection between Rank-S and Taily. Systems using Rank-S, a sample-based algorithm, spent a larger portion of time performing resource selection, particularly when using CW09-A. This also produces higher latency for Rank-S systems as seen in Figure 17. This result is expected because sample-based resource selection methods are more expensive than term-based methods. This topic is explained in more detail and further explored in Section 7.4.2.

When querying the CWo9-A collection (Figure 17b), selective search outperforms exhaustive search by a factor of more than ten, because only a small fraction of the 884 shards are searched. Query latency is also lower in selective search, despite the two-step process of resource selection followed by shard search. This is due to fewer resources being used by selective search, and the fact that at low query loads, latency is largely determined by the slowest partition that is polled. Topical shards are smaller than random shards, thus they can be searched more quickly. A larger fraction of the fifty possible shards are searched in the Gov2 collection (Figure 17a), so the performance improvement is not as high. Even so, selective search of Gov2 handles four to five times the rate of queries as exhaustive search before saturating.

Taily has better throughput and latency than Rank-S for the majority of settings tested. The only exception is for the AOL_W Test query set in CW09-A, where the larger number of shards searched by Taily eclipses the lower resource selection costs, resulting in better throughput for Rank-S. More broadly, selective search delivers markedly better performance characteristics than exhaustive search in all of the configurations investigated, reaffirming the findings of Kulkarni



Figure 18: Average utilization of machines using Rank-S, the CW09-A dataset, the AOL_W Test queries, M = 2, B = 1 and S = 2. Each point is the mean over 10 sequences of 1,000 queries; error bars represent 95% confidence intervals.

and Callan [2010a] and Kulkarni [2013] that selective search is a viable framework for improving the efficiency of parallel query processing.

7.4.2 *Resource allocation*

Figure 17 indicates that Taily is more efficient than Rank-S, with significantly better query latency and throughput in the majority of configurations tested. The difference is a direct consequence of the two approaches to resource selection. Aly et al. [2013] demonstrated that resource selection costs can be a substantial portion of the total retrieval cost in a system that uses sample-based resource selection algorithms, such as ReDDE and Rank-S. We now examine in detail how the different resource selection computations affect throughput and load of selective search, across a range of different system configurations in order to answer RQ 8 – How does the choice of resource selection algorithm affect throughput and load distribution in selective search, and how can any imbalances originating from resource selection be overcome?

In the experiments illustrated in Figure 17, only one machine acted as broker, and the shards were evenly distributed across the two machines (B = 1, S = 2). This is similar to the configuration described by Kulkarni [2013]. However, this configuration is not optimal for selective search, and produces an uneven machine load, especially when using Rank-S to select shards. Figure 18 plots the machine utilization of this configuration, using Rank-S, M = 2 machines, and searching CW09-A with the AOL_W Test query set. All broker tasks are handled by machine m_1 , and this machine is also a searcher. Consequently, m_1 is heavily loaded when compared to the other machine, and the majority of its load is generated by resource selection. A detailed examination of the m_1 workload reveals that resource selection is responsible for more than 70% of the machine's computational load when the arrival rate is 2.5 queries per second. Merging requires near-zero load. A similar imbalance is observed when other simulator parameters such as query streams, collections, and hardware are varied.

We then configured the machines in the Rank-S system to allow more broker processes. When tuning a selective search system, this decision can be based on training queries. We compare



Figure 19: Varying the resources assigned to broker processes using Rank-S, with M = 2, and random shard assignment.

the results of B = 1 to B = 2, a configuration where both machines perform resource selection. Figure 19 shows these parameter settings when applied to the Test queries and the Rank-S MQT_G Test and AOL_W Test configurations. When M = 2, the best Rank-S settings were B = 2 for Gov2 with an even distribution of shards for both MQT_G Train and AOL_G Train logs. This configuration gives small improvements in throughput over the baseline of using B = 1. For Rank-S on CW09-A, B = 2 was also the best configuration for AOL_W Train and MQT_W Train, but the throughput improvements are much larger. The difference in behavior for these datasets is due to the size of the CSI. Rank-S uses a 1% sample of the corpus to make decisions. For Gov2, a 1% sample is about half the size of an average shard, but for CW09-A, 1% is about eight times the size of an average shard. Most of the work for CW09-A is devoted to resource selection while for Gov2 most of the work is dedicated to shard search, as was seen in Table 18. Similar results were obtained using Rank-S with other settings.

Taily requires far less computation for resource selection than does Rank-S, and the best setting was B = 1 for all query logs, and all datasets. At M = 2 and B = 1, resource selection for Taily accounted for less than 2% of m_1 's processing capability. This illustrates the advantage of termbased algorithms – efficiency. In most configurations, Taily also consumes fewer computational resources overall than Rank-S. As is shown in Figure 17b, the resulting performance gains can be large.

Overall, the choice of resource selection algorithm has a significant impact on the throughput, latency, and load distribution of selective search; and unless care is taken, load imbalances can emerge even when just a few machines are being used. In the larger CW09-A dataset, Taily had significant latency and throughput advantages compared to Rank-S, because of the high cost of searching the larger Rank-S CSI. However, sample-based algorithms have other advantages, including the ability to run structured queries.



Figure 20: Popularity of shards in the CW09-A training query sets for the top 10% of most frequently accessed shards as selected by Rank-S. The vertical axis indicates the total number of times a shard is selected for all queries.

7.4.3 Shard assignment policies

The preceding experiments assigned shards to machines randomly. With two machines (M = 2), random assignment of shards to machines distributes query traffic evenly across machines, since there are many more shards than machines ($P \gg M$). However, this assumption does not hold as M increases because selective search deliberately creates shards that have skewed term distributions and topical coverage. So, in this section we examine RQ 9 – *How do different methods of shard allocation affect throughput and load distribution across multiple machines*?

Figure 20 shows the relative popularity of the 80 most used CW09-A shards, and demonstrates that the skewed term distributions means that some shards are selected more often than others. For example, when the Rank-S resource selection algorithm is used to select topic-based shards for the Gov2 queries from AOL_G Train query set, the five most frequently selected shards account for 29% of all shards selected – 1,302 out of 4,491 shard selections. In CW09-A, the AOL query log also exhibits a strongly skewed popularity pattern. The MQT query set displays a milder, but still noticeable, skew. This unevenness of popularity has the potential to adversely affect throughput in a system distributed over multiple machines by overloading machines that are responsible for popular shards and creating bottlenecks that then starve other machines of work. Even a small increase in the number of machines to M = 4 can lead to a load imbalance.

The next set of experiments investigates the effectiveness of *Random* assignment, and compares it with a *Log-based* algorithm, which uses training data to estimate and balance the average load across machines. The Log-based approach takes a set of training queries, performs resource selection, and computes the sum of the postings costs for each shard. That sum is used as an estimate of the shard's load. Shards are then ordered most to least expensive, and assigned to machines one by one, in each case choosing the machine that currently has the lowest estimated load. Finally, any remaining shards that were not accessed by the training queries are assigned based on size. In these experiments a set of 1,000 training queries was employed. Using Taily's shard selections, Gov2 had no unaccessed shards for either query set, and 7% and 4% of shards were unaccessed for AOL_W Train and MQT_W Train respectively in CWo9-A.



Figure 21: Utilization of machines for selective search on Gov2, using Taily, the MQT_G Test queries, M = 4, B = 1, S = 4 and a uniform shard distribution. Each point is the mean over 10 sequences of 1,000 queries; error bars represent 95% confidence intervals. The broker resides in m_1 for both configurations. The two frames share a common legend.

Table 19: Average load_i and range (max load_i – min load_i), where load_i is the time-averaged CPU load of m_i during the simulation, for two shard allocation policies and a range of query arrival rates, using Taily resource selection, M = 4, B = 1, and S = 4. Results for Gov2 using MQT_W Test queries are also shown in Figure 21.

Dataset	Oueries	Allocation Avg load i and range of load i for each query arr							y arrival	rate T
Dutuset	Queries	method	avg	rnge	avg	rnge	avg	rnge	avg	rnge
			6 q	ry/s	8 q	ry/s	10 0	ry/s	12 (ry/s
Gov2	MQT _G Test	Random	0.32	0.30	0.42	0.40	0.53	0.49	0.61	0.58
		Log-based	0.32	0.10	0.42	0.14	0.53	0.17	0.63	0.21
			20 q	ry/s	30 q	ry/s	40 9	ry/s	50 0	qry/s
Gov2	AOL _G Test	Random	0.40	0.19	0.59	0.28	0.73	0.34	0.78	0.36
		Log-based	0.40	0.14	0.59	0.20	0.73	0.25	0.78	0.26
			10 q	ry/s	15 q	ry/s	20 0	ry/s	25 0	ry/s
CW09-A	MQT _W Test	Random	0.36	0.08	0.53	0.11	0.70	0.15	0.81	0.17
		Log-based	0.36	0.05	0.53	0.07	0.70	0.09	0.82	0.10
			1.5 0	[ry/s	2.0 0	lry/s	2.5 0	qry/s	3.0	qry/s
CW09-A	AOL _W Test	Random	0.39	0.07	0.52	0.10	0.65	0.12	0.76	0.14
		Log-based	0.39	0.01	0.52	0.02	0.65	0.02	0.77	0.03

In this experiment, we focus only on the performance of Taily. The results for Rank-S are similar. Figure 21 shows the effect of the shard assignment policies on machine utilization as query arrival

Table 20: Average load_i and range (max load_i – min load_i) of shard search machine utilization as the training data ages, using the Log-based shard allocation policy. Other simulation settings are as described in Table 19. Test queries begin immediately after the training queries. Test_W and Test_M queries begin from 1 week and 1 month after the training queries, respectively. The MQT queries do not have timestamps and so were not used in this experiment.

Dataset	Quarrylag	Avera	Average load $_{i}$ and range of load $_{i}$ for each query arrival rate T						
Dataset Query log		avg	rnge	avg	rnge	avg	rnge	avg	rnge
		20 q	ry/s	30 q	ry/s	40	qry/s	50	qry/s
Gov2	AOL _G Test	0.40	0.14	0.59	0.20	0.73	0.25	0.78	0.26
	AOL _G Test _W	0.38	0.13	0.56	0.19	0.71	0.24	0.76	0.25
	AOL_G Test _M	0.32	0.14	0.47	0.20	0.62	0.26	0.72	0.30
		1.5 0	ry/s	2.0 0	[ry/s	2.5	qry/s	3.0	qry/s
CW09-A	AOL _W Test	0.39	0.01	0.52	0.02	0.65	0.02	0.77	0.03
	AOL_W Test _W	0.39	0.01	0.51	0.02	0.63	0.02	0.75	0.02
	AOL_W Test _M	0.40	0.02	0.53	0.02	0.66	0.02	0.77	0.03

rates are varied for the Gov2 dataset using the Million Query Track queries. Machine utilization (or *load*) varies from 0–100% (shown as 0.0 to 1.0). The range of machine loads show that the Random policy produces an uneven utilization of machines, and saturates relatively quickly due to a bottleneck on m₄. In comparison, the Log-based policy reduces variance of load noticeably, and produces a more even utilization of the machines. Detailed results for other configurations are given in Table 19, and display similar usage patterns. In all settings, Log-based assignment produces more uniform resource utilization than does Random.

The risk of using Log-based allocations is that the learned attributes may become outdated as a result of changes in the query stream. Table 20 investigates this potential shortcoming by showing machine usage when processing three query sets each containing 10,000 queries: one that arrived immediately after the training queries; a second set that arrived one week after the training queries; and a third set that arrived one month after the training queries. Average utilization is similar in each case, and variance increases marginally as the query stream evolves, but the changes are generally small. Results for the AOL log for Gov2 one month after assignment are an exception and have a markedly lower utilization due to a burst of shorter queries that occurred at this time (2.18 words on average versus 2.41 and 2.44 for Test_W and Test queries respectively), but the variance still remains similar. Shard assignments should be periodically revised to maximize throughput, but it is not necessary to do it frequently, and the cost of periodically refreshing shard assignments can be amortized.

An additional concern is that the Random allocations used in Figure 21 and Table 19 may not be representative of the distribution of all possible allocations, as there could be significant variation between two Random allocations. One allocation could have nearly perfect organization of shards by coincidence alone, while another might group the popular shards on a single machine, causing a severe load imbalance. In order to explore the variability of Random allocations, we generated ten different allocations per combination of collection and query stream, and measured the variance of the loads. Figure 22 presents the results for the AOL_G Test query set of Gov2 and the



Figure 22: The variance of the load spread (max load_i – min load_i) of ten Random shard allocations, with M = 4, B = 1. The midline of the box indicates the median, the outer edges the 25th and 75th percentile values, and the whiskers the most outlying values. The load spread for Log-based shard allocation is also plotted as a reference point. Note the different horizontal and vertical scales in the two panes.



Figure 23: The variance of the load spread (max load_i – min load_i) of ten Random shard allocations, now with M = 8, B = 8. All other parameters are as for Figure 22. Note the different horizontal scales in the two panes.

 MQT_W Test query set of CW09-A. Also plotted on the graph is the load spread of the Log-based allocation method. While there was at least one Random allocation that produced a more uniform load than the Log-based method, there was also significant variability in the load spread for different Random allocations, and the median values were all well above the corresponding value for the Log-based method. Figure 23 shows that when the number of machines is increased to M = 8, Log-based allocation produces the best load distribution even when compared to ten different Random allocations. That is, the Log-based method reliably picks an allocation with low spread when compared to Random allocation.

76

Balanced assignment of shards and higher utilization of machines can lead to higher query throughput, as shown in Figure 24. The Random policies shown in this figure are the median performing assignment from Figure 22 and Figure 23. At M = 4, the Log-based policies are distinctively better than a Random policy for the Gov2 dataset (especially for MQT). The difference between Random and Log-based policies widens when M = 8 machines are used. The differences between the allocation strategies for CWo9-A dataset are much smaller due to the smaller differences in load variance between the Random and Log-based assignment as can be seen in Table 19, Figure 22, and Figure 23. Correcting greater load imbalances produces a larger impact in throughput.

Finally, Log-based assignment produces better utilization than Random allocation even after one month in most settings, further supporting the stability of the assignment. It is clear that the method of allocating shards to machines has an impact in load distribution of selective search, and hence throughput.

7.4.4 Scalability

The main focus of this work is on low-resource environments (typically two to eight machines). But one advantage of a simulation model is that other configurations can also be explored, and we are also able to examine large-scale environments. Distributed IR systems typically achieve scalability in two ways: shard replication and/or increasing the number of machines available in the cluster so that each is responsible for a smaller volume of data. We explore the scalability of selective search using these two approaches, and consider RQ 10 – *Does selective search scale efficiently when adding more machines and/or shard replication?* Note that the experiments in this section are for CW09-A only, because Gov2 only has 50 shards.

Figure 25 compares selective search and exhaustive search with M = 64 machines in operation, and uses the methods for load balancing described in Section 7.4.2 and Section 7.4.3. With more machines in use, the latency of exhaustive search is decreased because the shard sizes are smaller. Selective search's latency remains the same because the number of topical shards is not determined by the number of machines. There are 512 random shards and 884 topical shards, thus one might expect selective search to have lower latency than exhaustive search, but exhaustive search is faster. Query term posting lists are shorter in the random shards than in the topical shards selected by Taily or Rank-S, which allows an exhaustive shard to be searched more quickly than a topical shard. However, selective search still provides substantially higher throughput than exhaustive search; selective search may take longer to search one shard, which produces higher latency, but it also searches many fewer shards, which produces higher throughput.

Figure 26 further demonstrates the throughput advantage held by selective search. In this experiment, the total time in simulation seconds required to process 10,000 queries is measured as a function of the number of machines available. In all configurations, selective search remains a better option for throughput, requiring less time to process queries. This experiment also shows that selective search scales with the number of machines available. Both exhaustive and selective search are nearly parallel to the red dash-dot line, which represents the ideal case where throughput is doubled with doubled machine capacity.

In Figure 25 and Figure 26, an *index spreading* strategy was assumed, in which the shards are distributed across the larger number of machines, and then the number of machines allocated to resource selection and shard search are adjusted to make the best use of the new hardware. However, when there are more machines than shards, index spreading is no longer feasible and

a different method must be used. One alternative is to increase throughput by using additional space and adopting a *mirrored* architecture. For example, if the number of machines is doubled, two copies of each shard, twice as many resource allocation cores, and twice as many shard search cores can be used. Mirroring requires double the disk space of index spreading, but may have other advantages that make it desirable. Either approach can roughly double throughput.

Figure 27 compares the effect of index spreading and mirroring when M = 64 machines are employed, again plotting median query latency as a function of query arrival rate. In most configurations, mirroring has a small throughput advantage, derived from the fact that as the number of machines increases, it becomes increasingly less likely that an evenly balanced shard assignment will occur. For example, at T = 280, the standard deviation of the machine load in the CWo9-A MQT Test queries is 5.2% for 64 machines, and 3.9% for a 32 machine mirrored configuration using the index spreading strategy with Log-based allocation. While the throughput differences are small, mirroring provides additional benefits, such as fault tolerance in case of failure. In addition, as mentioned previously, mirroring is clearly the best approach when there are more machines than total shards.

The best overall solution in environments with high query traffic, or many machines may be a mix of index spreading and strategic shard mirroring (replication), an option also noted by Moffat et al. [2006]. This is may be an interesting topic for future work.

7.5 SUMMARY

Selective search has been shown to be substantially more efficient than the standard distributed architecture if computational cost is measured by counting postings processed in a one-queryat-a-time environment [Kulkarni and Callan 2010a,b; Kulkarni 2013]. We have extended those findings using a simulator that models a realistic parallel processing environment and a wider range of hardware configurations to be explored; and long query streams extracted from the logs of web search engines. Using the simulator, we answered several research questions on the efficiency of selective search.

Previous investigations also demonstrated that selective search is as effective as exhaustive search, based on experiments with two large datasets [Kulkarni and Callan 2010a,b; Aly et al. 2013; Kulkarni 2013]. Although our work focused primarily on selective search efficiency, we refined the experimental methodology used to measure effectiveness, and achieved stronger baseline results for the CWo9-A dataset. One consequence of these changes was the discovery that in some cases selective search is less effective than exhaustive search, regardless of which resource selection algorithm is used. Effectiveness was not our focus in this work, but our findings are recorded to benefit other researchers that wish to pursue this topic.

Our investigation of RQ 7 makes it clear that selective search is more efficient than conventional distributed architectures for parallel query processing. In a low-resource environment with few machines, the two-step selective search architecture – resource selection followed by shard access – can deliver greater total throughput (number of queries processed per time interval) *as well as* lower latency (faster response time) than a conventional exhaustive architecture. The latter result is one that some readers may find surprising, and is a consequence of the small size of the shards used by selective search, about 500K documents per shard in our experiments. When more machines are available, exhaustive search has lower latency but selective search always has substantially higher throughput.

Other studies have shown that sample-based and term-based resource selection algorithms have different advantages and costs [Aly et al. 2013]. Our experiments investigated RQ 8 and the effects of the resource selection algorithm on load distribution, latency, and throughput. We found that Rank-S was more accurate than Taily, but also much more computationally expensive, usually resulting in higher latency and lower throughput than Taily. If one chooses to use Rank-S to obtain greater effectiveness, the computational costs cause a skew in the workload of machines that must be corrected by replicating resource selection on multiple machines.

Selective search uses topical shards that are likely to differ in their popularity. In our investigation of RQ 9, we discovered that typical random assignments of shards produce large imbalances in machine load, even when as few as M = 4 machines are assigned. A Log-based assignment policy using training queries provided higher throughput and more consistent query processing times than a random assignment policy. The Log-based assignment is also resistant to temporal changes, and even after a delay of a month, throughput degradation was minimal.

Previous studies only investigated selective search in computing environments that contained a small number of machines (*low resource* computing environments). To study RQ 10, with the aid of the simulator we also examined the behavior of selective search using clusters of up to M = 64 machines, each with eight processing cores. We found that selective search remains a viable architecture with high throughput in these larger-scale computing environments. When additional processing resources are available, mirroring (replicating all index shards) provides slightly better throughput in addition to advantages such as fault-tolerance and the ability to use more machines than there are shards when compared to index spreading (no replication). It may be sufficient to replicate a few high-load shards rather than the entire index, i.e. selective replication [Moffat et al. 2006], which would greatly reduce storage costs without sacrificing throughput. We reserve this interesting problem for future work.

After a thorough investigation of the efficiency characteristics of the selective search architecture, we conclude that it is highly efficient in low resource environments typical of academic institutions and small businesses, and that the load imbalances of a naive configuration can be readily addressed. At larger scale, the latency advantages may be lost, but selective search continues to provide substantially higher throughput. For batch-oriented or offline tasks where latency is less of a concern than total computational cost, selective search remains the better choice even at large scales.

This chapter presented a simulator which could be used to thoroughly investigate the *efficiency* of a selective search architecture prior to set up. The next chapter provides a tool for evaluating selective search *accuracy* without a full system.



Figure 24: The effect of shard assignment policies on throughput of selective search, using Taily with M = 4, B = 1 (left column) and M = 8, B = 8 (right column). Note the differing horizontal scales in each pair.



Figure 25: Selective search compared to exhaustive search with M = 64, B = 64 using CW09-A and logbased shard assignment for the AOL_W Test and MQT_W Test query sets (shown as AOL and MQT in the labels for brevity).



Figure 26: The total time required by a selective search system to process 10,000 queries as a function M, the number of machines. Assumes an infinite query arrival rate and B = M in all configurations. The two query streams used are MQT_G Test and AOL_G Test, denoted as MQT and AOL respectively. The red dash-dot line represents the ideal case, where throughput is doubled with doubled machine capacity.



Figure 27: Throughput achieved with Log-based shard assignment and M = 64, B = 64 to process queries against CW09-A. The mirrored configuration is a doubled M = 32, B = 32 system, including disk space. The two query streams are MQT_G Test and AOL_G Test, denoted as MQT and AOL respectively.

8

AUREC¹

The mapping of documents into shards, called a *shard map*, can be generated using many different *partitioning schemes* (e.g. text similarity, source domain, geography, random) [Kulkarni and Callan 2010a; Baeza-Yates et al. 2009b; Cambazoglu et al. 2010; Brefeld et al. 2011] and many resource selection algorithms are available for selective search [Aly et al. 2013; Kulkarni et al. 2012]. The shard map can greatly affect the accuracy of the end-to-end selective search system and is one of multiple such components (Chapter 3). Thus, it is desirable to be able to measure the quality of shard maps independent of the other components in the selective search system. In prior work [Kulkarni and Callan 2010a; Dai et al. 2016], the effectiveness of shard maps was evaluated with end-to-end selective search systems using queries with relevance judgements. This method is affected by the choices made in other system components. Building an end-to-end system and gathering relevance judgements also can be costly. A measurement of shard map quality independent of other system components would provide better diagnostic information and allow a selective search system to be tuned more easily and quickly.

This chapter introduces a new method for estimating shard map effectiveness called Area Under Recall Curve (AUReC). AUReC is calculated independent of other selective search system components. It requires only a set of queries and the results of the full collection retrieval for those queries; from this data, AUReC can then be used to identify shard maps for an effective selective search system. To our knowledge, AUReC is the first to fully decouple shard map evaluation from other selective search components.

We set up end-to-end selective search systems using three different resource selection algorithms. Using shard maps from Dai et al. [2016] and queries with relevance judgements, we compare the scores AUReC gives shard maps to the accuracy scores from the end-to-end selective search systems to answer the following research questions:

RQ 11 Can AUReC generate a relative ordering of multiple shard maps that is consistent with orderings generated by end-to-end system evaluations?

RQ 12 *Is AUReC consistent with end-to-end system evaluations on whether differences between two shard maps are statistically significant?*

RQ 13 Is AUReC robust when compared to end-to-end systems using different resource selection algorithms; compared to a variety of IR metrics at different retrieval depths; when the search engine generating the top-k retrieval is weaker; and when it is calculated with a different set of queries than the end-to-end evaluation queries?

¹ This chapter is a revised and expanded version of Yubin Kim and Jamie Callan. Measuring the effectiveness of selective search index partitions without supervision. In *Proceedings of the 4th ACM SIGIR International Conference on the Theory of Information Retrieval*, pages 91–98, 2018. Figure 28, Figure 30 and the RBO baselines of Table 22, Table 23, Table 24, and Table 25 are new additions of this dissertation.

84

Section 8.1 refocuses the discussion of pertinent past literature under the lens of shard map evaluation and presents the motivation for this work. Section 8.2 details the process for calculating AUReC. Section 8.3 describes the experimental methodology. Section 8.4 presents the experimental results and Section 8.5 concludes.

8.1 RELATED WORK AND MOTIVATION

Recall that a *shard* is a search index that contains a subset of the total document collection. A *shard map* is the mapping that describes to which shard each document in the collection belongs, which is determined by the *partitioning scheme*, a method or technique for generating shard maps, e.g. text-similarity clustering, grouping documents by geography, or splitting a collection randomly. Three areas of prior research created and used shard maps: cluster-based retrieval, distributed retrieval, and selective search.

Recall that *cluster-based retrieval* systems organize a corpus into topical clusters (shards) to improve search accuracy. A common method of assessing partition quality was to measure search results produced by a complete ("end-to-end") system that selects k optimal clusters [Tombros et al. 2002; Griffiths et al. 1986]. This method has the advantage of separating evaluation of the shard map from the problem of identifying the best shard(s) (*resource selection*). However, in Chapter 3 we saw that optimal selection gives misleading information because it measures only how well relevant documents are grouped together but ignores whether they are grouped in a way that allows them to be found. This method also requires manual relevance assessments and is sensitive to parameter choices for the end-to-end system.

Distributed retrieval divides a large collection into shards to improve efficiency. The *efficiency* enabled by different partitioning schemes is well-studied [Moffat et al. 2007; Cambazoglu et al. 2013; Baeza-Yates et al. 2009b; Cambazoglu et al. 2010; Brefeld et al. 2011]. However, the *effectiveness* of different shard maps has received little attention.

Selective search aims to improve the efficiency of large scale search while maintaining the same quality as searching the entire collection (*exhaustive search*). Prior work [Dai et al. 2016; Kulkarni and Callan 2010a; Kulkarni 2013] on selective search evaluated shard maps using the end-to-end retrieval accuracy (e.g., MAP, NDCG@k, or P@k) of a complete system using that shard map. This methodology has the advantage of being realistic. However, as noted above, it requires manual relevance assessments and because the performance of a selective search system is affected by multiple components (Chapter 3), often the end-to-end result is more representative of the effectiveness of the particular choice of components and how well they are tuned than the quality of the shard map. Thus, a thorough system designer would generate many shard maps and test each one against a variety of system configurations, using different resource selection algorithms and different efficiency levels [Dai et al. 2016]. This is cumbersome, so it is not done often. It is also more difficult to do well than one might expect.

Consider a case where shard maps containing a different numbers of shards are compared. Recall that Rank-S is a resource selection algorithm with dynamic shard cut-offs. Given two Gov2 shard maps containing 50 shards and 194 shards, the parameter B = 3 in Rank-S selects 3.7/50 shards and 4.0/194 shards on average for the TREC Terabyte Track queries, leading to very different efficiency and accuracy. The parameters of the resource selection algorithm must be tuned for each shard map such that approximately the same fraction of documents are searched. An end-to-end evaluation of these *two* shard maps requires two sweeps for this single parameter.

There is a need for shard map evaluation that is decoupled from other system components and parameters, and does not require relevance judgements. Such a tool would give better diagnostic information about the quality of the shard maps and enable rapid tuning of new selective search systems.

One method of decoupling evaluation from relevance judgments is to compare the results of a less thorough (less expensive) search to the results of a more thorough (more expensive) search that exhaustively searches the entire index. The more thorough, exhaustive search system is treated as the 'gold standard' that the less thorough system is intended to mimic. Exhaustive search has been used to measure a new system's accuracy [Carmel et al. 2001] and efficiency (Chapter 7). It has also been used as a target to train a supervised resource selection algorithm (Chapter 6). Clarke et al. [2016] used comparisons with exhaustive results to evaluate the effectiveness of early-stage filters in a multi-stage retrieval system. In a selective search context, this is equivalent to evaluating the end-to-end selective search pipeline by overlap with exhaustive results. We are the first to apply this concept to scoring shard maps, independent of the other stages in selective search.

We borrow ideas from French and Powell [2000], who describe a recall-like measure, $\hat{R}_k(E, B)$, used to compare the performance of resource selection algorithms in federated search. We use similar ideas for the purpose of shard map evaluation. Let B represent an ideal ordering for query q of a shard map containing n shards; and E the estimated ordering produced by a resource selection algorithm. Let B_i represent the number of relevant documents in the i'th ranked shard in B, and similarly for E_i and E. It must be the case that $B_i \ge B_{i+1}$; this ordering of shards is called relevance-based ranking (RBR). It is not necessarily the case that $E_i \ge E_{i+1}$.

$$\hat{\mathsf{R}}_{\mathsf{k}}(\mathsf{E},\mathsf{B}) = \frac{\sum_{i=1}^{\mathsf{k}}\mathsf{E}_{i}}{\sum_{i=1}^{\mathsf{n}}\mathsf{B}_{i}}$$

French and Powell plot $\hat{R}_k(E, B)$ from $k = \{1 ... n\}$ to visualize and compare the effectiveness of various resource selection algorithms. Our work develops a variation of the \hat{R}_k curve to evaluate shard maps quantitatively, thereby transforming a visualization aid into a powerful diagnostic tool.

8.2 AUREC

Our goal is a method of evaluating shard maps that that does not require relevance judgements and is independent of the resource selection algorithm, other system components, and parameter settings. While different resource selection algorithms have some differences, their shared goal is to find the shards with the most relevant documents and they tend to return similar shards. Thus, the quality of a shard map can be treated as an inherent property that can be measured on its own.

A few assumptions are made in defining AUReC. We assume that shard maps have shards that are approximately the same size; this is a reasonable assumption because shard maps for selective search are often size balanced for efficiency reasons [Kulkarni 2013].

AUReC also does not take into account the order of which documents appear in the result list. In practical applications, selective search is likely to be used in a multi-stage retrieval system, in which it would act as the fast initial filtering step to produce candidate documents that are passed to more sophisticated re-ranking algorithms to produce the final results [Clarke et al. 2016]. In this scenario, the order of the documents that are passed to the second stage does not matter.

Selective search reduces costs by searching as few shards as possible while trying to produce results that are as accurate as searching all the index shards (*exhaustive search*). For a given query q, a good shard map groups a set of important documents D_q in as few shards as possible, allowing selective search to ignore more shards while searching for D_q .

There are different ways to define D_q . French and Powell [2000] defined it as documents that have been judged relevant by a human assessor. However, we would like to avoid reliance on human relevance judgements. In addition, considering only the relevant documents when evaluating shard maps creates data sparsity issues. Often only a small number of queries with relevance judgements are available and some queries have very few relevant documents. The sparsity of data can produce highly variable results (Chapter 3).

Alternatively, note that the goal of selective search is to meet, rather than to exceed, the accuracy of exhaustive search, just at a much lower cost. Thus, D_q can be defined as the top-k documents that an exhaustive search system returns for the query. The size of k may vary based on the use-case of the selective search system. In the use-case of a first-stage retrieval system (which we assume), selective search must return deep result lists of up to a thousand of documents to support later stage re-rankers [Macdonald et al. 2013; Culpepper et al. 2016]. Therefore, in this chapter we define k = 1000.

For the exhaustive search system, a well-tuned, strong ranker is preferred (e.g. a trained learning to rank system) for more accurate signals on quality of the documents. This method avoids the sparsity issues of relevance judgements; our experiments show that the additional data generated by this method increases the robustness of AUReC.

Given the goals of selective search, a scoring metric f(p, q) which takes a shard map p and a query q should reward a high concentration of D_q . For example, consider two maps with the same number of shards $p_i = \{s_1^i, s_2^i\}$ and $p_j = \{s_1^j, s_2^i\}$, where $|s_1^i \cap D_q| = k$ and $|s_2^i \cap D_q| = 0$, and $|s_1^j \cap D_q| = k - 1$ and $|s_2^j \cap D_q| = 1$, that is, shard map p_i has all of D_q in a single shard, where p_j has the documents spread across multiple shards. In this scenario, the scoring function should produce scores $f(p_i, q) > f(p_j, q)$.

In addition, consider two shard maps with different number of shards, $p_i = \{s_1^i, \ldots, s_{n_i}^i\}$ and $p_j = \{s_1^j, \ldots, s_{n_j}^j\}$, where p_i has fewer shards than p_j , i.e. $n_i < n_j$. Assume $|s_1^i \cap D_q| = k$ and $|s_l^i \cap D_q| = 0$ for $l = \{2, \ldots, n_i\}$, and similarly, $|s_1^j \cap D_q| = k$, $|s_l^j \cap D_q| = 0$ for $l = \{2, \ldots, n_j\}$, that is, both shard maps have all of D_q in one shard. The scoring function should award the shard map that has all of D_q in a smaller shard, or equivalently, the shard map with more shards, since we assume shard sizes are approximately equal within a shard map. Thus, the function should produce scores $f(p_i, q) < f(p_j, q)$.

There are also two properties that are desirable in a shard map evaluation metric as a matter of convenience: the metric should be quick to compute and should allow statistical significance tests. In this section, we introduce AUReC, a shard map scoring method that satisfies all the above properties without relevance judgements or setting up an end-to-end system.

Given a shard map p containing n_p shards, a query q, and D_q , the top 1000 documents that an exhaustive search system returns for query q, we can calculate the maximum possible recall of D_q when searching up to a limit of k shards as follows.

For query q, let $count(D_q, s_i^p)$, $i = \{1...n_p\}$ represent the number of documents in D_q present in shard s_i^p of shard map p, where the shards were ordered such that $count(D_q, s_i^p) \ge count(D_q, s_{i+1}^p)$. $R_q(p, k)$ is the percentage of documents in D_q that appear in the first k shards of shard map p, that is:



Figure 28: Example of a recall versus fraction of shards searched curve for three different index partitions for the same query.

$$R_{q}(p,k) = \begin{cases} 1 & |D_{q}| = 0\\ \frac{\sum_{i=1}^{k} count(D_{q},s_{i}^{p})}{|D_{q}|} & |D_{q}| > 0, k = \{0...n_{p}\} \end{cases}$$

 R_q loosely represents recall (what portion of D_q can be found in the given shards) and k represents efficiency (searching less shards is more efficient). The relationship between the recall and efficiency in selective search can be described in graphical form by plotting k vs. $R_q(p,k)$ for $k = \{0...n_p\}$. It is a convex, monotonically increasing curve, with the x-axis normalized to be between [0,1] by dividing by n_p . When comparing multiple shard maps for a given query, the curves of effective shard maps will be higher and to the left of less effective shard maps. An example is presented in Figure 28. In this example, shard map A is more effective than B, achieving higher R_q at lower values of k. C is a shard map where documents in D_q were scattered evenly across all shards and thus R_q increases linearly with k.

Our method calculates the area under this recall versus fraction of shards searched curve and is named *Area Under Recall Curve* (AUReC). The AUReC for a query q for a shard map p is calculated as follows, using the formula for calculating the area of trapezoids:

AUReC_q(p) =
$$\frac{1}{n_p} \sum_{k=0}^{n_p-1} \frac{R_q(p,k) + R_q(p,k+1)}{2}$$

AUReC scores have the range [0.5, 1.0], where 0.5 indicates that the documents from D_q are completely evenly distributed across the shards and is the worst possible score, shown by shard map C in Figure 28. Scores closer to 1.0 indicate that the documents from D_q are tightly clustered in very few shards. 1.0 is a degenerate case that occurs only when D_q is empty.

 $R_q(p,k)$ is comparable to $\hat{R}_k(B,B)$ [French and Powell 2000], where B is the ideal ordering of shards in shard map p (refer to Section 8.1). It differs by using exhaustive search runs rather than human-judged relevant documents. It also has defined values when $|D_q| = 0$ and k = 0, whereas the equivalent in \hat{R}_k are undefined. The k = 0 case for R_q is critical when comparing shard maps that contain a different number of shards. Without this modification, AUReC plotted from $k = \{1...n_p\}$ (as opposed to $k = \{0...n_p\}$) is biased towards shard maps containing fewer shards. For example, consider a shard map p_{100} consisting of 100 shards with $R_q(p_{100}, 1) = 1$ and a shard map p_2 consisting of 2 shards with $R_q(p_2, 1) = 1$. Without the k = 0 case, AUReC(p_{100}) = AUReC(p_2) = 1, despite that D_q is more densely clustered in p_{100} . By starting at k = 0, AUReC(p_{100}) = 0.01 * 0.5 + 0.99 * 1 = 0.995 > AUReC(p_2) = 0.5 * 0.5 + 0.5 * 1 = 0.75, as desired.

First, efficiency is not taken into account; given two shard maps containing the same number of shards where one map has roughly equal sized shards and another map has a large shard containing 90% of the documents in the collection, AUReC would return a higher score for the unbalanced map, even though the first map is better for efficiency.

AUReC is calculated on a per query basis similar to traditional information retrieval metrics and can be averaged across queries to generate a summary value. Paired significance testing is possible. In this chapter, unless otherwise specified, the AUReC score refers to the AUReC_q averaged across all queries in the query set.

No relevance judgements are used to calculate AUReC and no other component of selective search needs to be implemented. While the size of $|D_q|$ is a parameter that can be tuned based on the usage of the selective search system or its target metric, there is no need to tune a shard cut-off value or to pick an efficiency level for resource selection. Instead, the shard-recall curve describes how well a given shard map performs for a query over all shard cut-offs. Using only the results of an exhaustive search engine, AUReC can be used to quickly tune a new, effective selective search system to replace the less efficient exhaustive search system [Clarke et al. 2016].

8.3 EXPERIMENTAL SETUP

We investigate the effectiveness of AUReC by comparing it to full end-to-end system evaluations of selective search of various configurations; shard maps were evaluated with both methods on two large web datasets and the results are compared. Generating an end-to-end evaluation requires queries with relevance judgements and choosing a resource selection algorithm. AUReC requires the results of a strong retrieval engine. This section describes these necessary components and the metrics used to compare the two methods of evaluation.

8.3.1 Data and query sets

The experiments were conducted on two large web datasets, Gov2 and ClueWebo9 Category B (ClueWebo9 B). The shard maps used in evaluation were generated by Dai et al. [2016] using three different methods: KLD-Rand, QKLD-Rand and QKLD-Qinit, where each successive method produces shard maps of higher quality than the last. These partitioning schemes have random components and different random seed initializations produce different shard maps. Using 10 different random seeds, Dai et al. [2016] generated 10 shard maps for each method for a total of 30 shard maps per dataset, 60 shard maps overall². This dataset contains shard maps that should

² Downloaded from http://boston.lti.cs.cmu.edu/appendices/CIKM2016-Dai/

be of similar qualities and different qualities, and is ideal for testing a new evaluation measure which should be able to distinguish the two cases.

To generate an end-to-end system evaluation, we utilize queries and relevance judgements published by TREC. Gov2 was evaluated with the TREC Terabyte Track queries spanning 2004–2006. ClueWeb09 B was evaluated with the TREC Web Track queries spanning 2009–2012. The relevance judgements for the TREC Web Track queries were filtered such that only documents in Category B were present. Furthermore, queries with zero judged relevant documents were removed from consideration (topics 20, 112, 143, 152 in the Web Track queries), since an end-to-end system evaluation of these queries cannot be used to rank shard maps. AUReC is calculated using the same queries but the relevance judgements are not used.

8.3.2 Document retrieval

Calculating AUReC requires the top 1000 documents that should be retrieved for a given query q, D_q . We create D_q using the top 1000 results of a strong retrieval engine to create the best possible substitute for relevance judgements. To create competitive near state-of-the-art results, SlideFuse-MAP [Anava et al. 2016], a data fusion technique was used to fuse the top 10 runs submitted to TREC in each year as ordered by MAP. Selecting the best runs by MAP creates an indirect dependence on relevance judgements. This is realistic since most state-of-the-art systems are supervised and trained with relevance judgements or user data. We later experiment with a version of AUReC that is completely independent of relevance judgements in Section 8.4.3. The accuracy of the resulting retrieval runs is summarized in Table 21. The scores of SlideFuse-MAP were also used to rank the documents in the end-to-end selective search system. Thus, the evaluation of an end-to-end system uses retrieval results of quality on par with the information used by AUReC, enabling apple-to-apple comparisons.

8.3.3 Resource selection and baselines

To compare AUReC to the current state-of-the-art, we must create several different configurations of end-to-end selective search systems, requiring a few different resource selection algorithms. Three resource selection algorithms were used in the experiments: Taily [Aly et al. 2013], a termbased algorithm; Rank-S [Kulkarni et al. 2012], a sample-based algorithm; and oracle method, relevance-based ranking (RBR) at various static shard cut-offs. As discussed in Section 8.1, RBR orders shards based on the relevance judgements of the query. Shards with the more relevant documents are ranked higher.

The parameters for Rank-S and Taily were set using the values from Chapter 7 and are summarized in Table 12. Note that Chapter 7 uses ClueWebo9 A, but we apply the same parameters to ClueWebo9 B as the latter is a subset of the former. When searching the CSI for Rank-S, SlideFuse-MAP is used to rank the documents. While Rank-S and Taily produce dynamic shard cut-offs (i.e. number of shards to search per query is built-in to the method and varies by query), RBR does not. Therefore, we evaluate RBR selective search systems at three different static shard cut-offs, $k = \{1, 3, 5\}$, which is a common range for shard cut-offs (Chapter 7) and expresses a range of efficiency levels.

The three algorithms used for end-to-end evaluation present two different families of resource selection, two different strategies for shard cut-offs, and the upper bound of resource selection

Query set	Best submitted run (MAP)	Fused run (MAP)						
2009	0.20	0.24						
2010	0.24	0.28						
2011	0.25	0.31						
2012	0.30	0.34						
Average	0.25	0.29						
	(a) ClueWebo9B							
Query set	Best submitted run (MAP)	Fused run (MAP)						
2004	0.31	0.35						
2005	0.40	0.44						
2006	0.42	0.45						
Average	0.38	0.41						
	(b) Gov2							

Table 21: Effectiveness of the top submitted TREC runs compared to the runs created from fusing the top 10 submitted runs for each query set with SlideFuse-MAP [Anava et al. 2016].

performance. By experimenting with three dissimilar resource selection algorithms, we demonstrate the robustness of AUReC.

As AUReC is the first method to completely decouple shard map evaluation from other selective search components, there are no apple-to-apple baselines to which it can be directly compared. However, a common heuristic used score shard maps in cluster-based retrieval systems is optimal cluster effectiveness, the effectiveness of performing retrieval on the one shard that contains the most relevant documents [Tombros et al. 2002]. This is exactly equivalent to an endto-end system evaluation using RBR resource selection with k = 1.

We also created a second shard map scoring baseline by using rank-biased overlap (RBO) [Webber et al. 2010] score of an end-to-end selective search system. RBO is a method used to evaluate the results of a lossy optimization method by how much it overlaps an exhaustive search result. However, the RBO-based baseline still requires an end-to-end selective search system to calculate the score, including resource selection and a shard cut-off.

In our experiments, RBO is parameterized by p = 0.999, which is roughly equivalent to evaluating the result list to a rank depth of 1000. For the resource selection and shard cut-off of the end-to-end selective search system, we use the RBR k = 3 resource selection method. RBR was chosen because it is most accurate resource selection method (as it uses oracle knowledge of relevance judgements) and k = 3 was chosen because common parameterizations of selective search resource selection algorithms select around 3–5 shards (see Table 12).

Although these scoring methods have dependencies on relevance judgements and shard cutoffs, we include them as baselines in our experiments to provide context and a point of comparison. Note that variations of these baselines that do not use relevance judgements can be easily created as noted in Section 8.1. However, these baselines are weaker because they use less infor-

AUReC EtE	p _i > p _j	No sig. diff.	p _i < p _j
$p_i > p_j$	а	b	С
No sig diff	dl		fl
NO SIG. UIII.	d2	e	f2
p _i < p _j	g	h	i

Figure 29: Contingency table for pair-wise comparison of index partitions using AUReC and end-to-end system evaluation (EtE).

mation. Thus, we chose to present the strictly stronger baselines of optimal cluster effectiveness and rank-biased overlap.

8.3.4 Metrics

To establish AUReC as a robust method, we compare it to end-to-end selective search evaluations of multiple different settings. An end-to-end evaluation score EtE(p) of shard map p is generated by evaluating queries on a selective search system based on shards dictated by shard map p. Multiple versions of EtE(p) are possible dependent on the type of resource selection algorithm used, and the evaluation metric used to score the final result list of selective search.

The primary evaluation metric used in this work is Precision at 1000 (P@1000). A metric that uses a deep-rank was chosen to support later stage re-rankers [Macdonald et al. 2013; Culpepper et al. 2016]. In addition, because the results are being re-ranked in later stages, the order in which the relevant documents are returned does not matter. Therefore, we use P@1000, which is a deep, rank-insensitive metric. The metric also parallels our decision to use 1000 for $|D_q|$. Shallower and rank-sensitive metrics are explored in Section 8.4.2.

AUReC and the end-to-end system evaluation are compared in two major ways: list-wise and pair-wise. In the list-wise comparison, the correlation of the scores of $EtE(p_i)$ and $AUReC(p_i)$ are calculated using Pearson's r. Pearson's r was chosen over non-parametric, ordinal methods such as Kendall's τ so that the correlation in the difference in the magnitude of the scores was measured, not just the relative rankings. This is important as the shard maps used in the experiments were generated with three different methods, with each method generating ten shard maps each. Thus, shard maps generated by the same method are expected to have similar scores, whereas shard maps generated by different methods should have larger score differences.

In the pair-wise comparison, for each pair of shard maps (p_i, p_j) where $i, j = \{1...30\}, i \neq j$, we determine the ordering of AUReC (p_i) and AUReC (p_j) and whether the difference is statistically significant under a paired two-tailed t-test with a significance level of $\alpha = 0.05$. This is repeated for EtE (p_i) and EtE (p_j) . Given that there are 30 shard maps for a dataset, the total number of pair-wise comparisons is $\binom{30}{2} = 435$. Based the decisions of AUReC and end-to-end evaluation on the direction and significance of the differences between p_i and p_j , a contingency table can be built as shown in Figure 29. Note that cells d and f are split in half. This is to indicate two separate scenarios; in d1 and f2, the end-to-end evaluation scores and AUReC scores agreed in the direction of the difference, but only AUReC found the difference to be significant. In d2 and f1, the two methods disagreed on the direction of differences.

Three summary metrics are reported from this table. First is *Pairs recall* = $\frac{a+i}{a+b+c+g+h+i}$. This indicates the fraction of statistically significant differences found by the end-to-end evaluation that was also recovered by AUReC. Second is *Overlap* = $\frac{a+e+i}{435}$ the number of shard map pairs where the end-to-end system and AUReC agreed exactly on the direction and significance of the differences. Lastly, *Additional pairs* = $\frac{d1+f2}{435}$ is reported. As described above, this indicates situations where AUReC was able to detect significant differences where the end-to-end evaluation could not, due to the greater amount of information used by AUReC and the sparsity of relevance judgements used by the end-to-end evaluation. Additionally, *Overlap*+*Additional pairs* is reported together, loosely representing the total amount of agreement between AUReC and end-to-end evaluation. Other cells indicate varying levels of disagreement between the two methods. Cells *c*, *g* indicates strong disagreements where the two methods disagreed on the direction of significance; *b*, *h* indicate statistically significant differences that were missed by AUReC; and d2, f1 are situations of slight disagreement where AUReC specifies statistically significant differences, but in a different direction from the end-to-end evaluation.

8.4 EXPERIMENTAL RESULTS

AUReC is compared against a full end-to-end system evaluation to investigate the correlation between the two metrics and see whether it can substitute for a full end-to-end evaluation. We then investigate the robustness of AUReC by exploring its behaviour when compared against different retrieval metrics and depths; when calculated with D_q , the set of documents that should be retrieved for query q, that was generated by a weaker retrieval engine; and when calculated with queries from a different query set.

8.4.1 *Comparison with end-to-end systems*

AUReC, optimal cluster effectiveness (equivalent to end-to-end evaluation with RBR, k = 1), and rank-biased overlap (RBO score of an end-to-end evaluation with RBR, k = 3) baselines are compared against end-to-end selective search systems utilizing various resource selection techniques. The results are presented in Table 22 for CWo9-B and Table 23 for Gov2.

The experiments demonstrate that AUReC is highly correlated with all end-to-end system evaluations in both list-wise and pair-wise comparisons. The high Pearson's r indicates that the AU-ReC and end-to-end evaluation scores order shard maps similarly throughout the entire score range. Figure 30 visualizes the correlation. In addition, the best shard map selected by AUReC and the end-to-end systems were either identical or were shard maps where there were no statistically significant differences, indicating agreement in the highest end of the scores (not shown). In other words, AUReC and end-to-end system evaluations will produce very similar decisions when picking the best shard map.

AUReC also had a high *Pairs recall*, indicating that it recovered most of the statistically significant differences between shard map pairs found by the end-to-end systems. Finally, the high *Overlap+Additional pairs* sum shows that AUReC and end-to-end evaluations mostly agree on the direction of differences between pairs of shard maps. In addition, while not shown in the table, there were no pairs in which AUReC and end-to-end evaluations disagreed in the direction of statistical significance in any of the settings; that is, there were no pairs of shard maps (p_i, p_j) where
Table 22:	Comparison of AUReC and two baseline metrics against the evaluation of end-to-end (EtE) sys-
	tems with various resource selection algorithms in the CWo9-B dataset. The end-to-end systems
	used P@1000 to measure the effectiveness of the shard maps. r is Pearson's correlation. Optimal
	cluster effectiveness is a commonly used heuristic in prior work that is identical to an end-to-end
	system evaluation using RBR $k = 1$. The rank-biased overlap baseline uses the the rank-biased
	overlap ($p = 0.999$) score of an RBR $k = 3$ selective search system against exhaustive search.

Resource selection	r	Pairs recall	Overlap+Additional pairs		
	Opti	imal cluster effectiv	eness		
Rank-S	0.94	176/236 = 0.75	0.84 + 0.03 = 0.86		
Taily	0.90	161/184 = 0.88	0.89 + 0.06 = 0.94		
RBR ($k = 1$)	-	-	-		
RBR ($k = 3$)	0.95	181/228 = 0.79	0.88 + 0.01 = 0.89		
RBR ($k = 5$)	0.91	173/215 = 0.80	0.87 + 0.03 = 0.90		
Rank-biased overlap					
Rank-S	0.94	216/236 = 0.92	0.75 + 0.17 = 0.93		
Taily	0.91	180/184 = 0.98	0.71 + 0.20 = 0.91		
RBR ($k = 1$)	0.93	184/187 = 0.98	0.72 + 0.22 = 0.94		
RBR ($k = 3$)	0.97	219/228 = 0.96	0.78 + 0.19 = 0.98		
RBR ($k = 5$)	0.94	204/215 = 0.95	0.74 + 0.20 = 0.95		
AUReC					
Rank-S	0.96	221/236 = 0.94	0.71 + 0.21 = 0.93		
Taily	0.92	181/184 = 0.98	0.65 + 0.23 = 0.88		
RBR ($k = 1$)	0.94	185/187 = 0.99	0.66 + 0.27 = 0.93		
RBR ($k = 3$)	0.98	219/228 = 0.96	0.72 + 0.23 = 0.96		
RBR ($k = 5$)	0.96	206/215 = 0.96	0.69 + 0.25 = 0.94		

AUReC believed that p_i was statistically significantly better than p_j but an end-to-end evaluation believed vice versa.

AUReC correlates better with end-to-end systems than optimal cluster effectiveness, across both datasets and nearly all settings and comparisons, including systems that use RBR with different cut-offs. In particular, AUReC is superior in replicating statistically significant differences between pairs of shard maps (*Pairs recall*). This is because optimal cluster effectiveness produces high query variance. Other resource selection algorithms select 3–5 shards on average. Selecting only one shard means optimal cluster effectiveness sees fewer relevant documents and has less information to make judgements. Consequently, compared to AUReC, optimal cluster effectiveness produces higher variance and is less able to distinguish statistically significant differences and is less correlated with other systems overall. While often used in prior research [Tombros et al. 2002], it is a less reliable metric.

There are two places where optimal cluster effectiveness correlates better than AUReC. First is the pair-wise comparisons against a Taily system in ClueWebo9 B, where optimal cluster ef-

Table 23:	Comparison of AUReC against the evaluation of end-to-end (EtE) systems with various resource
	selection algorithms in the Gov2 dataset. The end-to-end systems used P@1000 to measure the
	effectiveness of the shard maps. r is Pearson's correlation. Optimal cluster effectiveness is a com-
	monly used heuristic in prior work that is identical to an end-to-end system evaluation using
	RBR $k = 1$. The rank-biased overlap baseline uses the the rank-biased overlap (p = 0.999) score
	of an RBR $k = 3$ selective search system against exhaustive search.

Resource selection	r	Pairs recall	Overlap+Additional pairs		
Optimal cluster effectiveness					
Rank-S	0.95	169/226 = 0.75	0.84 + 0.03 = 0.87		
Taily	0.86	115/149 = 0.77	0.77 + 0.15 = 0.92		
RBR ($k = 1$)	-	-	-		
RBR ($k = 3$)	0.93	169/224 = 0.75	0.85 + 0.02 = 0.87		
RBR ($k = 5$)	0.89	166/229 = 0.72	0.82 + 0.03 = 0.85		
Rank-biased overlap					
Rank-S	0.95	205/226 = 0.91	0.94 + 0.01 = 0.95		
Taily	0.82	124/149 = 0.83	0.74 + 0.19 = 0.93		
RBR ($k = 1$)	0.91	166/181 = 0.92	0.86 + 0.11 = 0.97		
RBR ($k = 3$)	0.93	201/224 = 0.90	0.92 + 0.02 = 0.94		
RBR ($k = 5$)	0.94	202/229 = 0.88	0.91 + 0.02 = 0.93		
AUReC					
Rank-S	0.95	197/226 = 0.87	0.89 + 0.05 = 0.93		
Taily	0.84	121/149 = 0.81	0.71 + 0.20 = 0.92		
RBR ($k = 1$)	0.93	166/181 = 0.92	0.85 + 0.12 = 0.96		
RBR ($k = 3$)	0.93	193/224 = 0.86	0.87 + 0.05 = 0.92		
RBR ($k = 5$)	0.92	195/229 = 0.85	0.87 + 0.03 = 0.90		

fectiveness has a higher overall agreement (*Overlap+Additional pairs*). This was because Taily also produces high variance results as seen in Chapter 3 and most of the *Overlap* was from pairs where both methods did not find significant differences (58% of *Overlap* pairs). In contrast, most of the *Overlap* between AUReC and Taily system were from concordances on statistically significant differences (64% of *Overlap* pairs), which is more informative.

Optimal cluster effectiveness also produced a slightly higher Pearson's r than AUReC when compared to the Taily system in Gov2. However, this difference is less meaningful; due to the high variance of the Taily systems, most of the pair-wise differences between shard maps in the end-to-end evaluation are not significant. This means that in an ordered list of shard maps, there may be sections where multiple shard maps are interchangeable in order because the confidence intervals of the shard maps' scores overlap each other. This introduces noise when calculating Pearson's r against a shard map ordering created by a Taily system and thus small differences become less meaningful.



Figure 30: Correlation between AUReC scores and P@1000 scores of an end-to-end selective search system using Rank-S resource selection. Different shapes indicate that the shard map was created using the KLD-Rand (★), QKLD-Rand (×), or QKLD-Qinit (+) method as described in Dai et al. [2016].

The RBO-based baseline, which uses k = 3 shards, is more reliable than optimal cluster effectiveness and makes better use of relevance data. AUReC performs similarly to this stronger baseline, while not requiring an end-to-end implementation of selective search. The trends of the two methods are largely similar. However, while the sum in the *Overlap+Additional pairs* column of the two methods are similar, the AUReC consistently discovers more *Additional pairs* and has less *Overlap*, especially in the ClueWebo9 B dataset. This is a similar effect as discussed above with optimal cluster effectiveness; across all end-to-end selective search systems tested, a larger portion of the *Overlap* between AUReC and end-to-end systems were from agreement on statistically significant differences than that of RBO. To continue the example from above, for the Taily end-to-end system in the ClueWebo9 B dataset, only 59% of RBO *Overlap* pairs were concordances on statistically significant differences compared to the 64% of AUReC. This indicates that AUReC is more informative in its decisions than the baselines.

Overall, when comparing Table 22 and Table 23, AUReC has slightly stronger correlations with end-to-end retrieval systems than RBO in ClueWebo9 B and slightly weaker correlations than RBO in Gov2. This is against expectations; one would expect that RBO, a baseline in which relevance data is used to rank shards, would consistently perform better than AUReC, which only uses relevance data for picking the best TREC systems to construct D_q . This due to the relative lack of relevance data for ClueWebo9 B. The TREC Web Track queries have shallow judgement pools and produce high variance due to the dearth of signals [Lu et al. 2016]. This causes variance in the shards that are selected based on said relevance data, as well (Chapter 3). In this situation, AUReC is more robust to the lack of data and is able to estimate the performance of shard maps more accurately than RBO. In Gov2, where relevance data is more plentiful, RBO performs better than AUReC.

In the work that described the shard maps used in our experiments, Dai et al. [2016] discovered that while the differences in the means for the end-to-end evaluation scores of the three categories of partitioning schemes (KLD-Rand, QKLD-Rand, and QKLD-QInit) were greater in ClueWebo9 B than Gov2, the differences were sometimes not statistically significant in ClueWebo9 B due to

the higher variance. It was unclear which data set benefits more from the improved partitioning scheme. This can now be answered using AUReC.

AUReC is calculated based on 1000 retrieved documents and uses more information than evaluating an end-to-end retrieval with sparse relevance judgements. Thus, AUReC finds significant differences where an end-to-end system evaluation cannot. Note the amount of Additional pairs found by AUReC on the ClueWebo9 B dataset. These are shard map pairs where the end-to-end evaluation saw the same direction of difference as AUReC but could not conclude their statistical significance. The end-to-end evaluation found as few as 42% of shard map pairs to be significantly different (the denominator of the Pairs recall column), but AUReC found significant differences in 76% of shard map pairs in ClueWebo9 B (cells a + d + g + c + f + i in Figure 29). A visual representation can be seen in Figure 30b. The three groups of shard maps are clearly distinguished when projected along the x-axis (AUReC) while the groups overlap when projected along the yaxis (P@1000). In contrast, Terabyte Track queries for Gov2 have deeper pooling for the relevance judgements, producing stable results. AUReC finds fewer Additional pairs for Gov2 and found significant differences in 50% of shard map pairs in Gov2, similar to end-to-end evaluations and lower than ClueWebo9 B. Figure 30a illustrates this. The differences between the groups of shard maps are similar for both AUReC and P@1000, and the QKLD-Qinit and QKLD-Rand shard maps are less separated in Gov2 compared to ClueWebo9 B. Thus, thanks to evidence supplied by AUReC, we can now conclude that the choice of partitioning scheme affects ClueWebo9 B more strongly than Gov2.

AUReC typically generates fewer *Additional pairs* in Gov2. However, when the resource selection algorithm used by the end-to-end system has high variance (i.e. Taily and RBR k = 1) AUReC finds significant differences that the end-to-end systems miss.

With this experiment, RQ 11, RQ 12 and a portion of RQ 13 were answered. The ordering of shard maps generated by AUReC is highly correlated with the ordering determined by end-toend system evaluations (RQ 11). In pair-wise comparisons, AUReC found significant differences in pairs of shard maps where the end-to-end system found differences. In addition, AUReC was able to discover additional pairs of significantly different shard maps, where end-to-end systems were unable to ascertain the significance (RQ 12). Finally, AUReC performed well across two data sets and three different resource selection algorithms, and substantially better than the common baseline heuristic used by prior cluster-based retrieval research and similarly to the stronger baseline of RBO-based scoring, which use relevance data and an end-to-end selective search system (RQ 13). Further sections elaborate on RQ 13 and examine AUReC under less ideal scenarios to test its robustness.

8.4.2 Different metrics and evaluation depths

So far P@1000 was used for the end-to-end system evaluation, which is a metric directly comparable to AUReC. AUReC is calculated with a top 1000 retrieval ordering of shards that maximizes the recall. Because there are 1000 documents in D_q , in most cases this is equivalent to maximizing P@1000 (where a document is considered 'relevant' if it is in D_q). However, not all metrics are so directly related to AUReC. If a metric takes rank position into account, a recall-oriented ordering of shards is not always optimal. In fact, for metrics such as NDCG and MAP, the optimal solution set of k shards is not always a subset of the optimal set for k + 1 shards. To explore how well AU-ReC correlates with metrics that are less directly related, AUReC is compared against end-to-end systems evaluations using two rank sensitive metrics, MAP and NDCG at two different evalu-

Metric	r	Pairs recall	Overlap+Additional pairs			
	С	Optimal cluster effe	ctiveness			
P@1000	0.94	176/236 = 0.75	0.84 + 0.03 = 0.86			
NDCG@1000	0.89	97/248 = 0.39	0.64 + 0.01 = 0.65			
MAP@1000	0.78	17/217 = 0.08	0.53 + 0.01 = 0.54			
P@100	0.86	116/204 = 0.57	0.76 + 0.03 = 0.79			
NDCG@100	0.75	10/214 = 0.05	0.53 + 0.00 = 0.53			
MAP@100	0.62	8/190 = 0.04	0.57 + 0.01 = 0.58			
Rank-biased overlap						
P@1000	0.94	216/236 = 0.92	0.75 + 0.17 = 0.93			
NDCG@1000	0.91	219/248 = 0.88	0.74 + 0.16 = 0.90			
MAP@1000	0.89	195/217 = 0.90	0.70 + 0.19 = 0.89			
P@100	0.86	189/204 = 0.93	0.70 + 0.17 = 0.88			
NDCG@100	0.86	187/214 = 0.87	0.67 + 0.20 = 0.87			
MAP@100	0.86	166/190 = 0.87	0.63 + 0.24 = 0.87			
AUReC						
P@1000	0.96	221/236 = 0.94	0.71 + 0.21 = 0.93			
NDCG@1000	0.94	229/248 = 0.92	0.72 + 0.19 = 0.91			
MAP@1000	0.91	204/217 = 0.94	0.68 + 0.22 = 0.90			
P@100	0.90	194/204 = 0.95	0.66 + 0.21 = 0.88			
NDCG@100	0.90	196/214 = 0.92	0.65 + 0.23 = 0.88			
MAP@100	0.88	175/190 = 0.92	0.61 + 0.27 = 0.88			

Table 24: Comparison of AUReC and two baseline metrics against a Rank-S end-to-end system at different rank depths in the CW09-B dataset.

ation depths, 1000 and 100. The results are presented in Table 24 for CW09-B and Table 25 for Gov2. The table shows results for Rank-S end-to-end systems for brevity. Other resource selection choices behaved similarly except RBR k = 1.

Optimal cluster effectiveness (i.e. RBR k = 1) was much less robust than AUReC and experienced a sharp drop in correlation in every setting and method of comparison. Both the change of metric and the change of evaluation depth adversely affected the correlation and a combination of both produced dismal results, especially in ClueWebo9 B (*Pairs recall* was 0.04 for MAP@100).

RBO is more robust and remains highly correlated with MAP and NDCG, and metrics evaluated at a shallower depth. The trends of RBO and AUReC remain very similar; AUReC also remains highly correlated with the different conditions tested. In particular, differences in the type of evaluation metric did not impact the correlations in a major way. In Gov2, AUReC was slightly *better* correlated with NDCG, a rank sensitive metric, than Precision.

Metric	r	Pairs recall	Overlap+Additional pairs			
	Optimal cluster effectiveness					
P@1000	0.95	169/226 = 0.75	0.84 + 0.03 = 0.87			
NDCG@1000	0.94	153/221 = 0.69	0.78 + 0.06 = 0.84			
MAP@1000	0.94	126/228 = 0.55	0.73 + 0.03 = 0.77			
P@100	0.84	68/173 = 0.39	0.73 + 0.03 = 0.76			
NDCG@100	0.84	50/131 = 0.38	0.77 + 0.05 = 0.81			
MAP@100	0.80	36/115 = 0.31	0.79 + 0.03 = 0.82			
Rank-biased overlap						
P@1000	0.95	205/226 = 0.91	0.94 + 0.01 = 0.95			
NDCG@1000	0.97	201/221 = 0.91	0.93 + 0.03 = 0.95			
MAP@1000	0.95	201/228 = 0.88	0.91 + 0.03 = 0.94			
P@100	0.91	145/173 = 0.84	0.78 + 0.15 = 0.94			
NDCG@100	0.89	114/131 = 0.87	0.74 + 0.22 = 0.96			
MAP@100	0.86	94/115 = 0.82	0.68 + 0.27 = 0.95			
AUReC						
P@1000	0.95	197/226 = 0.87	0.89 + 0.05 = 0.93			
NDCG@1000	0.96	196/221 = 0.89	0.89 + 0.05 = 0.94			
MAP@1000	0.96	197/228 = 0.86	0.88 + 0.05 = 0.93			
P@100	0.91	145/173 = 0.84	0.77 + 0.16 = 0.93			
NDCG@100	0.88	112/131 = 0.85	0.71 + 0.23 = 0.95			
MAP@100	0.86	95/115 = 0.83	0.67 + 0.27 = 0.95			

Table 25: Comparison of AUReC and two baseline metrics against a Rank-S end-to-end system at different rank depths in the Gov2 dataset.

Evaluation depth had a stronger effect on the correlation of AUReC, yet still not necessarily negative. The results show that at shallower evaluation depths, the variance of the end-to-end systems increase, as seen by the lower number of statistically significant shard map pairs found by the end-to-end system, e.g. a P@1000 Gov2 system found 226 pairs with significant differences whereas the P@100 system found only 173. When compared against the shallower, more variable system evaluations, AUReC found more *Additional pairs* and Pearson's r was reduced. This is in line with the observations from Table 22 and Table 23 where similar effects were present when AUReC was compared to a Taily-based end-to-end system which has higher variance. In addition, AUReC continues to find more *Additional pairs* than RBO as seen in Section 8.4.1. While AUReC remains robust against changes in evaluation depth, an interesting avenue for future work would be to extend AUReC to tune it for shallower metrics.

The increased variance was more pronounced in Gov2 than ClueWebo9 B. The queries for Gov2 have more relevant documents per query than those of ClueWebo9 B (182 vs. 50) and the pooling

during the assessing phase was deeper. Gov2 queries have more judged documents further down in the ranked list than ClueWeb09, and thus were impacted more by the loss of this information with the use of a shallower evaluation depth.

Results for shallower depths were not reported because the increase in variance makes the results less interesting. For example, in ClueWebo9 B with P@10, more than 75% of shard map pairs have no significant differences in the Rank-S system. With such noise, list-wise comparison using Pearson's r becomes less informative and any cluster effectiveness metric that is high in variance would have a good pair-wise overlap with the end-to-end evaluation.

This experiment demonstrates that AUReC is well-correlated with different metrics and different evaluation depths. AUReC is a robust, low-variance method that can identify significant differences better than end-to-end system evaluations that have high variance configurations, e.g. using a shallow evaluation depth.

Table 26: Comparison of AUReC scores and Rank-S end-to-end system evaluation using P@1000, when D_q was generated from weaker rankers.

Туре	MAP	r	Pairs recall	Overlap+Additional pairs		
SDM	0.21	0.95	213/236 = 0.90	0.82 + 0.12 = 0.93		
BOW	0.20	0.95	214/236 = 0.91	0.83 + 0.11 = 0.94		
(a) CW09-B, compare with Table 22						
Туре	MAP	r	Pairs recall	Overlap+Additional pairs		
SDM	0.32	0.95	204/226 = 0.90	0.89 + 0.06 = 0.94		
BOW	0.29	0.92	185/226 = 0.82	0.86 + 0.04 = 0.90		

(b) Gov2, compare with Table 23

8.4.3 Weaker retrieval engine

In previous sections, AUReC was calculated based on a very high-quality ranker chosen using relevance judgements. However, results from a strong retrieval engine may not be easy to obtain. In this experiment, we investigate the results of using a good, but less accurate retrieval engine to generate D_{g} .

We use the Indri³ search engine to generate D_q using default search parameters. The indexes for both Gov2 and ClueWebo9 B were stopped and stemmed using the Indri stopword list⁴ and the Krovetz stemmer. Runs of two different types were generated: bag-of-word queries and sequential dependency model (SDM) queries with 0.8 weight given to the original query and 0.1 to the bigrams and 0.1 to the unordered windows. In a post-retrieval step, ClueWebo9 B results were filtered for spam, removing any documents that had a Waterloo Fusion spam score⁵ of below 50. AUReC scores were generated using these weaker runs and were compared to a full end-to-end

³ https://www.lemurproject.org/indri/

⁴ http://www.lemurproject.org/stopwords/stoplist.dft

⁵ http://plg.uwaterloo.ca/gvcormac/cluewebo9spam/

selective search evaluation, unchanged from Table 22 and Table 23. The accuracy of the Indribased runs and correlation of the resulting AUReC scores are presented in Table 26. For brevity, only Rank-S end-to-end system results are shown. Other resource selection methods displayed similar trends.

Despite the fact that the Indri-based runs were significantly worse (approximately -30% change in MAP scores) than the data fusion run, the resulting AUReC scores were well-correlated with the end-to-end evaluation. However, there is some degradation of results. The bag-of-words Gov2 run especially saw a reduction in correlation scores across all methods of comparison.

The ClueWebo9 B results were also degraded, but in a more subtle way. While the final correlation scores were mostly similar, the composition of the overall agreement scores in the Indri runs shifted; the weaker Indri based runs were less able to uncover additional significant differences and instead agreed more with the end-to-end system that the differences were not significant. For example, AUReC scores based on the data fusion run and Indri SDM run both agreed with the end-to-end system evaluation on 93% of all shard map pairs. However, 21% of pairs were *Additional pairs* in the data fusion run whereas only 12% of pairs were new significance discoveries in the Indri SDM run.

Although AUReC using the Indri-based runs produced slightly worse correlation with the endto-end evaluations, the reduction of effectiveness was much less than the difference in accuracy of the runs. This implies that while strong retrieval results can improve AUReC, the robustness of metric indicates that a good out-of-the-box retrieval engine is sufficient to calculate reliable scores.

8.4.4 Different queries

The final experiment further tests the robustness of AUReC by experimenting with queries that lack relevance judgements and high accuracy runs, using an out-of-the-box search engine. The experiments use 10,000 queries from the 2009 TREC Million Query Track (MQT). Queries which were duplicates of the Terabyte Track query set or Web Track query set were removed. The queries were then expanded into SDM queries and 1000 results were retrieved per query from the Indri index using settings and post-retrieval spam filtering for ClueWebo9 B as described in Section 8.4.3.

In order to explore the effect of the number of queries on the reliability of AUReC, AUReC scores were generated using various sized subsets of the MQT queries. These runs were then compared against end-to-end systems that are, as usual, evaluated with queries that have relevance judgements. The results are presented in Table 27. For brevity, only the Rank-S end-to-end system results are shown. Other resource selection methods displayed similar trends.

AUReC generalizes well to different queries and the scores calculated are well-correlated with a full system evaluation. When the best result of Table 27 is compared to the Rank-S entry of Table 22, in ClueWebo9 B, AUReC computed with MQT queries converges to a slightly lower Pearson's r than what could be achieved if the TREC queries are used. In Gov2, the convergence value is slightly higher than in Table 23. Both values remain high and suggest that AUReC shard map rankings remain robust when queries change.

As the number of queries increase, there is a general increase in correlation between AUReC and the end-to-end system evaluation. However, there is a clear diminishing returns effect on list-wise comparisons as more queries are used to calculate AUReC scores. Pearson's r hits a saturation point quickly and moves slowly or not at all with increasing queries. The pair-wise correlation comparisons continue to grow somewhat and *Additional pairs* in particular continues to

grow more with additional queries. This is expected. As additional queries are utilized, the mean of the AUReC scores for a given shard map converge quickly. Thus, the shard map's position in an ordered list shifts rarely. However, with more evidence, the confidence interval around the mean continues to shrink and AUReC uncovers more significant differences. Note that *Overlap* decreases because *e* in Figure 29 decreases (i.e. cases where both AUReC and end-to-end believe the differences between the two shard maps are not significant).

We have calculated AUReC with queries without relevance judgements or high accuracy runs, using an out-of-the-box search engine. The results were highly correlated with full end-to-end system evaluations. When this less optimal configuration is compared to the optimal cluster effectiveness baseline from Section 8.4.1 which used the same queries and relevance judgements as the end-to-end system evaluations, AUReC had better pair-wise correlation and better or near-equal list-wise correlation while using a different set of queries and no relevance judgements, demonstrating that it is robust and reliable.

Table 27: Comparison of AUReC and Rank-S end-to-end system evaluation using P@1000, when using varying number of MQT queries. The end-to-end system was evaluated with TREC queries, as usual.

Num of queries	r	Pairs recall	Overlap+Additional pairs
100	0.89	186/236 = 0.79	0.79 + 0.08 = 0.86
200	0.91	198/236 = 0.84	0.80 + 0.09 = 0.89
400	0.92	202/236 = 0.86	0.77 + 0.12 = 0.89
600	0.92	210/236 = 0.89	0.71 + 0.18 = 0.89
800	0.93	217/236 = 0.92	0.70 + 0.21 = 0.91
1000	0.93	219/236 = 0.93	0.66 + 0.24 = 0.90
2000	0.92	223/236 = 0.94	0.62 + 0.26 = 0.88
10000	0.93	224/236 = 0.95	0.58 + 0.28 = 0.86
		(a) CW09-B	
Num of queries	r	Pairs recall	Overlap+Additional pairs
100	0.89	178/226 = 0.79	0.72 + 0.12 = 0.84
200	0.93	211/226 = 0.93	0.78 + 0.14 = 0.92
400	0.93	211/226 = 0.93	0.74 + 0.16 = 0.90
600	0.94	217/226 = 0.96	0.73 + 0.18 = 0.90
800	0.95	218/226 = 0.96	0.71 + 0.20 = 0.91
1000	0.94	218/226 = 0.96	0.69 + 0.21 = 0.90
2000	0.96	224/226 = 0.99	0.65 + 0.26 = 0.90
10000	0.96	225/226 = 1.00	0.58 + 0.28 = 0.86

8.5 SUMMARY AND RECOMMENDATIONS

Prior work evaluated shard maps by measuring the accuracy of end-to-end selective search systems. This is a cumbersome method that relies on relevance judgements and is sensitive to the specific system configuration. This chapter introduces AUReC, a new way to measure the effectiveness of shard maps that does not require gathering relevance judgments and is the first to completely decouple shard map evaluation from other components and parameters of a selective search system. By freeing shard map quality from other system components, AUReC provides robust diagnostic information that can be used to quickly sort through a large number of shard maps to tune a new selective search system, a process which was previously time-consuming and difficult.

AUReC evaluates shard maps by the area under a recall curve using the retrieval results of an exhaustive search system. It is highly-correlated to end-to-end selective search system evaluations while being simple to implement and not requiring: the implementation of other selective search components; picking a fixed efficiency level; or human-assessed relevance judgements. An examination of the effectiveness and robustness of AUReC found it produces scores that are highly-correlated with the evaluation of end-to-end systems under a variety of configurations. When compared to a rank biased overlap based baseline that uses an end-to-end selective search system with an oracle resource selection method that uses relevance judgements, AUReC performed equally well and had minimal degradation in results when all reliance on relevance judgements were removed.

Given a set of shard maps, the ordering of the shard maps determined by AUReC scores closely resembled the ordering by different end-to-end evaluations, usually with Pearson's r > 0.9, positively answering RQ 11. To investigate RQ 12, pairs of shard maps were compared and it was found that most shard maps that had significant differences under an end-to-end evaluation also were significantly different when compared with AUReC scores. AUReC scores are calculated from easy-to-generate, plentiful data points and therefore produce stable results. Thus, AUReC was able to ascertain significant differences in pairs of shard maps where end-to-end system evaluations could not due to the scarcity of relevance data.

The robustness of AUReC was tested in several ways to answer RQ 13. First, the end-to-end system evaluation methodology was altered to explore how well AUReC generalized. End-to-end system evaluations were conducted with three different resource selection algorithms, and it was found that AUReC was well-correlated with the results of all three systems. AUReC was also compared to rank-sensitive metrics NDCG and MAP in addition to Precision and was found to track all three metrics closely. Correlation was affected when the evaluation depth was decreased, but the slight drop in Pearson's r was largely due to the increased variance in the end-to-end system evaluations.

In addition, D_q was generated with different methods used to investigate how sensitive AUReC is to changes in the source of information. When D_q was generated with a weaker search engine, the correlated degraded somewhat but not nearly to the degree of differences in the accuracy of the search engines. When D_q was generated from different queries, the resulting AUReC scores remained highly correlated. More queries produced better correlated results, but there were diminishing returns.

AUReC allows system designers to quickly test a large number of shard maps to tune the accuracy of a new selective search system, a task which used to be prohibitively expensive. We present some guidelines on using AUReC. First, to generate D_q , the set of documents that should be retrieved for query q, the strongest search engine available is preferred. However, an out-of-

the-box retrieval still produces reliable results. More queries generate more consistent results with less variance. However, there are diminishing gains after about 800 queries.

AUReC as presented in this chapter used the assumption that selective search needs to return deep result lists in which the order of the documents do not matter. In cases where this is not true, AUReC could be extended to support different use-cases. For example, to better tune shard maps to a rank-sensitive metric, rather than ordering shards by the number of important documents they contain, one could order shards based on a rank-discounted sum of the scores of the documents. Or, by considering a smaller D_q , it may be possible to better tune AUReC to shallower metrics. AUReC is an effectiveness-driven measure. Another interesting exploration would be to incorporate the evaluation of efficiency into shard map scoring to create a unified metric that encompasses both areas and allows for a system designer to easily trade off between the two concerns. We leave these extensions for future work.

CONCLUSION

This dissertation describes the selective search architecture then identifies and addresses several outstanding questions on the suitability of its use in large, practical search and text analysis pipelines. The dissertation provides assurances on: the variance in effectiveness of different selective search instances; selective search performance in combination with dynamic pruning optimization techniques; the ability of selective search to serve as a high-recall first-stage retrieval system through accurate resource selection; and selective search performance related to parallel query processing, including load balancing. It demonstrates that selective search is an effective distributed search architecture that greatly reduces the computational cost of search in realistic production environments. It also provides useful tools for systems administrators to evaluate and test selective search to encourage wider adoption. This chapter details the findings and the wider impact of this dissertation, and suggests future research directions.

9.1 SUMMARY OF FINDINGS

Selective search uses non-deterministic processes for shard creation and for sample-based resource selection algorithms. This work explores the impact of these random decisions on selective search effectiveness. Overall, the random decisions did not strongly affect selective search accuracy (RQ 1). The results of the system were stable at the top most ranks and still reasonable at deeper ranks. Most of the variance came from a small number of queries, where relevant documents were placed into clusters that were not representative of them. In particular, the queries with the highest variance were typically rare queries, suggesting that query type can influence the effectiveness of selective search.

Another outstanding issue explored is the possible effects of combining selective search with dynamic pruning optimization methods such as WAND, which are common in practical systems (RQ 2). This dissertation shows that selective search and WAND have independent cost savings. In addition, with good shard selection, selective search and WAND has *better*-than-additive gains due to the long, skewed posting lists generated by the clustering process. In experimenting with different ways of combining WAND and selective search, it was shown that when shards are searched in sequence and the WAND state is passed on between shards, significant computational cost savings can be realized (RQ 3). While a strict sequential shard search is likely undesirable for interactive use-cases, it is practical for offline tasks and it opens the possibility of a tiered shard search that can realize some of the savings while retaining competitive latency.

New light-weight methods of resource selection were presented, which utilize existing collection statistics gathered by Block-Max WAND (RQ 4). For single-term queries, these methods are statistically non-inferior to exhaustive search, an improvement over prior methods. This suggests that with a good way to combine individual term data, it would be possible to perform resource selection using existing statistics. Furthermore, it is possible predict a reasonably sized set of shards which would exactly reproduce the exhaustive search results to a given depth in the rank. This creates an avenue for performing exact search with selective search, i.e. a guaranteed exact reproduction of exhaustive search, a new use-case that is currently unsupported.

This thesis improves the overall accuracy of selective search by introducing a new resource selection algorithm, learning to rank resources. This supervised method is suitable for quickly ranking hundreds of resources. The presented method does not require human judged training data and is able to produce good rankings by using exhaustive search results as a training target (RQ 6). Experimental results showed that learning to rank resources is able to closely replicate exhaustive search results in deep, recall-oriented metrics such as MAP@1000, which was not previously possible in selective search (RQ 5). Finally, in a feature ablation study it was shown that the learning to rank resources can be run with efficient features while largely maintaining the accuracy of the full set of slower features. The high accuracy in recall-oriented metrics enables the use of selective search as a first-stage retrieval in a multi-stage text analysis pipeline.

Kulkarni and Callan [2010a] has shown that selective search is very efficient in an environment evaluating a single query at a time. In this thesis, we extend this claim to a more realistic fully parallel processing environment, where multiple queries can be processed by the system in parallel (RQ 7). Using a long query stream and large datasets, we evaluate selective search under load. It was found that in a wide range of hardware configurations, selective search had higher throughput than a typical distributed search setup.

Selective search can be used with various resource selection algorithms, which affects its performance characteristics (RQ 8). We found that while Rank-S was more accurate, it was also more computationally expensive than Taily. The higher computational cost of Rank-S can cause load imbalances in a naive set up, which can be alleviated by replicating the resource selection database in multiple machines.

We also address concerns of a potential load imbalance in selective search in a multi-machine, parallel processing environment under load. The shards produced by selective search are topically focused. Some topics are naturally more popular than others, causing certain shards to be selected at a higher frequency. This causes hotspots in the machines hosting those shards, which become more severe as the number of machines are increased. This issue was addressed using training queries to estimate shard popularity and by assigning the shards to available machines in a greedy fashion (RQ 9). This method outperforms 10 different random shard partitions. In addition, experiments with queries a week and a month into the future from the initial shard allocation suggest that the broad trends of topic popularity are relatively stable and shards do not need to be reassigned on a frequent basis.

Combining this load balancing method with an efficient resource selection algorithm such as Taily, selective search displays higher throughput and lower latency than typical distributed search in the parallel processing environment containing a small number of machines. We also briefly explored the use of selective search in medium scale environments and found that selective search maintains its throughput advantage (RQ 10).

Finally, this work presents AUReC, a metric for evaluating the effectiveness of shard maps. AUReC does not require human judged relevance assessments to calculate and can be calculated using only the exhaustive retrieval from an out-of-the-box search engine. It is independent of resource selection algorithms and does not require an end-to-end evaluation of selective search to calculate, making it efficient. This is especially useful when comparing a large number of shard maps.

AUReC is highly correlated with end-to-end selective search system evaluations. Given a set of shard maps, AUReC produces a relative ordering among them that is consistent with end-to-end accuracy (RQ 11) and it agrees with end-to-end system evaluations on whether differences between shard maps are statistically significant (RQ 12).

AUReC is a robust metric (RQ 13); it is highly correlated with the end-to-end accuracy of selective search using several different resource selection algorithms evaluated with multiple traditional IR metrics at different depths. AUReC remains highly correlated even when it is calculated from weaker exhaustive search results and when it is calculated using similar, but different queries than the end-to-end evaluation queries. AUReC enables the rapid evaluation of different selective search configurations, allowing system administrators to make the best choices for their circumstances.

9.2 CONTRIBUTIONS AND WIDER IMPACT

With the tools and insights developed by this thesis, an organization can deploy a selective search system with confidence that it will have high efficiency and effectiveness. This both benefits immediate, practical users of selective search and empowers future research.

Prior work in selective search presented a proof-of-concept of a new architecture and demonstrated that it works in laboratory settings. It was initially billed as an architecture aimed primarily for small institutions such as research labs and start-ups and was meant to enable these organizations to work with large, web-scale corpora.

The research insights in this thesis established that selective search is effective in realistic production environments of larger scale that use parallel query processing, dynamic optimization and have large query loads. This thesis also demonstrated, for the first time, that selective search has high accuracy in recall-oriented metrics. This is a critical feature for modern multi-stage information retrieval systems that use slower, more complex re-rankers on top of a fast first-stage retrieval. The newly established accuracy in recall enables selective search to be used as a fast first-stage retrieval process in modern retrieval pipelines. In short, we now have confidence that selective search will increase the efficiency of real production systems.

This thesis also provided two useful new tools for prototyping a selective search system. First, it presented a detailed simulator software package that models different hardware configurations for selective search with configurable I/O and network costs, and accurately characterizes its efficiency. This tool was used develop an efficient way to allocate shards to machines to prevent load imbalances. With a few measurements of their current system, the simulator allows an architect to quickly prototype and evaluate the efficiency of a wide variety of selective search configurations without the prohibitive cost of building multiple real implementations.

In addition, it presented AUReC, a simple, efficient, and accurate metric for shard maps. AU-ReC can be calculated from the result list of an exhaustive search retrieval from an out-of-the-box search engine for a reasonable number of queries with no relevance judgements required. This can be used to quickly evaluate the effectiveness of a large number of shard maps so that the best possible way to partition the collection can be found. Together, these critical tools enable system administrators to compare and evaluate a large variety of configurations of selective search to find the settings for optimal efficiency and effectiveness without laborious implementation work.

Adoption of selective search may have significant impact on industry and academia alike. This thesis paved the path for the acceptance of selective search in production environments in industry. The key advantage of selective search in this setting is its high throughput obtained

from its low consumption of total computational resources. With more powerful computing resources becoming available to search increasingly larger data collections, electricity consumption has become a significant cost of large scale search in industry. Selective search, with its low total computational cost, is an alternative to traditional distributed search with a smaller carbon footprint.

In academia, adoption of selective search enables organizations with fewer computing resources such as university research labs to search larger collections. This thesis provided assurances for high accuracy in recall-oriented metrics. This is especially important for tasks that build on top of large scale text search such as text mining or question answering. With these assurances, researchers who use information retrieval as a first stage in a longer pipeline can use a selective search architecture. Selective search empowers research labs with limited resources and gives them tools to innovate in the big data space that may be difficult to access otherwise.

The contributions of this thesis also have wider impact on future research directions as well. AUReC, the newly proposed metric for evaluating the effectiveness of shard maps, is not limited to selective search. It can be used as a criterion for general clustering algorithms, especially if there is a small subset of objects that are more important than others. Because calculating AUReC is simple and fast, it can be used as an efficient optimization target, empowering clustering research.

Confidence in selective search as an architecture opens up future research directions that relax assumptions made in this dissertation for deeper analysis and finer tuning. The core assumption of selective search, which is that clustering related documents together leads to efficiency gains with minimal accuracy loss, was shown to hold true thus far and is likely to remain so. However, much of the work in the dissertation also made an implicit assumption that it is desirable that the relevant documents for a single should be placed in as few shards as possible. If this assumption is relaxed and documents relevant to different facets of a query are placed in different shards, it may be possible to use selective search to enhance or control the amount of diversity in the result set.

In addition, this dissertation took a component-by-component approach to evaluating and improving selective search. However, some components of selective search are intrinsically tied, such as resource selection and shard map creation, and may be targets for joint optimization. This is now feasible due to the fast shard map evaluations enabled by AUReC.

Finally, with the assurance that the underlying core architecture is sound, researchers can extend selective search in interesting ways to adapt the architecture to specialized domains and tasks.

9.3 FUTURE WORK

Selective search has been shown to be a robust architecture suitable for a practical, large-scale environments. This dissertation established that selective search is capable of being the first-stage retrieval in a more complex pipeline. For wider adoption of selective search, continued concrete investigation into this setup must be conducted.

While research in this dissertation has been conducted with web data, there are many other different domains and tasks where selective search could be of benefit, and there may be additional accuracy and effectiveness gains to be had in these areas if the selective search architecture is judiciously adapted. By supporting additional types of data beyond web documents, it increases the appeal of selective search. This section presents two possible research directions that specializes selective search to different types of corpora and purposes: extending selective search to faceted, hierarchical data; and selective search in large, online data streams.

9.3.1 Selective search in pipelines

Research in this dissertation achieved good performance on recall-oriented metrics for selective search; showed that it plays well with other dynamic optimization; established that it can be load-balanced with easy solutions; and showed that the variance of different instances are small. This opens the avenue for selective search to be used as a fast, first-stage retrieval system in complex text analysis pipelines, which would positively impact a wide range of use-cases such as modern multi-stage information retrieval systems, text mining, and question answering. This possibility warrants more substantive research and a thorough investigation. In addition, by better understanding the downstream processes using selective search for each use-case and increase its effectiveness.

In order to increase confidence that selective search would be truly effective in a large-scale, multi-stage pipeline, some questions must be answered. First, the interaction of selective search with other common search components must be investigated in detail. In efficiency, selective search should be studied in combination with tiering and replication, which are common strategies for improving efficiency of a large-scale system. Various configurations of these combinations are possible and discovering the best setting is an open research problem. For example, selective search could be implemented within a tier, or tiers could be created within topical shards. The differences in topic popularity of selective search can influence replication strategies; perhaps it would be easier to predict which shards will need dynamic replication with topically focused shards rather than randomly distributed shards.

There are components that affect accuracy that must be studied in combination with selective search as well, such as machine learned re-rankers. While this dissertation showed that selective search can achieve good recall, the exact interactions between the results from selective search and downstream re-rankers are still unknown. The documents that selective search retrieve are usually from a small number of topical shards. The narrower focus of topic in the candidate set may help or hinder re-rankers.

Finally, after individual component interactions have been examined, it is important to conduct integration tests, i.e. studying a whole pipeline of multiple components in combination with selective search, to ensure that secondary interactions between components and selective search are understood. A careful examination of the interaction of components may open the possibility of adjusting selective search to create further benefits in the pipeline.

While in general good recall is necessary for multi-stage text analysis pipelines, exactly how deep the result list must be is application dependent. By studying how much recall is needed, efficiency gains could be made by dynamically adjusting the level of recall of selective search. However, currently there is no well-defined way to adjust the level of recall of selective search. While resource selection parameters can be changed, the relationship between the parameters and the recall they generate is not well-understood and even small adjustments in parameters can result in large changes in recall. In addition, selective search also lacks guarantees for its level of recall; while the overall recall accuracy is good, the recall of the results may vary from query to query. Providing adjustable knobs for recall up to and including complete recall with varying levels of guarantees would be advantageous.

Because selective search is powered by creating shards, some of the filtering effort that would normally be further downstream in the text pipeline, such as temporal or reading level filtering, may be easily shifted to the initial retrieval stage by adjusting the features used to create the selective search shards. The different types of tasks may also benefit from different organization of shards, e.g. from documents clustered by popularity, time, vocabulary difficulty etc. Determining what similarity features best benefit which tasks is an open problem.

Similar adjustments can also be made to the resource selection and result merging stage of selective search to better fit the needs of a downstream process. For example, to increase diversity in the results, resource selection could be modified to select topically diverse set of shards and in the merging step preferentially show documents from many different shards rather than focusing on documents from a single shard.

9.3.2 Selective search in hierarchical corpora

Extant selective search research has dealt with web documents and assumed a single bag-of-words representation. This is a major conceptual limitation that prevents selective search from benefiting from richer, structured data commonly present in applications such as news, product search in e-commerce, medical search, and legal search. In these domains, documents are often organized in hierarchies and have well-defined attribute labels in addition to free text descriptions. For example, an iPhone is categorized under *Cell phones*, which is under *Electronics*. Selective search may then be modified to be aware of the multi-leveled nature of the hierarchy in various processes. Extending selective search to take advantage of the additional information allows us to apply the efficiency benefits of selective search to these new data sources.

Rather than using a clustering algorithm such as K-means to create topical clusters, selective search can leverage the hierarchical organization to create shards; each leaf-level node in the hierarchy becomes a shard, with shards retaining the parent-child relationship from the hierarchy. If the dataset has very fine-grained categories, interesting research questions arise on the level of granularity to create the shards, because very small shards can cause a significant search overhead.

Resource selection in a hierarchical setting requires specialization. Rather than ranking shards, a resource selection algorithm for hierarchical selective search must generate candidate hierarchy branches to search. In addition to determining which and how many different branches to search, hierarchical resource selection also needs to consider the granularity of search. For example, if the query is broad, coarser grained categories may be searched than if a query is very specific, in which case finer grained categories may be chosen. Interesting research questions remain in what level of hierarchy should be selected for a particular query; whether the entire branch should be searched or a select few sub-branches should be searched; and the effect of these policies on the efficiency and accuracy of the system. Conversely, it may be that efficiency constraints drive these decisions as well. When merging the results from the shard search, the categorical hierarchy of the shards may inform strategies such as diversification; e.g. distance in the hierarchy could be used as a similarity metric.

In addition, selective search does not need to be limited to one particular shard map for a dataset. At the cost of some disk space, which is typically cheap, clustering on different attributes of structured data can lead to creating multiple different ways of partitioning or views of the data which may be advantageous for different types of queries. Queries can then be routed to a specific view of the corpus or multiple views may be used for a query and the results combined. For

example, one view of product corpora could cluster on price in addition to product hierarchies. This price-based view of the corpus could help eliminate problems such as iPhone search results being flooded with cheap iPhone accessories. Determining how many views to create and which view to use for a given query are open problems.

9.3.3 Selective search in streaming datasets

While viewing corpora as static is convenient, there are an increasing number of important data sources that are temporally-oriented and rapidly updated, e.g. Twitter. These data sources are often high volume so search efficiency in this space is much needed and would enable systems to hold and search more data, efficiently; i.e. this domain is potentially a good match for selective search. There are many challenges in adapting selective search to online data because of the characteristics of the data and because user search behavior in Twitter differ significantly from web search [Teevan et al. 2011]. Users search Twitter for temporally relevant information and to monitor changes. In other words, the data that the users are most interested in are in a stage of constant flux.

Selective search must be modified to be able to organize constant updates into shards. Streaming clustering algorithms [Ailon et al. 2009] can be used for this purpose. However, there are still interesting research problems surrounding the issue. One such problem is determining policies on how long to keep the data. Simple policies may just use a strict time cut-off (i.e. only keep data from the past n months), one can imagine using quality as a component in the decision to keep or discard a tweet. For example, documents that have been highly ranked in many queries may be kept longer than documents were never relevant to any queries. Sudden bursts in a topic due current events may also pose challenges. This may cause a topic shard to become size-skewed which is undesirable for efficiency reasons and may warrant splitting the topic into further shards. Conversely, there may be times when shards with less incoming documents should be merged. Finally, as the shards are being updated, the resource selection database must be updated as well; policies on how frequently this must occur are an open question and would represent a trade-off in accuracy and efficiency.

Online corpora often have a strong temporal component that is critical for accurate search [Teevan et al. 2011] and selective search must be modified to be aware of this attribute throughout its process. During clustering, the time stamp of the data should inform the similarity metric so that items from similar times are more likely to be clustered together. During resource selection, careful attention should be paid to the temporal component of the shards as relevant documents may occur around the same time period. Finally, during result merging the final ranking of the documents should take their time stamps into account.

Finally, in online data sets, using predetermined shard assignments to machines may not a practical method of load balancing because the sizes of the shards may change as current events influence the volume of documents in different topics in recent time periods; e.g. sports generate a lot more volume during the Olympics. An online search system must be able to respond to rapidly changing data and queries. To achieve this, selective search must be able to load balance on-the-fly. It must be able to detect sudden increases in load on a particular topic and alleviate the load through methods such as dynamic replication of the hot shard.

BIBLIOGRAPHY

- Nir Ailon, Ragesh Jaiswal, and Claire Monteleoni. Streaming k-means approximation. In *Advances in Neural Information Processing Systems* 22, pages 10–18, 2009.
- Mohammad Aliannejadi, Hamed Zamani, Fabio Crestani, and W. Bruce Croft. Target apps selection: Towards a unified search framework for mobile devices. In *Proceedings of the 41st International ACM SIGIR Conference on Research and Development in Information Retrieval*, pages 215–224, 2018.
- Ismail Sengor Altingovde, Engin Demir, Fazli Can, and Özgür Ulusoy. Incremental cluster-based retrieval using compressed cluster-skipping inverted files. *ACM Transactions on Information Systems*, 26(3):15:1–15:36, 2008.
- Robin Aly, Djoerd Hiemstra, and Thomas Demeester. Taily: Shard selection using the tail of score distributions. In *Proceedings of the 36th Annual International ACM SIGIR Conference on Research and Development in Information Retrieval*, pages 673–682, 2013.
- Yael Anava, Anna Shtok, Oren Kurland, and Ella Rabinovich. A probabilistic fusion framework. In *Proceedings of the 25th ACM International Conference on Information and Knowledge Management*, pages 1463–1472, 2016.
- Vo Ngoc Anh and Alistair Moffat. Impact transformation: Effective and efficient web retrieval. In *Proceedings of the 25th Annual International ACM SIGIR Conference on Research and Development in Information Retrieval*, pages 3–10, 2002.
- Jaime Arguello. Aggregated search. *Foundations and Trends in Information Retrieval*, 10(5):365–502, 2017.
- Jaime Arguello, Jamie Callan, and Fernando Diaz. Classification-based resource selection. In *Proceedings of the 18th ACM International Conference on Information and Knowledge Management*, pages 1277–1286, 2009a.
- Jaime Arguello, Fernando Diaz, Jamie Callan, and Jean-Francois Crespo. Sources of evidence for vertical selection. In *Proceedings of the 32nd Annual International ACM SIGIR Conference on Research and Development in Information Retrieval*, pages 315–322, 2009b.
- Claudine S. Badue, Ricardo Baeza-Yates, Berthier Ribeiro-Neto, Artur Ziviani, and Nívio Ziviani. Analyzing imbalance among homogeneous index servers in a web search system. *Information Processing Management*, 43(3):592–608, 2007.
- Ricardo Baeza-Yates, Carlos Castillo, Flavio Junqueira, Vassilis Plachouras, and Fabrizio Silvestri. Challenges on distributed web retrieval. In *Proceedings of the 23rd IEEE International Conference on Data Engineering*, pages 6–20, 2007a.
- Ricardo Baeza-Yates, Aristides Gionis, Flavio Junqueira, Vanessa Murdock, Vassilis Plachouras, and Fabrizio Silvestri. The impact of caching on search engines. In *Proceedings of the 30th*

Annual International ACM SIGIR Conference on Research and Development in Information Retrieval, pages 183–190, 2007b.

- Ricardo Baeza-Yates, Aristides Gionis, Flavio Junqueira, Vassilis Plachouras, and Luca Telloli. On the feasibility of multi-site web search engines. In *Proceedings of the 18th ACM International Conference on Information and Knowledge Management*, pages 425–434, 2009a.
- Ricardo Baeza-Yates, Vanessa Murdock, and Claudia Hauff. Efficiency trade-offs in two-tier web search systems. In *Proceedings of the 32nd Annual International ACM SIGIR Conference on Research and Development in Information Retrieval*, pages 163–170, 2009b.
- Krisztian Balog. Learning to combine collection-centric and document-centric models for resource selection. In *Proceedings of the 23rd Text REtrieval Conference*, 2014.
- Luiz Andró Barroso, Jeffrey Dean, and Urs Hölzle. Web search for a planet: The Google cluster architecture. *IEEE Micro*, 23(2):22–28, 2003.
- Ulf Brefeld, B. Barla Cambazoglu, and Flavio P. Junqueira. Document assignment in multi-site search engines. In *Proceedings of the 4th ACM International Conference on Web Search and Data Mining*, pages 575–584, 2011.
- Daniele Broccolo, Craig Macdonald, Salvatore Orlando, Iadh Ounis, Raffaele Perego, Fabrizio Silvestri, and Nicola Tonellotto. Query processing in highly-loaded search engines. In *Proceedings* of the 20th International Symposium on String Processing and Information Retrieval, pages 49–55, 2013.
- Andrei Z. Broder, David Carmel, Michael Herscovici, Aya Soffer, and Jason Zien. Efficient query evaluation using a two-level retrieval process. In *Proceedings of the 12th ACM International Con-ference on Information and Knowledge Management*, pages 426–434, 2003.
- Fidel Cacheda, Vassilis Plachouras, and Iadh Ounis. Performance analysis of distributed architectures to index one terabyte of text. In *Advances in Information Retrieval: 26th European Conference on IR Research*, pages 394–408, 2004.
- Fidel Cacheda, Victor Carneiro, Vassilis Plachouras, and Iadh Ounis. Performance comparison of clustered and replicated information retrieval systems. In *Advances in Information Retrieval: 29th European Conference on IR Research*, pages 124–135, 2007a.
- Fidel Cacheda, Victor Carneiro, Vassilis Plachouras, and Iadh Ounis. Performance analysis of distributed information retrieval architectures using an improved network simulation model. *Information Processing and Management*, 43(1):204–224, 2007b.
- Brendon Cahoon, Kathryn S. McKinley, and Zhihong Lu. Evaluating the performance of distributed architectures for information retrieval using a variety of workloads. *ACM Transactions on Information Systems*, 18(1):1–43, 2000.
- Jamie Callan. Advances in Information Retrieval: Recent Research from the Center for Intelligent Information Retrieval, chapter Distributed Information Retrieval, pages 127–150. 2000.
- B. Barla Cambazoglu, Emre Varol, Enver Kayaaslan, Cevdet Aykanat, and Ricardo Baeza-Yates. Query forwarding in geographically distributed search engines. In *Proceedings of the 33rd Annual*

International ACM SIGIR Conference on Research and Development in Information Retrieval, pages 90–97, 2010.

- B. Barla Cambazoglu, Enver Kayaaslan, Simon Jonassen, and Cevdet Aykanat. A term-based inverted index partitioning model for efficient distributed query processing. *ACM Transactions on the Web*, 7(3):15:1–15:23, 2013.
- Fazli Can, Ismail Sengör Altingövde, and Engin Demir. Efficiency and effectiveness of query processing in cluster-based retrieval. *Information Systems*, 29(8):697–717, 2004.
- David Carmel, Doron Cohen, Ronald Fagin, Eitan Farchi, Michael Herscovici, Yoëlle S. Maarek, and Aya Soffer. Static index pruning for information retrieval systems. In *Proceedings of the 24th Annual International ACM SIGIR Conference on Research and Development in Information Retrieval*, pages 43–50, 2001.
- Ben Carterette, Virgil Pavlu, Hui Fang, and Evangelos Kanoulas. Million query track 2009 overview. In *Proceedings of the 18th Text REtrieval Conference*, 2009.
- Suleyman Cetintas, Luo Si, and Hao Yuan. Learning from past queries for resource selection. In *Proceedings of the 18th ACM International Conference on Information and Knowledge Management*, pages 1867–1870, 2009.
- Mon-Shih Chuang and Anagha Kulkarni. Balancing precision and recall with selective search. In *Proceedings of the 4th Annual International Symposium on Information Management and Big Data*, pages 151–160, 2017a.
- Mon Shih Chuang and Anagha Kulkarni. Improving shard selection for selective search. In *Information Retrieval Technology:* 13th Asia Information Retrieval Societies Conference, pages 29–41, 2017b.
- Charles L. A. Clarke, Nick Craswell, Ian Soboroff, and Ellen M. Voorhees. Overview of the TREC 2011 web track. In *Proceedings of the 20th Text REtrieval Conference*, 2011.
- Charles L. A. Clarke, J. Shane Culpepper, and Alistair Moffat. Assessing efficiency–effectiveness tradeoffs in multi-stage retrieval systems without using relevance judgments. *Information Retrieval*, 19(4):351–377, 2016.
- Gordon V. Cormack, Mark D. Smucker, and Charles L. A. Clarke. Efficient and effective spam filtering and re-ranking for large web datasets. *Information Retrieval*, 14(5):441–465, 2011.
- W. Bruce Croft. A model of cluster searching based on classification. 5(3):189–195, 1980.
- J. Shane Culpepper, Charles L. A. Clarke, and Jimmy Lin. Dynamic cutoff prediction in multistage retrieval systems. In *Proceedings of the 21st Australasian Document Computing Symposium*, pages 17–24, 2016.
- Zhuyun Dai and Jamie Callan. Inverted list caching for topical index shards. In *Advances in Information Retrieval: 40th European Conference on IR Research*, pages 577–583, 2018.
- Zhuyun Dai, Yubin Kim, and Jamie Callan. How random decisions affect selective distributed search. In *Proceedings of the 38th Annual International ACM SIGIR Conference on Research and Development in Information Retrieval*, pages 771–774, 2015.

- Zhuyun Dai, Chenyan Xiong, and Jamie Callan. Query-biased partitioning for selective search. In *Proceedings of the 25th ACM International Conference on Information and Knowledge Management*, pages 1119–1128, 2016.
- Zhuyun Dai, Yubin Kim, and Jamie Callan. Learning to rank resources. In *Proceedings of the 40th Annual International ACM SIGIR Conference on Research and Development in Information Retrieval*, pages 837–840, 2017.
- DESMO-J. Desmo-j. http://desmoj.sourceforge.net/ [Accessed: Oct 19, 2014].
- Constantinos Dimopoulos, Sergey Nepomnyachiy, and Torsten Suel. Optimizing top-k document retrieval strategies for block-max indexes. In *Proceedings of the 6th ACM International Conference on Web Search and Data Mining*, pages 113–122, 2013.
- Mauro Dragoni, Andi Rexha, Hermann Ziak, and Roman Kern. A semantic federated search engine for domain-specific document retrieval. In *Proceedings of the Symposium on Applied Computing*, pages 303–308, 2017.
- Jonathan L. Elsas, Jaime Arguello, Jamie Callan, and Jaime G. Carbonell. Retrieval and feedback models for blog feed search. In *Proceedings of the 31st Annual International ACM SIGIR Conference* on Research and Development in Information Retrieval, pages 347–354, 2008.
- Guillem Francès, Xiao Bai, B. Barla Cambazoglu, and Ricardo Baeza-Yates. Improving the efficiency of multi-site web search engines. In *Proceedings of the 7th ACM International Conference on Web Search and Data Mining*, pages 3–12, 2014.
- Ana Freire, Craig Macdonald, Nicola Tonellotto, Iadh Ounis, and Fidel Cacheda. Hybrid query scheduling for a replicated search engine. In *Advances in Information Retrieval: 35th European Conference on Advances in IR Research*, pages 435–446, 2013.
- James C. French and Allison L. Powell. Metrics for evaluating database selection techniques. *World Wide Web*, 3(3):153–163, 2000.
- Luis Gravano, Héctor García-Molina, and Anthony Tomasic. GlOSS: Text-source discovery over the internet. *ACM Transactions on Database Systems*, 24(2):229–264, 1999.
- Alan Griffiths, H. Claire Luckhurst, and Peter Willett. Using inter-document similarity information in document retrieval systems. *Journal of the American Society for Information Science*, 37(1): 3–11, 1986.
- Feng Guan, Shuiyuan Zhang, Chunmei Liu, Xiaoming Yu, Yue Liu, and Xueqi Cheng. ICTNET at federated web search track 2014. In *Proceedings of the 23rd Text REtrieval Conference*, 2014.
- Fatih Hafizoglu, Emre Can Kucukoglu, and Ismail Sengor Altingovde. On the efficiency of selective search. In *Advances in Information Retrieval: 39th European Conference on IR Research*, pages 705–712, 2017.
- Baoli Han, Ling Chen, and Xiaoxue Tian. Knowledge based collection selection for distributed information retrieval. *Information Processing & Management*, 54(1):116–128, 2018.
- David Hawking and Paul Thistlewaite. Methods for information server selection. *ACM Transactions on Information Systems*, 17(1):40–76, 1999.

- Dzung Hong, Luo Si, Paul Bracke, Michael Witt, and Tim Juchcinski. A joint probabilistic classification model for resource selection. In *Proceedings of the 33rd Annual International ACM SIGIR Conference on Research and Development in Information Retrieval*, pages 98–105, 2010.
- Gaya K. Jayasinghe, William Webber, Mark Sanderson, Lasitha S. Dharmasena, and J. Shane Culpepper. Evaluating non-deterministic retrieval systems. In *Proceedings of the 37th Annual International ACM SIGIR Conference on Research and Development in Information Retrieval*, pages 911–914, 2014.
- Gaya K. Jayasinghe, William Webber, Mark Sanderson, Lasitha S. Dharmasena, and J. Shane Culpepper. Statistical comparisons of non-deterministic IR systems using two dimensional variance. *Information Processing & Management*, 51(5):677–694, 2015.
- Thorsten Joachims. Training linear SVMs in linear time. In *Proceedings of the 12th ACM SIGKDD International Conference on Knowledge Discovery and Data Mining*, pages 217–226, 2006.
- Changsung Kang, Xuanhui Wang, Yi Chang, and Belle Tseng. Learning to rank with multi-aspect relevance for vertical search. In *Proceedings of the 5th ACM International Conference on Web Search and Data Mining*, pages 453–462, 2012.
- Jinyoung Kim and W. Bruce Croft. Ranking using multiple document types in desktop search. In *Proceedings of the 33rd Annual International ACM SIGIR Conference on Research and Development in Information Retrieval*, pages 50–57, 2010.
- Yubin Kim and Jamie Callan. Measuring the effectiveness of selective search index partitions without supervision. In *Proceedings of the 4th ACM SIGIR International Conference on the Theory of Information Retrieval*, pages 91–98, 2018.
- Yubin Kim, Jamie Callan, J. Shane Culpepper, and Alistair Moffat. Does selective search benefit from WAND optimization? In *Advances in Information Retrieval:* 38th European Conference on IR *Research*, pages 145–158, 2016.
- Yubin Kim, Jamie Callan, J. Shane Culpepper, and Alistair Moffat. Efficient distributed selective search. *Information Retrieval*, 20(3):221–252, 2017.
- Anagha Kulkarni. *Efficient and Effective Large-Scale Search*. PhD thesis, Carnegie Mellon University, 2013.
- Anagha Kulkarni. ShRkC: Shard rank cutoff prediction for selective search. In *String Processing and Information Retrieval*, pages 337–349, 2015.
- Anagha Kulkarni and Jamie Callan. Document allocation policies for selective searching of distributed indexes. In *Proceedings of the 19th ACM International Conference on Information and Knowledge Management*, pages 449–458, 2010a.
- Anagha Kulkarni and Jamie Callan. Topic-based index partitions for efficient and effective selective search. In *SIGIR Workshop on Large-Scale Distributed Information Retrieval*, 2010b.
- Anagha Kulkarni and Jamie Callan. Selective search: Efficient and effective search of large textual collections. *ACM Transactions on Information Systems*, 33(4):17:1–17:33, 2015.

- Anagha Kulkarni, Almer Tigelaar, Djoerd Hiemstra, and Jamie Callan. Shard ranking and cutoff estimation for topically partitioned collections. In *Proceedings of the 21st ACM International Conference on Information and Knowledge Management*, pages 555–564, 2012.
- Daniel Lemire and Leonid Boytsov. Decoding billions of integers per second through vectorization. *Software: Practice & Experience*, 41(1):1–29, 2015.
- Tie-Yan Liu. Learning to rank for information retrieval. *Foundations and Trends in Information Retrieval*, 3(3):225–331, 2009.
- Xiaoyong Liu and W. Bruce Croft. Cluster-based retrieval using language models. In *Proceedings* of the 27th Annual International ACM SIGIR Conference on Research and Development in Information Retrieval, pages 186–193, 2004.
- Jie Lu and Jamie Callan. User modeling for full-text federated search in peer-to-peer networks. In *Proceedings of the 29th Annual International ACM SIGIR Conference on Research and Development in Information Retrieval*, pages 332–339, 2006.
- Xiaolu Lu, Alistair Moffat, and J. Shane Culpepper. The effect of pooling and evaluation depth on IR metrics. *Information Retrieval*, 19(4):416–445, 2016.
- Claudio Lucchese, Salvatore Orlando, Raffaele Perego, and Fabrizio Silvestri. Mining query logs to optimize index partitioning in parallel web search engines. In *Proceedings of the 2nd International Conference on Scalable Information Systems*, pages 43:1–43:9, 2007.
- Craig Macdonald, Nicola Tonellotto, and Iadh Ounis. Learning to predict response times for online query scheduling. In *Proceedings of the 35th Annual International ACM SIGIR Conference on Research and Development in Information Retrieval*, pages 621–630, 2012.
- Craig Macdonald, Rodrygo L. Santos, and Iadh Ounis. The whens and hows of learning to rank for web search. *Information Retrieval*, 16(5):584–628, 2013.
- Donald Metzler and W. Bruce Croft. A Markov random field model for term dependencies. In *Proceedings of the 28th Annual International ACM SIGIR Conference on Research and Development in Information Retrieval*, pages 472–479, 2005.
- Alistair Moffat, William Webber, and Justin Zobel. Load balancing for term-distributed parallel retrieval. In *Proceedings of the 29th Annual International ACM SIGIR Conference on Research and Development in Information Retrieval*, pages 348–355, 2006.
- Alistair Moffat, William Webber, Justin Zobel, and Ricardo Baeza-Yates. A pipelined architecture for distributed text query evaluation. *Information Retrieval*, 10(3):205–231, 2007.
- Hafeezul Rahman Mohammad, Keyang Xu, Jamie Callan, and J. Shane Culpepper. Dynamic shard cutoff prediction for selective search. In *Proceedings of the 41st International ACM SIGIR Conference on Research and Development in Information Retrieval*, pages 85–94, 2018.
- Salvatore Orlando, Raffaele Perego, and Fabrizio Silvestri. Design of a parallel and distributed web search engine. In *Proceedings of Parallel Computing (PARCO) 2001 Conference*, pages 197–204, 2001.

- Georgios Paltoglou, Michail Salampasis, and Maria Satratzemi. Integral based source selection for uncooperative distributed information retrieval environments. In *Proceedings of the 2008 ACM Workshop on Large-Scale Distributed Systems for Information Retrieval*, pages 67–74, 2008.
- Matthias Petri, J. Shane Culpepper, and Alistair Moffat. Exploring the magic of WAND. In *Proceedings of the Australasian Document Computing Symposium*, pages 58–65, 2013.
- Allison L. Powell, James C. French, Jamie Callan, Margaret Connell, and Charles L. Viles. The impact of database selection on distributed searching. In *Proceedings of the 23rd Annual International ACM SIGIR Conference on Research and Development in Information Retrieval*, pages 232–239, 2000.
- Diego Puppin, Fabrizio Silvestri, and Domenico Laforenza. Query-driven document partitioning and collection selection. In *Proceedings of the 1st International Conference on Scalable Information Systems*, page 34, 2006.
- Knut Magne Risvik, Yngve Aasheim, and Mathias Lidal. Multi-tier architecture for web search engines. In *Proceedings of the 1st Conference on Latin American Web Congress*, pages 132–143, 2003.
- Oscar Rojas, Veronica Gil-Costa, and Mauricio Marin. Distributing efficiently the block-max WAND algorithm. In *Proceedings of the International Conference on Computational Science*, pages 120–129, 2013.
- Michail Salampasis. *Federated Patent Search*, pages 213–240. Springer Berlin Heidelberg, Berlin, Heidelberg, 2017.
- Erik Selberg and Oren Etzioni. The metacrawler architecture for resource aggregation on the web. *IEEE Expert*, 12(1):11–14, Jan 1997.
- Jangwon Seo and W. Bruce Croft. Blog site search using resource selection. In *Proceedings of the* 17th ACM International Conference on Information and Knowledge Management, pages 1053–1062, 2008.
- Anand Lal Shimpi. Western digital's VelociRaptor VR200M: 10K new RPM at 450GB and 600GB. http://www.anandtech.com/show/3636/ [Accessed: western-digitals-new-velociraptor-vr200m-10k-rpm-at-450gb-and-600gb Oct 19, 2014].
- Milad Shokouhi. Central-rank-based collection selection in uncooperative distributed information retrieval. In *Advances in Information Retrieval: 29th European Conference on IR Research*, pages 160–172, 2007.
- Milad Shokouhi and Luo Si. Federated search. *Foundations and Trends in Information Retrieval*, 5 (1):1–102, 2011.
- Luo Si and Jamie Callan. Relevant document distribution estimation method for resource selection. In *Proceedings of the 26th Annual International ACM SIGIR Conference on Research and Development in Informaion Retrieval*, pages 298–305, 2003.
- Luo Si and Jamie Callan. The effect of database size distribution on resource selection algorithms. In *Distributed Multimedia Information Retrieval*, pages 31–42, 2004a.

- Luo Si and Jamie Callan. Unified utility maximization framework for resource selection. In *Proceedings of the 13th ACM International Conference on Information and Knowledge Management*, pages 32–41, 2004b.
- Luo Si and Jamie Callan. Modeling search engine effectiveness for federated search. In *Proceedings* of the 28th Annual International ACM SIGIR Conference on Research and Development in Information Retrieval, pages 83–90, 2005.
- Luo Si, Rong Jin, Jamie Callan, and Paul Ogilvie. A language modeling framework for resource selection and results merging. In *Proceedings of the 11th ACM International Conference on Information and Knowledge Management*, pages 391–397, 2002.
- Trevor Strohman, Howard Turtle, and W. Bruce Croft. Optimization strategies for complex queries. In *Proceedings of the 28th Annual International ACM SIGIR Conference on Research and Development in Information Retrieval*, pages 219–225, 2005.
- Jaime Teevan, Daniel Ramage, and Merredith Ringel Morris. #TwitterSearch: A comparison of microblog search and web search. In *Proceedings of the 4th ACM International Conference on Web Search and Data Mining*, pages 35–44, 2011.
- The Lemur Project. The Lemur Project. http://lemurproject.org/ [Accessed: Jun 19, 2015].
- Paul Thomas and Milad Shokouhi. SUSHI: Scoring scaled samples for server selection. In *Proceedings of the 32nd Annual International ACM SIGIR Conference on Research and Development in Information Retrieval*, pages 419–426, 2009.
- Anthony Tomasic and Héctor García-Molina. Performance of inverted indices in shared-nothing distributed text document information retrieval systems. In *Proceedings of the 2nd International Conference on Parallel and Distributed Information Systems*, pages 8–17, 1993.
- Anastasios Tombros, Robert Villa, and C. J. van Rijsbergen. The effectiveness of query-specific hierarchic clustering in information retrieval. *Information Processing Management*, 38(4):559–582, 2002.
- Nicola Tonellotto, Craig Macdonald, and Iadh Ounis. Efficient and effective retrieval using selective pruning. In *Proceedings of the 6th ACM International Conference on Web Search and Data Mining*, pages 63–72, 2013.
- Günter Urak, Hermann Ziak, and Roman Kern. Source selection of long tail sources for federated search in an uncooperative setting. In *Proceedings of the 33rd Annual ACM Symposium on Applied Computing*, pages 720–727, 2018.
- Cornelis Joost van Rijsbergen. Information Retrieval. Butterworths, 1979.
- Ellen M. Voorhees. The effectiveness and efficiency of agglomerative hierarchic clustering in document retrieval. Technical report, Cornell University, 1985.
- Esteban Walker and Amy S. Nowacki. Understanding equivalence and noninferiority testing. *Journal of General Internal Medicine*, 26(2):192–196, 2011.
- Yulu Wang and Jimmy Lin. Partitioning and segment organization strategies for real-time selective search on document streams. In *Proceedings of the 10th ACM International Conference on Web Search and Data Mining*, pages 221–230, 2017.

- William Webber and Alistair Moffat. In search of reliable retrieval experiments. In *Proceedings of the 10th Australasian Document Computing Symposium*, pages 26–33, 2005.
- William Webber, Alistair Moffat, and Justin Zobel. A similarity measure for indefinite rankings. *ACM Transactions on Information Systems*, 28(4):20:1–20:38, 2010.
- Peter Willett. Recent trends in hierarchic document clustering: A critical review. *Information Processing and Management*, 24(5):577–597, 1988.
- Fen Xia, Tie-Yan Liu, Jue Wang, Wensheng Zhang, and Hang Li. Listwise approach to learning to rank: Theory and algorithm. In *Proceedings of the 25th International Conference on Machine Learning*, pages 1192–1199, 2008.
- Jinxi Xu and W. Bruce Croft. Cluster-based language models for distributed retrieval. In *Proceedings of the 22nd Annual International ACM SIGIR Conference on Research and Development in Information Retrieval*, pages 254–261, 1999.
- Budi Yuwono and Dik Lun Lee. Server ranking for distributed text retrieval systems on internet. In *Proceedings of the International Conference on Database Systems for Advanced Applications*, pages 41–49, 1997.
- Jiangong Zhang and Torsten Suel. Optimized inverted list assignment in distributed search engine architectures. In *Proceedings of the Parallel and Distributed Processing Symposium*, pages 1–10, 2007.
- Shuo Zhang and Krisztian Balog. Design patterns for fusion-based object retrieval. In *Advances in Information Retrieval: 39th European Conference on IR Research,* pages 684–690, 2017.